MENTAL HEALTH AND SEXUAL RISK BEHAVIORS AMONG SOCIAL NETWORKS OF YOUNG TANZANIAN MEN

Lauren M. Hill

A dissertation submitted to the faculty at the University of North Carolina at Chapel Hill in partial fulfillment of the requirements for the degree of Doctor of Philosophy in the Department of Health Behavior in the Gillings School of Global Public Health.

Chapel Hill 2017

Approved by:

Suzanne Maman

Vivian Go

Nisha Gottfredson

James Moody

Brian Pence

© 2017 Lauren M. Hill ALL RIGHTS RESERVED

ABSTRACT

Lauren M. Hill: Mental Health and Sexual Risk Behaviors Among Social Networks of Young Tanzanian Men (Under the direction of Suzanne Maman)

Background: Young men are important targets in HIV prevention in Tanzania and throughout sub-Saharan Africa. Anxiety and depression are common among youth and may be important predictors of HIV risk behaviors. Evidence of these relationships in high-risk populations is needed, as is understanding of how normative and supportive elements of the social environment might be leveraged to buffer the risk posed by poor mental health.

Methods: I conducted two studies using baseline and one-year follow-up data from 1113 sexually active male participants in an ongoing HIV and intimate partner violence prevention trial taking place among 59 social groups known as "camps" in Dar es Salaam, Tanzania. In the first study, I assessed the relationship between mental health (anxiety, depression) and sexual risk behaviors (condom use, concurrency) using multilevel modeling. In the second study, I built upon these models to test the moderating role of sexual behavior norms, camp network social cohesion, and men's network centrality in the relationship between mental health and sexual risk behaviors. I further assessed peer behavioral influence in men's camp friendship networks using guadratic assignment procedures.

Results: Both anxiety and depression were independently associated with concurrency but only depression was independently associated with condom use. In the second study, changes in perceived descriptive norms were associated with both condom use and concurrency, perceived injunctive norms were associated with concurrency, and direct encouragement was associated with condom use.

Changes in condom use norms (perceived descriptive norms and encouragement) moderated the relationship between changes in anxiety symptoms and condom use such that the negative relationship was amplified with worsening norms for condom use, and attenuated by improving norms for condom use. Network centrality and cohesion were not significantly related to sexual risk, nor did they interact with anxiety or depression in relation to sexual risk.

Conclusions: The results of this dissertation indicate the importance of screening and providing treatment for depression and anxiety in high HIV-prevalence contexts, and the need to develop effective HIV prevention interventions targeting young men living with anxiety and depression. Such interventions may benefit from the combination of norms-based and mental health promotion approaches.

ACKNOWLEDGEMENTS

This dissertation is an accomplishment not of one but of many. It represents the culmination of years of training, mentorship, and support which have made me the scientist that I am today. I would like to thank my advisor, mentor, and dissertation chair, Dr. Suzanne Maman for her constant support and selfless encouragement of my research interests. You have been with me through my many twists and turns as a graduate student, always there as a sage advisor and a cheerleader as well. Few graduate students are as lucky as I have been to have mentorship like yours, and I will carry your support and the confidence it has fostered in me for the rest of my career.

I would also like to thank my wonderful committee for their guidance and support. Dr. Nisha Gottfredson, my guide to multilevel methods, you were always there for impromptu discussions which have fed my intellectual curiosity in statistics and have quelled any fears of asking "stupid" questions. Dr. James Moody, my social network analysis guru, how much I have learned from you about an area of social science research which was entirely new to me. Your enthusiasm for social network research has made me excited to continue to learn and develop methodologically as I progress through my career. Dr. Brian Pence, an expert in mental health and HIV, I want to thank you for always helping me to think through the logic of my dissertation analyses, and for always keeping me connected to the forefront of the literature on these topics. Last, but not least, I would like to thank Dr. Vivian Go, for always being a problem-solver, a listening ear, and a mentor to me.

My dissertation research would not have been possible without our fantastic research team in Dar es Salaam, Tanzania. I am so thankful for the opportunity to work with Lusajo Kajula, the co-PI of the parent trial of this study. Your interest in mental health among the men in the study and the many

v

conversations which we have had on the topic make me enthusiastic about the possibility of addressing mental health in settings like Tanzania. Likewise, I would like to thank Mrema Noel Kilonzo, as a colleague also interested in mental health in this context; it has been a pleasure collaborating with you. I would like to thank the larger research team in the US and in Tanzania, who contributed to the larger parent trial and supported my research: Nina Yamanis, Peter Balvanz, Marta Mulawa, Donaldson Conserve, Deus Kajuna, and Gema Lambert.

I would not have had the tools to conceive of and complete this dissertation without the tremendous training I received in the Health Behavior doctoral program. For her tireless enthusiasm for ever improving the quality of our training and mentorship, I would like to thank Dr. Susan Ennett. I could not have asked for a better training experience which met all of my goals to develop as an applied and interdisciplinary health researcher. To my cohort, thank you so much for traveling with me through this experience of a lifetime. The social, academic, and professional experience of the doctoral program would have not been the same without you. I have been so fortunate to learn and work with a group of such supportive and brilliant peers. I truly hope we will be colleagues for a lifetime.

I owe a debt to my parents for helping me to get where I am today. The financial and personal sacrifices which you made for my education I hope will ultimately be richly repaid in the form of research which serves the well-being of humanity in some important though perhaps not large way. I would like to thank my father, Dr. Christopher Hill, for encouraging dogged intellectual curiosity and a thirst for logical thinking. Because of your influence, I will never cease to question the status quo and search for solutions to humanities' problems. To my mother, Sharon Hill, I want to thank you for the value for education and strategic thinking which you instilled in me. I know it hasn't been easy to see your daughter make difficult choices, and I thank you for embracing the professional choices that I have made.

This dissertation research would not have been possible without generous funding from

vi

several sources. The parent study was supported by the National Institute of Mental Health (NIMH) through grant R01MH098690. I received financial support for my pre-doctoral training and dissertation research through the UNC Training Program in Sexually Transmitted Diseases and HIV (NIH/NIAID T32 AI007001-39). The financial support and training which I received in the Program, in addition to financial support from the UNC Graduate School Summer Research Fellowship and the Gillings Merit Award have been paramount in my ability to focus on preparing for and completing my dissertation research. I hope to be able to repay these investments in me as a scientist through my future contributions to public health research.

TABLE OF CONTENTS

LIST OF TABLESxii	i
LIST OF FIGURES	v
LIST OF ABBREVIATIONS	v
CHAPTER 1: INTRODUCTION	1
1.1 Problem statement	1
1.2 Study purpose	2
1.3 Research approach	2
Specific Aims	3
Study context	5
Overview of the parent trial	5
Overview of analyses	7
1.4 Significance of the dissertation study	3
1.5 Organization of the dissertation	3
CHAPTER 2: RATIONALE AND EMPIRICAL AND THEORETICAL SUPPORT FOR AIMS	C
2.1 Aim 1 rationale10	C
2.2 Aim 2 rationale18	3
Empirical evidence	3
Theoretical rationale	9

2.3 Aim 3 rationale	23
Defining constructs	23
Theoretical rationale	24
Empirical evidence	27
CHAPTER 3: STUDY DESIGN AND METHODS	28
3.1 Research strategy overview	28
3.2 Parent trial context and procedures	28
Parent trial aims	28
Recruitment and sampling	29
Data collection	31
Sample description	32
Ethical review	33
3.3 Measures	33
Dependent variables	33
Individual-level independent variables	34
Individual-level covariates	35
Camp-level covariates	36
Camp network moderators	36
3.4 Analytic strategy	38
Aim 1 analysis	
Aim 2 analysis	42
Aim 3 analysis	44
3.5 Sample size and power	46

CHAPTER 4: PAPER 1	
4.1 Background	
4.2 Methods	
Study context	
Sampling and data collection	50
Measures	50
Analysis	
Ethical Review	53
4.3 Results	53
Participant characteristics	53
Change in symptoms of anxiety and depression and sexua	Il behaviors54
Anxiety models	56
Depression models	57
Independent effects of anxiety and depression	57
4.4 Discussion	58
Limitations	60
Future directions	61
Conclusion	61
CHAPTER 5: PAPER 2	63
5.1 BACKGROUND	63
5.2 METHODS	
Study context	
Sampling and data collection	67

Measures
Analysis70
Ethical Review
5.3 RESULTS
Participant characteristics
Cross-sectional description and change in key variables74
Association between observed and perceived predictors76
Peer similarity in behaviors, anxiety, and depression7
Condom use models
Concurrency models
5.4 DISCUSSION
Limitations
Implications for future research and intervention8
Conclusion
CHAPTER 6: AIM 3 RESULTS
6.1 BACKGROUND
6.2 METHODS92
Study context
Sampling and data collection92
Measures93
Analysis
6.3 RESULTS
6.4 DISCUSSION

CHAPTER 7: DISCUSSION AND CONCLUSION
7.1 Summary of Findings100
7.2 Study Strengths
7.3 Study Limitations105
7.4 Implications for Future Research108
7.5 Intervention Implications113
APPENDIX A: HSCL-25 ITEMS
APPENDIX B: SEXUAL BEHAVIOR ITEMS118
APPENDIX C: SOCIAL NORM ITEMS
APPENDIX D: PERCEIVED SOCIAL COHESION ITEMS
REFERENCES

LIST OF TABLES

Table 1. Systematic review articles	15
Table 2. Systematic review results	16
Table 3. Baseline participant characteristics	33
Table 4. Aim 1 baseline participant characteristics	54
Table 5. Cross-sectional description of symptoms of anxiety and depression and sexual risk	56
Table 6. Condom use change	56
Table 7. Concurrency change	56
Table 8. Aim 1 model results	58
Table 9. Aim 2 baseline participant characteristics	74
Table 10. Aim 2 cross-sectional description of key variables	76
Table 11. Baseline correlations between observed and perceived norms	77
Table 12. Condom use peer norm interaction model results	79
Table 13. Concurrency peer norm interaction model results	82
Table 14. Aim 3 baseline description of hypothesized moderators	97
Table 15. Condom use – cohesion and centrality interaction model results	98
Table 16. Concurrency – cohesion and centrality interaction model results	98

LIST OF FIGURES

Figure 1. Conceptual model of hypotheses	3
Figure 2. Berkman's model	20
Figure 3. Transactional model of stress and coping	26
Figure 4. Box plot distributions of QAP correlations by camp	77
Figure 5. Interaction plots for condom use models	80

LIST OF ABBREVIATIONS

AIDS	Acquired immune deficiency syndrome
aOR	Adjusted odds ratio
САРІ	Computer assisted personal interviewing
CI	Confidence interval
EMA	Ecological momentary assessment
GIS	Geographic information system
GPS	Global positioning system
HIV	Human immunodeficiency virus
HSCL-25	Hopkins Symptom Checklist-25
ICC	Intra-class correlation
IRB	Institutional Review Board
N/No.	Number
OR	Odds ratio
PI	Principal Investigator
PLACE	Priorities for Local AIDS Control Efforts
PLHIV	People living with HIV
QAP	Quadratic assignment procedure
SD	Standard deviation
SES	Socioeconomic status
STI	Sexually transmitted infection
UNC	University of North Carolina
US	United States
USAID	United States Agency for International Development

CHAPTER 1: INTRODUCTION

1.1 Problem statement

Youth in Tanzania are disproportionately affected by HIV [1], and men are important targets in preventing HIV among Tanzanian youth [1]. Seven percent of the population of Tanzania's commercial capital, Dar es Salaam, is living with HIV [2], and youth are particularly vulnerable to infection with 40 percent of new infections in Tanzania occurring among 16-24 year olds [1]. An estimated 17 percent of females and 8 percent of males ages 15-19 in Tanzania are living with HIV [3]. Men are important targets in HIV prevention as they report higher rates than women of sexual risk behaviors including sexual partner concurrency and inconsistent condom use [1]. Half of sexually active male youth (ages 15-24) surveyed in Tanzania in 2010 did not use a condom at last intercourse [4]. Further, nearly one third of young men in Tanzania reported concurrent sexual relationships in 2010 [4], making condom use and concurrency important target risk behaviors in this population.

Anxiety and depression have been identified as important predictors of sexual risk [5-10], but we have little understanding of these relationships beyond cross-sectional associations, particularly in the sub-Saharan context. Cross-sectional associations have repeatedly been found between both anxiety and depression, and sexual risk behaviors (condom use and concurrency) [5-10], but we need more longitudinal evidence of these relationships to clarify the temporality and causality of these relationships. Further, the bulk of the evidence supporting these relationships comes from the United States and other high resource contexts [5, 8, 10-15]. Therefore, we need better evidence of the nature

of the relationship between mental health and sexual risk in the sub-Saharan context. The dissertation research serves to fill this gap in the literature.

We know little about the role that social networks play in shaping the relationship between mental health and sexual risk. Previous social network studies indicate that peers' mental health status affects an individual's own mental health [16, 17], and the same holds true for sexual risk behaviors [18]. However, we have little understanding of the potential influence of peers in patterns of mental health and sexual risk within the same network. Understanding peer influence on the relationship between mental health and sexual risk in young men's social networks will allow us to discern if there is something social or learned about sexual risk as a response to poor mental health. Lastly, we know little about how we may be able to intervene within networks to reduce sexual risk. The dissertation research aimed to address this gap in our knowledge by examining potential social network characteristics that buffer in the relationship between mental health and sexual risk behaviors.

1.2 Study purpose

The proposed research furthers our understanding of social and psychological determinants of HIV risk in a population of high risk young men by addressing three major gaps in the literature: a lack of evidence of the temporality of the relationship between mental health and sexual risk in this context, a particular lack of evidence of the connection between anxiety and sexual risk, and a lack of understanding of social network influences on the relationship between mental health and sexual risk.

1.3 Research approach

To address these gaps in the literature, I conducted a longitudinal study of the relationship between mental health (anxiety and depression) and sexual risk behaviors (condom use and sexual partner

concurrency). This study was completed using data from 1113 sexually active male participants taken from the baseline and one year follow-up assessments from the parent trial: *A Multilevel Intervention to Reduce HIV Risk Among Networks of Men in Tanzania* (PI: Suzanne Maman; 1R01MH098690-01). The dissertation Aims were completed using multilevel modeling and social network analysis.

Specific Aims



Figure 1. Conceptual model of hypotheses

The three Specific Aims of the dissertation research and their corresponding hypotheses were as follows:

Aim 1: Evaluate the longitudinal relationship between mental health (anxiety, depression) and sexual risk behaviors (condom use, concurrency) among young men.

Hypothesis 1: Young men experiencing increased symptoms of anxiety and depression from time 1 to time 2 will report less frequent condom use and be more likely to report engaging in sexual

partner concurrency at time 2, controlling for time 1 behaviors, than young men experiencing decreased symptoms of anxiety and depression.

Aim 2: Assess the level of social influence on young men's sexual risk behaviors, and the moderating role of social norms in the relationship between mental health and sexual risk behaviors within camp networks.

Hypothesis 2a: Young men's sexual risk behaviors will be more correlated with those of friends than with those of other members of their camp network.

Hypothesis 2b: The relationship between mental health and sexual risk will be moderated by peer sexual risk behavior norms such that the relationship will be stronger for young men exposed to riskier peer norms than for men exposed to less risky peer norms.

Aim 3: Test social network characteristics (centrality, social cohesion) as moderators of the relationship between mental health and sexual risk behaviors.

Hypothesis 3a: Young men's centrality within their camp social network will moderate the relationship between mental health and sexual risk behaviors such that the relationship will be weaker for men with higher centrality than for those with lower centrality.

Hypothesis 3b: Network social cohesion will moderate the relationship between mental health and sexual risk behaviors such that the relationship will be weaker for young men in camps with higher levels of social cohesion than for those in camps with lower levels of social cohesion.

Study context

The parent trial was conducted in Dar es Salaam, Tanzania. Tanzania is an east African country with a population of 45 million (2015 census) [19]. It is a low-income country with an estimated annual \$2,700 gross domestic product per capita (2014 estimate) and a growing economy (2014 growth rate estimated at 7%) [20]. Tanzania's population has a median age of 18 with a life expectancy of 61 years at birth [21]. Dar es Salaam is the nation's largest city and commercial capital, with a population 4.4 million comprising approximately 10% of the nation's population [19]. Located on the country's eastern coast on the Indian Ocean, it lost its position as the nation's capital to the more centrally located Dodoma in 1974 but remains the focus of much of the permanent central government bureaucracy. An estimated 70% of the city's population lives in informal settlements [22].

HIV/AIDS in Tanzania. Over 1,400,000 people in Tanzania are living with HIV. HIV/AIDS remains the leading cause of death in Tanzania [21], with approximately 80,000 AIDS-related deaths annually [23]. 80% of new HIV infections in Tanzania are estimated to be attributable to heterosexual transmission [23]. 5% of Tanzanian adults ages 15-49 are HIV-positive (6% of women and 4% of men) [1]. The proportion of people living with HIV (PLHIV) rises in successively higher age cohorts; the prevalence among those ages 20-24 is 3% (4% among women and 2% among men), 5% among those 25-29 (7% among women, 3% among men), and 8% among those 30-34 (9% among women, 7% among men) [1]. As in many settings, women are at greater risk of being HIV infected than men [1]. Other groups at heightened risk include men who have sex with men (22% prevalence), people who inject drugs (16% prevalence), and female sex workers (31% prevalence) [23]. Youth in Tanzania are at particular risk of HIV infection; 60% of new infections in Tanzania occur among 16-24 year olds [24], and an estimated 17% of females and 8% of males ages 15-19 are living with HIV in Tanzania [3]. The populations of the

country's urban areas also have a higher than average prevalence of HIV (7 % versus the national average of 5 %) [1]. The HIV prevalence in Dar es Salaam is estimated at 7% [1].

In 2010, 34% of health facilities in Tanzania offered HIV testing and counseling services and 13% offered antiretroviral therapy [1]. In 2013, 62% of women and 47% of men had ever been tested for HIV and received the results of their test [1]. Testing in ante- and perinatal care settings matches the rate among the general population of women (62%) [1]. Though nationally the majority of clients receiving testing services are women, in Dar es Salaam 76% of new clients are men [1]. Only 57 % of the country's estimated 1,400,000 PLHIV are enrolled in HIV care [1].

Mental health in Tanzania. Very little is known about the mental health of Tanzania's population. There are no estimates of the prevalence of common mental disorders including anxiety and depression to my knowledge. In one study of primary care attendees, the prevalence of depression and "mixed anxiety-depressive disorder" were 15% and 7%, respectively [25]. In another study of HIV-positive patients in rural Tanzania, 16% of patients presented symptoms of depression or mixed anxiety and depression, and 5% suffered from other anxiety disorders [26]. Further, there are very few treatment services available for people suffering from anxiety, depression, and other common mental disorders in Tanzania; per 100,000 population there are only 2 psychiatric nurses, 0.04 psychiatrists, and 0.005 practicing clinical psychologists [27].

Overview of the parent trial

The dissertation research was conducted within the chair's NIMH R01, *A Multilevel Intervention to Reduce HIV Risk Among Networks of Men in Tanzania* (PI: Suzanne Maman; 1R01MH098690-01), a cluster randomized trial aiming to determine whether men in social groups known as "camps" randomized to receive a microfinance and health leadership intervention have a lower incidence of

sexually transmitted infections and report perpetrating less physical and sexual violence against female sexual partners as compared to men belonging to camps randomized to a control condition.

Dr. Maman and the research team of the parent study identified social networks of young men occupying designated venues known as "camps" in Dar es Salaam where men socialize and engage in small scale enterprise [28]. Participants in the parent study were male members of 59 of these camps. Camps are stable social units; those included in the trial had been in existence for an average of eight years. Men typically belong to only one camp and pay membership fees to belong to that camp [28]. Camps are made up of entrepreneurial youth who frequent the venue for its supportive social environment [28]. Most men in the camps are not formally employed and spend several hours each day at their camp [28].

These camps provided unique access to social networks of high risk young men who are otherwise difficult to reach, as they are largely neither in school nor formally employed [28]. Complete longitudinal, sociocentric network data from 59 camps of 1249 young men were available from the parent trial, providing a unique opportunity to examine social network influences on young men's sexual risk. Men in these camps show increasing sexual risk behaviors over the teen years [29], and report high levels of anxiety and depression (19% and 21%, respectively based on common clinical cutoffs) [30].

Overview of analyses

The Aims were completed using multilevel modeling and social network analysis. In the analysis for all Aims, individual- and camp-level sources of variation in sexual risk behaviors were accounted for using multilevel models. In Aim 1 I assessed the longitudinal relationship between mental health and sexual risk. Evaluation of Aim 2 hypotheses was completed using quadratic assignment procedures to test the correlations between peers' reported sexual behaviors (hypothesis 2a), and extensions of the multilevel models from Aim 1 to test peer norms as moderators of the relationship between mental health and

sexual risk (hypothesis 2b). Finally, for Aim 3, I again expanded the models from Aim 1 to test the interactions between mental health variables and each hypothesized moderator of network cohesion and centrality. For all significant interactions found, I probed and interpreted the nature of the interaction.

1.4 Significance of the dissertation study

The dissertation research provides novel evidence of the potential causal relationship between mental health and sexual risk in a population of young men living in Dar es Salaam, and the extent to which this relationship is socially influenced. The sociocentric network data from 59 camps available from the parent trial presented the rare opportunity to explore the influence of social network peers on the relationship between mental health and sexual risk. The results of this research provide implications for the combination of sexual risk reduction and mental health promotion interventions in this population, as indicated in other settings [31]. The findings further provide evidence essential to the future development of interventions to reduce HIV risk among marginalized populations of young men, by providing understanding of potential ways to leverage the influence of social networks to address mental health as a risk factor for HIV transmission.

1.5 Organization of the dissertation

This dissertation has seven chapters. This first chapter serves as an introduction to the dissertation and provides an overview of the problem motivating the dissertation research, the purpose and significance of the study, the research approach, and a description of the study context. Chapter 2 provides the rationale for the dissertation Aims based upon theory and empirical evidence; Aims 1, 2, and 3 are presented with corresponding hypotheses. Chapter 3 includes a detailed description of the research

design and methods used in the study. Chapter 4 presents the results for Aim 1 on the association between mental health and sexual risk behaviors in manuscript form. Chapter 5 includes the results of Aim 2 on the moderating role of peer norms in the relationship between mental health and sexual risk, also in manuscript form. The results of Aim 3 on the moderating role of network cohesion and centrality in the relationship between mental health and sexual risk are presented in Chapter 6; these results were excluded from the second manuscript (Chapter 5) and are not intended for further publication. A discussion of the results across the three Aims is presented in Chapter 7, as well as a summary of the study strengths, limitations, and implications for future research and interventions.

CHAPTER 2: RATIONALE AND EMPIRICAL AND THEORETICAL SUPPORT FOR AIMS

2.1 Aim 1 rationale

Aim 1

Evaluate the longitudinal relationship between mental health (anxiety, depression) and sexual risk behaviors (condom use, concurrency) among young men.

Youth in Tanzania are at high risk for HIV infection [1], and young men are important targets in preventing HIV transmission as gender norms position them to control the terms and conditions of sexual relationships [32], and encourage them to engage in high risk sexual behaviors such as inconsistent condom use and sexual partner concurrency [1, 33-36]. To date, few studies have attempted to understand the unique predictors of risky sexual behaviors among young men in low resource settings such as Tanzania [37, 38]. A better understanding of predictors of sexual risk in young men in this context is needed to inform effective HIV prevention strategies.

Key predictors of HIV infection in sub-Saharan Africa include, but are not limited to, condom use and sexual partner concurrency [39]. Condom use has been shown to be effective in preventing the transmission of HIV for both men and women; lower levels of condom use for penetrative sex acts are associated with increased risk of HIV transmission [40]. Concurrent sexual partnerships (multiple coinciding sexual relationships) also present an important risk factor in the transmission of HIV and other STIs [41-43], by decreasing the time from an exposure to the next sexual contact, and increasing the likelihood of exposure to acute infection when HIV is the most transmissible [44-46]. These risk behaviors are of particular relevance for young men in Tanzania as the majority of the young men in Dar es Salaam report condom use at last sex, but this proportion declines from 92% among 15-year-olds to 70% among 19-year-olds [29], indicating a decreasing motivation for condom use over adolescence. Among the same population, the proportion reporting concurrent sexual partnerships increases with age, with 5% reporting concurrency at age 15, to 44% reporting concurrency by age 19 [29]. The low levels of condom use and high rates of concurrency found among men in Dar es Salaam are consistent with national studies in Tanzania; half of sexually active youth ages 15-24 surveyed in Tanzania in 2010 did not use a condom at last intercourse [4]. Further, nearly one third of young men in Tanzania reported concurrent partners in 2010 [4], making sexual concurrency an important target risk behavior in this population.

Anxiety and depression as predictors of sexual risk have been studied in high-income settings, primarily in adult populations. Mental health may influence sexual risk through multiple mechanisms including substance use [10], maladaptive coping mechanisms to deal with stress [47], and impaired decision making [48]. Primarily cross-sectional associations have been found between both anxiety and depression, and sexual risk behaviors (condom use and concurrency) [5-10], with a few previous studies having observed a predictive association between mental health and sexual risk in longitudinal studies [8, 9, 11, 12, 49].

A review of the literature on the relationship between mental health and sexual risk behaviors primarily yielded evidence from high income settings. To ensure an exhaustive understanding of studies from sub-Saharan Africa, I conducted a systematic review of the literature on the relationship between mental health (anxiety and depression) and sexual risk (condom use concurrency). To be included in the review studies had to have any measure of condom use, and/or any measure equivalent to sexual partner concurrency. For the purposes of the dissertation research, studies among HIV-positive only populations were excluded. The search yielded a total of 138 articles, with a final yield of 9 eligible

articles. A description of each article is presented in Table 1 and the results of these studies are presented in Table 2. Systematic review results

The majority of the studies included in the review either sampled exclusively undergraduate college students or were conducted among household-based samples of a given geographic area; there were few studies of high risk populations, with one study among female clients of drinking venues [50], and one study among men who have sex with men [51]. The latter study was the only one to focus exclusively on men. Moreover, all studies were cross-sectional, precluding the ability to draw temporal conclusions about any relationships seen. Given that young men in this context are at heightened risk of HIV infection [1], and the majority of cases of anxiety and depression begin by age 24 [52], it is particularly important to understand if there is indeed evidence in this population of increased HIV risk behaviors in relation to levels of anxiety and depression.

Only one study included anxiety as a predictor of sexual risk. Further, there were far fewer studies of concurrency than condom use; several studies which were excluded looked at number of sexual partners within a given time period, but very few looked at concurrency specifically. In general, there was poor evidence of the relationship between depression and condom use or concurrency. The one study that tested the relationship between anxiety and condom use found a significant association among male undergraduates [7]. Two population-based studies found a significant association between depression and sexual risk behavior, one association with condom use [53], and one association with concurrency [9].

Altogether, the studies provided mixed support for the relationship between mental health and sexual risk behaviors. There may have been methodologic reasons for which these relationships were not observed, contributing to the risk of type II error (failure to reject a false null hypothesis). In general, studies suffered from over-adjustment for covariates, and often included covariates with high likelihood of collinearity with anxiety or depression, such as posttraumatic stress disorder (PTSD) symptoms.

Further, some studies had relatively small sample sizes and thus may have not been powered to detect significant relationships between mental health predictors and sexual risk behaviors. Lastly, most studies of condom use utilized a dichotomous measure of consistent condom use, i.e. participants were classified as either always using condoms or not. This treatment of condom use fails to capture important differences in condom use, specifically whether or not a person uses condoms ever. The dissertation study contributes to this literature by capturing qualitatively important levels of condom use while minimizing over-sensitivity to biases in participant recall through the use of a three-level variable.

This review made it clear that more studies from sub-Saharan Africa evaluating the link between mental health and sexual risk behaviors were needed, with a particular need for greater evidence of the potential link between anxiety and sexual risk, and of the risk of sexual partner concurrency in relation to mental health. The dissertation research addresses these gaps in the literature. It further provides novel evidence in this setting of these relationships in a population primarily composed of men who have sex with women. Despite poor evidence among review articles supporting the relationships between depression, anxiety, and sexual risk behaviors among populations in sub-Saharan Africa to date, support for these relationships from high income settings indicated that the relationship in this setting may have been under-explored and merited further investigation. Specifically, I hypothesized:

Hypothesis 1: Young men experiencing increased symptoms of anxiety and depression from time 1 to time 2 will report less frequent condom use and be more likely to report engaging in sexual partner concurrency at time 2, controlling for time 1 behaviors, than young men experiencing decreased symptoms of anxiety and depression.

The results of this aim provide much needed longitudinal evidence of the nature of the relationship between mental health and sexual risk in the sub-Saharan context. This contribution to the understanding of a potential causal relationship between mental health and sexual risk behaviors has implications for the importance of integrating mental health promotion and HIV prevention interventions targeting the growing population of male youth in sub-Saharan urban areas.

Authors and year	Title	Location	Population	Study design	Sample size by sex
Agardgh et al. 2012 [7]	Youth, Sexual Risk-Taking Behavior, and Mental Health: a Study of University Students in Uganda	Mbarara, Uganda	Undergraduate university students	Cross- sectional	633 men; 329 women
Asante & Andoh- Arthur 2014 [54]	Prevalence and determinants of depressive symptoms among university students in Ghana	Accra, Ghana	Undergraduate university students	Cross- sectional	132 men; 138 women
Gupta et al. 2010 [55]	Depression and HIV in Botswana: A Population- Based Study on Gender- Specific Socioeconomic and Behavioral Correlates	5 Districts in Botswana	Household-based probability sample of 18–49 year-olds	Cross- sectional	580 men; 597 women
Lundberg et al. 2011 [9]	Poor mental health and sexual risk behaviours in Uganda: A cross-sectional population-based study	Kampala and Mbarara, Uganda	Household-based probability sample	Cross- sectional	334 men; 312 women
Otieno et al. 2015 [56]	Risky HIV sexual behaviour and depression among University of Nairobi students	Nairobi, Kenya	Undergraduate university students	Cross- sectional	525 men; 365 women
Peltzer et al. 2013 [57]	Mental health, childhood abuse and HIV sexual risk behaviour among university students in Ivory Coast	Cocody, Ivory Coast	Undergraduate university students	Cross- sectional	412 men; 412 women
Pitpitan et al. 2012 [50]	Co-occurring Psychosocial Problems and HIV Risk Among Women Attending Drinking Venues in a South African Township: A Syndemic Approach	Capetown area township, South Africa	Female clients of drinking venues	Cross- sectional	560 women
Smit et al. 2006 [53]	Mental health and sexual risk behaviours in a South African township: A community-based cross- sectional study	Capetown area township, South Africa	Population- based sample: Participants in larger HIV risk study	Cross- sectional	252 men; 357 women
Stahlman et al. 2015 [51]	Depression and Social Stigma Among MSM in Lesotho: Implications for HIV and Sexually Transmitted Infection Prevention	Maseru and Maputsoe, Lesotho	Men who have sex with men recruited	Cross- sectional	530 men

Authors/ year	Mental Health Constructs	Mental Health Measures	Sexual Risk Measures	Results: Men and women	Results: Men only	Results: Women only
Agardgh et al. 2011	Anxiety, Depression	HSCL-25 (median score as cutoff)	Inconsistent condom use (at last sex)		Anxiety but not depression significantly associated with inconsistent condom use Anxiety AOR: 1.9 (1.1– 3.6) Depression AOR: 1.6 (0.9-3.0)	No significant association Anxiety AOR: 2.1 (0.9– 4.7) Depression AOR: 1.1 (0.5-2.5)
Asante & Andoh- Arthur 2014	Depression	CES-D 10 (cutoff ≥ 10)	- Concurrency (multiple sexual partners) - Inconsistent condom use	No significant associations Concurrency AOR 0.67(0.32–1.42) Condom use AOR: 0.74(0.36–1.53)		
Gupta et al. 2010	Depression	HSCL-25 depression subscale (cutoff ≥ 1.75 mean score)	Inconsistent condom use (past 12 months)		No significant association AOR: 1.23 (0.78–1.95)	No significant association AOR: 0.82 (0.52–1.31)
Lundberg et al. 2011	Depression	HSCL-25, depression sub- scale (cutoff ≥ 16)	 Concurrency (0-1 vs. 2 or more current sexual partners) Consistent condom use (always vs. sometimes/ never) 		No significant association Concurrency AOR: 1.52 (0.60-3.86) Condom use AOR: 2.71 (0.72-10.20)	Depression significantly associated with concurrency Concurrency AOR: 3.92 (1.25-12.27) Condom use AOR: 2.12 (0.79-5.68)
Otieno et al. 2015	Depression	CES-D 10 (10–14 moderate; ≥15 severe)	Inconsistent condom use (past 3 months with primary partner)	No significant association Moderate depression AOR: 0.92 (0.61–1.39) Severe depression AOR: 1.18 (0.47–2.97)		
Peltzer et al. 2013	Depression	CES-D 10 (cut-off ≥10)	Inconsistent condom (past 3 months)		No significant association	No significant association

Table 2. Systematic review results

Authors/ year	Mental Health Constructs	Mental Health Measures	Sexual Risk Measures	Results: Men and women	Results: Men only	Results: Women only
					Condom use OR = 0.69 (0.35–1.36)	Condom use OR: 1.04 (0.54–2.01)
Pitpitan et al. 2012	Depression	CES-D (cutoff ≥ 16)	% unprotected sex acts (past 4 months)			No significant association AOR = 1.30 (0.67, 2.50)
Smit et al. 2006	Depression	CES-D (cutoff ≥ 20)	No condom use (past 6 months)	Depression associated with higher levels of condom use AOR: 5.32 (2.74-10.29); p<.001		
Stahlman et al. 2015	Depression	Patient Health Questionnaire (cutoff ≥ 10)	Unprotected sex (past 30 days)		No significant association AOR = 0.72 (0.38-1.39)	

2.2 Aim 2 rationale

Aim 2

Assess the level of social influence on young men's sexual risk behaviors, and the moderating role of social norms in the relationship between mental health and sexual risk behaviors within camp networks.

Empirical evidence

Previous studies have shown that peers' mental health status affects an individual's own mental health [16, 17] and sexual risk behaviors [18], but we have little understanding of the potential influence of peers in patterns of mental health and sexual risk within the same network. Understanding such an influence on the relationship between mental health and sexual risk in young men's social networks would allow us to discern if there is something social or learned about sexual risk as a response to poor mental health. Experts have called for combined sexual risk reduction and mental health promotion interventions [31], and the results of the dissertation research provide essential evidence to determine the appropriateness of such interventions in this setting.

Past studies indicate that individuals' sexual behaviors, including partner concurrency [58] and condom use [59-62], reflect their peers' behavior. Specifically, these studies found that youth who perceived higher levels of concurrency or lower levels of condom use among their peers were more likely to display concurrency or lower levels of condom use themselves. Peer influence and similarity in sexual behavior has been less frequently studied from a social network perspective than mental health, indicating a need for further research in this area. Though the influence of mental health and peer norms have been studied separately as predictors of sexual risk, to my knowledge there were no studies prior to the dissertation study assessing the interaction between individual-level mental health and peer norms in predicting sexual risk behaviors.

Theoretical rationale

Berkman's model. Development of hypotheses pertaining to social network influence on the relationship between mental health and sexual risk was guided by Berkman's model of social network influence on health behaviors [63]. This model offers a multilevel framework for understanding the influence of social network structure and ties on individual behaviors. Berkman posits that both the structure of a social network and qualities of the ties between individuals within the network shape interpersonal social processes including social support, social influence, social engagement and role formation, and access to goods and resources; in turn, these social processes influence individual behaviors [63]. I used this model to guide my thinking and review of the literature on the potential role of social networks in shaping young men's sexual risk behaviors in interaction with their mental health. Social influence emerged as the most salient psychosocial mechanism through which the form and function of social networks might shape the behaviors in question.



Figure 2. Berkman et al., 2000

Social influence

Focusing on the mechanism of social influence from Berkman's model, the hypotheses related to peer influence on individual sexual risk behaviors, and the relationship between mental health and sexual risk were guided by additional theories specific to social influence. Social networks shape social influence by providing structural opportunities for interactions between network members [64, 65]. Network peers who are structurally proximal, or who have direct connections, have the ability to directly influence one another. In this way peers develop shared attitudes, perceptions, and behaviors with structurally proximal peers in social networks [65]. Network members not only have the ability to influence one another through direct contact, but can also indirectly influence peers to whom they are not directly tied [66]. An individual may watch and model a friend's behavior, such as partner concurrency, and may in turn influence other friends, thereby mobilizing a process of behavioral contagion throughout a whole network [67].

In this Aim, I focused on direct interactions as the basis of peer influence. I hypothesized that network members who have strong relationships (i.e. friendships) interact frequently and by consequence develop similar behaviors. Young men's interactions with friends in their network will create opportunities to directly observe and model how their friends think, feel, and behave. Following Marsden and Friedkin's hypothesis that the more proximal two individuals are in a network, the more influence these actors will have on one another [34], I hypothesized:

Hypothesis 2a: Young men's sexual risk behaviors will be more correlated with those of friends than with those of other members of their camp network.

I further explored the interaction between individual mental health and peer risk behavior norms in shaping young men's sexual risk behaviors. Peer influence not only acts independently to shape behavior but peers also serve as a reference groups which individuals look to in behavioral decision-making [68]. In Aim 1, I hypothesized that increases in symptoms of anxiety or depression would increase the likelihood of engagement in sexual risk behaviors. In Aim 2, I further hypothesized that more normative risk behaviors among direct peers will serve to magnify this individual-level risk. In other words, men who experience increased levels of anxiety or depression may have a higher inclination toward risk behaviors, and the influence of riskier norms among their friends may increase the likelihood that this inclination manifests in actual risk behaviors. Specifically, I tested the following hypothesis:
Hypothesis 2b: The relationship between mental health and sexual risk will be moderated by peer sexual risk behavior norms such that the relationship will be stronger for young men exposed to riskier peer norms than for men exposed to less risky peer norms.

2.3 Aim 3 rationale

Aim 3

Test social network characteristics (centrality, social cohesion) as moderators of the relationship between mental health and sexual risk behaviors.

Defining constructs

Centrality is a measure of an individual's position within a given social network; measures of centrality capture the closeness of an individual to their peers in a network [69, 70]. There are many operationalizations of centrality; the one used in this study is *in-degree centrality*, or the number of friendship nominations that an individual receives from other members of his network [69]. Borgatti describes degree centrality as relating to an individual's ability to contribute to the spread of information in a network [69]. He also describes degree centrality as reflecting the volume of ties an individual has rather than the relative distance (in terms of ties), or closeness, to other members in the network [69]. For the purposes of this Aim I selected in-degree centrality as measure to capture the extent to which men are to be supported and influenced by their closest peers. This type of centrality can be thought of as a measure of popularity, as the more peers who nominate that individual as a friend, the more prominent or popular they can be thought of as being in their network [71].

Social cohesion has a wide range of definitions across disciplines, but roughly relates to the connectedness and togetherness of a defined social group, either perceived or observed [72]. Among the earliest to discuss this construct, Durkheim observed the relationship between social integration and suicide rates, finding higher levels of suicide in less integrated societies [73]. In contemporary research, many choose to place an emphasis on the importance of perceived social cohesion, as seen in the research of Bollen and Hoyle who proposed that individuals' perceptions of their group's cohesion influences both the individual and the group sense of belonging and morale [74]. In contrast, other researchers have focused on structural rather than perceived measures of cohesion including Moody

and White [75]. Moody and White's measure of *structural cohesion* captures a dimension of *relational* cohesion in a network defined as the "extent that the social relations of [a network's] members hold it together" [75]. Operationally, this concept translates to the extent to which the connectedness of a network depends on a few or many actors in that network, or the number of independent paths between any two given individuals in a network [75]. In other words, a network is more cohesive if the relational togetherness of the group does not depend on only a few people, and this dependence is decreased by greater connections between individuals in the network. Taking the example of a group of friends, if one person is the primary "hub" (strongly tied to all/most others) and everyone else in the group is primarily friends with the hub and only friends by association with everyone else, that group is not very cohesive given that their cohesion as a group is dependent upon a single person. On the other hand, if there a many independent strong friendships within a friend group, if one person leaves the group is not likely to fall apart, for there are many ties that bind the group together. To understand the influence of both structural and perceived cohesion, I assessed the moderating role of both men's reports of cohesion in their network, and structural cohesion measured as discussed by Moody and White using sociocentric data from each camp network.

Theoretical rationale

Hypotheses related to social network moderators of the relationship between mental health and sexual risk were again guided by Berkman's model, specifically by the indication of this model that social network structure shapes social support via increased opportunities for social interaction. Macro-level network structure and an individuals' connections within the network will shape exposure to micro-level mechanisms, including social support [76]. Specifically, I posited that greater social cohesion increases opportunities for social support. Centrality reflects an individual's connectedness to other members in the network, and greater centrality increases the individual's ability to receive social support from

others in the network [77]. As with cohesion, I hypothesized that the higher an individual's centrality within the network, the more opportunities they have for social support. Both cohesion (at the group level) and centrality (at the individual level) shape opportunities to interact with peers, who can provide social support in the face of emotional distress.

The ability of social support to reduce the likelihood of risk behaviors is supported by Cohen's stress buffering theory, which indicates that social support can operate to intervene between the experience of stress (in this case, the experience of anxiety and/or depression) and maladaptive coping behaviors (such as sexual risk behaviors) by reducing the effect of the stressor or by affecting the psychological processes which occur in response to the stressor [78]. Lazarus and Folkman's Transactional Model of Stress and Coping provides further support for social support's role in promoting positive rather than negative coping outcomes [79], again indicating network social support as a potential moderator between stress related to anxiety and depression and maladaptive coping outcomes (sexual risk behaviors). Lazarus and Folkman posited that social support buffers the relationship between stressors and health behaviors by acting upon coping efforts (actions and processes to deal with the effect of stress), including problem management and emotional regulation (see Figure 3. Transactional model of stress and copingbelow) [79].



Figure 3. Transactional model of stress and coping

Rather than testing the role of social support as buffer of the relationship between mental health and sexual risk directly, I aimed instead to test the buffering effect of network position (centrality) and connectivity (cohesion) on this relationship as measures of the structural opportunity for social support. While there is much interest and previous research on the health promoting effects of social support, a functional aspect of social relationships, little attention has been paid to the health impact of the structure of social relations. While I highlight social support as an explanatory mechanism through which social structure functions to affect health, I was primarily interested in network structure as a novel point of investigation and potential future public health intervention. Guided by the above theoretical logic, I hypothesized that higher levels of social network cohesion and centrality would buffer the association between mental health and sexual risk. Specifically, I hypothesized:

Hypothesis 3a: Young men's centrality within their camp social network will moderate the relationship between mental health and sexual risk behaviors such that the relationship will be weaker for men with higher centrality than for those with lower centrality.

Hypothesis 3b: Network social cohesion will moderate the relationship between mental health and sexual risk behaviors such that the relationship will be weaker for young men in camps with higher levels of social cohesion than for those in camps with lower levels of social cohesion.

Empirical evidence

Prior to the dissertation research, there were no studies to my knowledge evaluating these hypothesized relationships, nor the direct relationship between network centrality or cohesion and sexual risk behaviors. There is evidence in the literature, though limited, that network centrality predicts receipt of social support [80], which in turn is related to sexual risk behaviors [81, 82]. This evidence further supports social support as the mechanism through which network cohesion and centrality may work to buffer the relationship between mental health and sexual risk behaviors. It also further indicates a need to better understand the effects of network structure and position on this and other stress and coping relationships.

CHAPTER 3: STUDY DESIGN AND METHODS

3.1 Research strategy overview

The dissertation research was completed using two waves of data from the chair's NIMH R01, *A Multilevel Intervention to Reduce HIV Risk Among Networks of Men in Tanzania* (PI: Suzanne Maman; 1R01MH098690-01). All data were taken from the behavioral and social network questionnaires from this trial. The analyses for all three aims were conducted in SAS v 9.4 using multilevel modeling and quadratic assignment procedures. Social network measures used in Aim 3 were calculated using UCINET 6.

3.2 Parent trial context and procedures

Parent trial aims

The parent trial is a cluster randomized controlled trial assessing the efficacy of a combined microcredit and health leadership intervention to reduce HIV risk and intimate partner violence. The first aim of the parent study is to enumerate and characterize camps in four wards of Dar es Salaam where young men at high risk for HIV acquisition and transmission socialize, in order to generate a pool of camps for inclusion in the larger trial. The second aim is to conduct a cluster randomized trial to determine whether men in camps randomized to receive a microfinance and health leadership intervention have a lower incidence of sexually transmitted infections and report perpetrating less physical and sexual violence against female sexual partners as compared to men belonging to camps that are randomized to a control condition.

Recruitment and sampling

The parent study team identified camps for inclusion using an adaptation of PLACE (Priorities for Local AIDS Control Efforts) methodology, a rapid assessment designed to locate venues where people meet new sexual partners through community informant and camp verification interviews. Approximately two community informants per 1,000 people in each of the four participating study wards were recruited for interviews, or 489 individuals completing a total of 619 verification interviews. A list of potential camps in the four study wards, and directions to these camps were documented through these interviews. A total of 303 camps were verified and all camps were mapped with GIS software. To be eligible for inclusion in the trial, camps had to have between 20 and 80 members, have been in existence for at least one year prior to the baseline assessment, and report no violent incidents with weapons in the past 6 months. A total of 303 camps were verified, of which 205 were eligible. Of the 303 verified camps, 98 camps did not meet the parent study's inclusion criteria (6 camps had previously been included in a pilot study and were ineligible for this reason, 51 camps had fewer than 20 members, 16 camps had more than 80 members, 4 camps had not been existence for at least 1 year, 23 camps reported that a weapon had been used in a fight in the last year, and 9 camps were removed because the research assistants felt unsafe during the preliminary data collection).

GIS was used to map the coordinates of each camp and to group geographically contiguous camps into 93 clusters prior to random selection of 60 study camps. In the event that multiple camps from the same cluster were selected for inclusion in the trial, these camps were assigned to the same condition to minimize contamination. All camps identified through community informant interviews were visited for camp verification interviews to confirm camp existence and current operation, record GPS coordinates, and record characteristics of the camp through interviews with camp leaders. Camps included in the sampling frame were those confirmed to exist and be in operation, for which GPS coordinates were recorded, and for which at least half of the camp verification interview questions were

completed. Of the 205 eligible camps identified, 60 camps were randomly selected for inclusion in the study. One camp was later excluded from the trial during midpoint data collection when its members reported to the study team that they were not a legitimate camp. Data from the remaining 59 camps were used in this dissertation.

All members of camps selected into the trial were preliminarily eligible for inclusion in the study. Following random selection, camp leaders were given a camp roster template to complete. The camp leaders were responsible for recording the name of every camp member as well as their birthdate, gender, phone numbers, length of time as camp member, whether or not the member intended to stay in the city for the next 30 months, and whether or not members were willing to give contact information for two friends or family members. To be eligible for the parent study at baseline, participants had to: 1) be a registered camp member for at least three months; 2) plan on residing in Dar es Salaam for the next 30 months; 3) be 15 years or older; 4) visit the camp at least once per week; and 5) be willing to provide contact information for themselves and two family members or friends. Those members listed on the roster as being younger than 15 or having not been a member for at least three months were deemed ineligible. All other members were deemed preliminarily eligible and were contacted for an interview to further assess their eligibility. Participants who were unwilling to provide locator information or were unable to participate due to psychological disturbance, cognitive impairment, or threatening behavior were excluded from the quantitative assessment. All eligible members were offered participation in the trial and asked to complete the baseline survey.

Of 1,948 camp members approached for participation, 1,500 completed the baseline survey. This level of non-response is primarily accounted for by various sources of ineligibility and inability to participate in the parent study. 284 camp members could not be reached for enrollment either because the study team was unable to contact them (101, 5.2%), or because the study team was unable to schedule an appointment with them (183, 9.4%). A further 112 (5.8%) individuals did not meet the

study's eligibility criteria, and an additional 49 (2.5%) refused to participate. Soon after baseline data collection, camp members from one camp (n=9) requested to be removed from the study because they were not a real camp and came together only for the purpose of being in the trial. This camp was removed from the study, resulting in a final sample of 1,491 individuals within 59 camps. The final response rate among potentially eligible participants (n=1,948) was 76.5% after the removal of this camp. The camps enrolled in the parent trial had an average of 33 members, ranging in size from 20 members to 77 members (SD = 12). The mean number of baseline survey participants per camp was 25 (range 7-66), for an average camp-level response rate of 78% (range 36-100%).

Written informed consent was required for participation in the community informant and camp verification interviews. Potential interviewees were read a script outlining the purpose of the interviews and given the opportunity to ask questions or clarifications. Participants consented to the interview by signing and writing the date on the form. Potential participants were excluded from participation if they were unable to provide informed consent. At the baseline behavioral assessment participants were asked to provide written consent to participate in the trial. Participants were informed that they would be re-contacted to participate in follow-up behavioral assessments. Upon providing written consent to participate in the study, the participant provided contact information. In all study data, reports, and study data collection, process, and administrative forms participants were identified by a coded number only to maintain confidentiality. All study data were stored separately from study records that contained participate names or other personal identifiers.

Data collection

After random selection and informed consent, 1249 male participants and 242 female participants completed the baseline assessment starting in Fall 2013, and a follow up assessment 12 months after the launch of the intervention. The baseline survey was administered by trained Tanzanian interviewers

in Swahili using computer assisted personal interviewing (CAPI) from October 2013 through March 2014. The follow-up survey was administered using the same methods from March 2015 through September 2015. The behavioral assessment included questions about the following topics: demographics, hope, future orientation, camp social cohesion, camp collective efficacy, anxiety symptoms, depression symptoms, gender equity, conflict tactics, HIV stigma, alcohol and drug use, sexual relationships and risk behaviors, attitudes towards HIV risk behaviors, sexual networks, social networks, work, and expenditures. Participants also completed a social network questionnaire in which they were offered a list of all of the members of their camp and asked to: a) state whether they knew each person; and b) characterize their relationship with each known person as an acquaintance, a friend, or someone they didn't get along with. The programming of the baseline assessment on tablet devices streamlined all skip patterns, and interviewers were intensively trained using both paper and electronic versions of the assessment tools to ensure their understanding of the logic built into the questionnaire. The interviews lasted approximately 60-90 minutes.

All data collected were subject to strict quality assurance measures at several levels. At the field level, team leaders in each field office had the ability to review data on the tablets for completeness before transmitting the data for processing by the central data manager, based in Delhi, India. Additional quality control measures were in place as the data was processed. I conducted my statistical analyses using a dataset with no individually identifying information.

Sample description

1249 men were interviewed at baseline, of whom 1113 reported being sexually active. All analyses were carried out among these 1113 sexually active men. Sexually active men interviewed at baseline had an average age of 27 years (range: 15 to 59). Over half had a primary school education or less (59%), nearly

a third had graduated from secondary school (31%), and the remaining 11% had some secondary school but had not graduated. A quarter of the men had ever been married (25%) and 38% had children.

bie 5. baseline i articipant enalacteristics (n=1115)					
	Total				
Age (y)***	26.8 ± 7.1				
Currently in school	97 (8.7%)				
Education level (ref = less than primary school completed)					
Primary school or less	652 (58.7%)				
Some secondary school	116 (10.5%)				
Secondary school completed or greater	342 (30.8%)				
Socioeconomic status					
Lowest	291 (26.2%)				
Middle	435 (39.1%)				
Highest	386 (34.7%)				
Ever married	277 (25.0%)				
Has children	423 (38.0%)				

Table 3. Baseline Participant Characteristics (n=1113)^a

^aData are expressed as No.(%) or Mean ± SD

Ethical review

The parent study was approved by the ethical review committees at the University of North Carolina at Chapel Hill and Muhimbili University of Health and Allied Sciences in Dar es Salaam, Tanzania. Individual written informed consent was obtained from all study participants. Secondary data analysis of the parent trial data for the dissertation research was approved by the UNC IRB.

3.3 Measures

Dependent variables

Condom use was measured through self-report. For each of their three most recent sexual partners, men reported how many times they engaged in sex with these partners over the most recent month of

the relationship, and how many of these times they used a condom. I calculated men's proportion of condom use by dividing the number of times condoms were used by the total number of sex acts. Using these proportions, I coded observations into one of three categories: "never use" (0% use), "some use" (greater than 0%, less than 100%), or "always use" (100%). Because of the possibility of some degree of recall bias over the course of the three most recent relationships, use of a continuous variable of condom use is not indicated as each unit increase would not necessarily indicate an important difference in condom use. Dichotomizing condom use, however, is not recommended by Noar et al. [83], thus use of a three-level categorical condom use variable preserves information in differences in use while reducing the influence of recall bias.

Sexual concurrency was evaluated by self-report of any overlapping sexual partnerships for an individual's past 3 partners. Participants were asked to enumerate their three most recent sexual relationships and to report if they had sex with anyone else during any of these partnerships. Participants reporting any instance of simultaneous sexual relationships either currently or in the past 12 months were coded as reporting concurrency. This measure was developed following best practices recently recommended by USAID [84].

Individual-level independent variables

Symptoms of anxiety and depression were measured using the Hopkins Symptom Checklist-25 (HSCL-25) [85], specifically a version of the scale previously translated into Kiswahili and tested in Tanzania [86]. This measure inventories symptoms of anxiety and depression in 25 items (10 related to anxiety, 15 to depression) rated on a 4-point Likert-type scale. The scale has been previously validated in Tanzania [86, 87]. Anxiety scores were calculated by taking the mean of the 10 anxiety-related items in the HSCL-25. In the sample the anxiety subscale showed good internal consistency at baseline (Cronbach's alpha = 0.94). Depression scores were calculated by taking the mean of the 15 depression-

related items in the HSCL-25. At baseline the depression subscale also showed good internal consistency (a = 0.91). Where depression and anxiety are described dichotomously, the recommended cutoff of 1.75 was used for clinical caseness [88]. This cutoff has been used in other studies using the HSCL-25 in sub-Saharan Africa [55, 89]. There is some indication that 1.67 may be a better cutoff for men [90], but I elected to use the more conservative cutoff of 1.75.

Individual-level covariates

Age Participants reported both their age in years and their date of birth. Participant age at baseline was calculated based on reported date of birth. This measure was used as the default value of the age variable with the exception of cases where the birthdate was not reported, in which case the reported age in years was used.

Education Participants reported the highest level of education they had completed by baseline based on 8 possible categories (no education, Standard 4 or less, Standard 5-7, Form 1, Form 2, Form 3, Form 4, Greater than Form 4). For analysis, education responses were collapsed into three categories: *Primary school or less* (responses: no education, Standard 4 or less, Standard 5-7); *Some secondary school* (responses: Form 1, Form 2, Form 3); or *Secondary school completed or greater* (responses: Form 4, Greater than Form 4).

Economic status was evaluated through an asset index, the Filmer Pritchett Wealth Index [91]. Participants were asked to indicate which of 10 possible assets they owned (e.g. cellphone, television) [92], and a composite score was created by weighting each asset by its factor loading on the first component in a principle components analysis and then placing individuals on a continuous scale of relative wealth [91]. The composite score for each participant was categorized into terciles based on the entire sample of men and women in our baseline dataset (the lowest 33% of participants were classified as "lowest SES," the highest 33% were classified as "highest SES," and the remainder were classified as "middle SES").

Marital status was evaluated through a single yes/no item in which men were asked if they had ever been married.

Camp-level covariates

Treatment condition. Each camp was randomized to either treatment or control condition in the parent trial. The treatment condition, which consisted of access to microfinance loans and peer health leadership training, was designed to affect condom use and sexual partner concurrency, thus it was included as important covariate to account for design effect.

Camp network moderators

Peer behavioral norm scores were calculated for each participant based on their friends' reported behaviors as well as their own reported perceptions of their friends' behaviors and views of condom use and concurrency. Four types of behavioral norms were measured for both condom use and concurrency: friend-reported behaviors (observed descriptive norms), perceived descriptive norms, perceived injunctive norms, and friend encouragement of the behavior. Men were asked to name their three closest friends in their camp by selecting names from a list of all camp members. The name of each friend was associated with a unique identifier which was used to link this nomination record to the friend's reported behavior according to their own completed questionnaire. Taking the reported behavior of all three friends for each individual, I created an average percent condom use and a percent of friends reporting concurrency measure for each participant. To assess perceived descriptive norms, for condom use men were asked for each of these three closest friends, *"Do you think Friend X uses condoms all the time?"* For injunctive norms men were asked *"Do you think Friend X thinks that he*

should be using condoms all the time?" Finally, to assess friends' encouragement of behaviors men were asked "Has Friend X encouraged you to use condoms all the time?" Men were asked analogous questions for partner concurrency and provided yes/no answers. I created a measure of the proportion of friends for which the respondent answered "yes" for each item.

Structural social cohesion measures for each camp were calculated as described by Moody and White [75]. Each camp network received a social cohesion score. Structural social cohesion measures for each camp specifically captured the construct of structural cohesion, or the minimum number of nodes who, if removed, would disconnect the network, using methods describes by Moody and White [75]. Each network received a structural social cohesion score calculated as the average pairwise connectivity, or the mean of the total number of possible independent paths between all pairs of any two given individuals in the network [75]. I calculated these scores in UCINET 6 [93]. First, I created matrices of pairwise connectivity scores for each dyad by camp. I then converted these matrices to pairwise lists of connectivity values for each camp and used mean of these values as the structural cohesion value for each camp to test hypothesis 3b.

Perceived social cohesion was measured using an adapted scale for social cohesion developed by Sampson, Raudenbush, and Earls [94]. Participants responded to statements related to cohesion of camp members on a 4-point Likert-type scale ranging from 1 = strongly disagree to 4 = strongly agree. Statements included "*people in my camp are willing to help each other*," and "*the members of my camp share the same values*."

Centrality can be thought of as the structural importance of a given individual within a network and can reflect the individual's ability to influence and be influenced by others in the network. Using a roster list including all members of each camp, men were asked to nominate their friends belonging to their camp by selecting their names from this list. Centrality measures, specifically in-degree centrality, were calculated as the proportion of people in the individual's camp network who nominated the individual as

a friend [95]. In-degree centrality was calculated within each network for each actor using UCINET 6 [93]. Standardized centrality scores were calculated as follows: **Centrality** = $InTie_{ij}/(n_j-1)$, where $InTie_{ij}$ is the number of ties received by individual *i* within camp *j*, and n_i is the total number of actors in camp *j*.

3.4 Analytic strategy

Analysis for the completion of the proposed aims included multilevel modeling and social network analysis. In all aims individual- and camp-level sources of variation in sexual risk behaviors were accounted for using multilevel models, given the nested data structure and the fact that I hypothesized both individual and camp-level determinants of sexual risk.

Descriptive analysis. I began analysis by describing the means and proportions of the individual-level and camp-level independent variables, dependent variables, and covariates in the sample. I created histograms of the continuous predictors of anxiety and depression, and described the means of these continuous predictors and proportions of the categorical outcomes by camp to further assess the level of variance in these key variables [96].

Missing data. To address missing data from the behavioral survey (primarily due to attrition), I applied sequential multiple imputation using the fully conditional specification in *proc MI*. Forty imputations were created using a linear regression specification for continuous variables and a logistic regression specification for categorical variables. To account for missing network ties in the quadratic assignment procedure discussed below, I conducted imputation by reconstruction as recommended by Huisman [97]. Specifically, I used random imputation of ties proportional to the observed network density (i.e., the probability of a tie is equal to the observed density of each network), as described by Huisman as an extension of Stork and Richard's method of reconstruction [97].

Aim 1 analysis

Aim 1, Hypothesis 1

Young men experiencing increased symptoms of anxiety and depression from time 1 to time 2 will report less frequent condom use and be more likely to report engaging in sexual partner concurrency at time 2, controlling for time 1 behaviors, than young men experiencing decreased symptoms of anxiety and depression.

To test this hypothesis, I fit multilevel models and regressed time two sexual risk behaviors on time one sexual risk behaviors, change score measures of mental health (time 2 – time 1), and baseline values of the covariates listed above. Multilevel modeling was used in all aims to adjust for the nested data structure by camp. I compared the variance from this model to that of the null model to assess the proportion of the variance in sexual risk behaviors explained by level-one predictors. The formulas for the respective null models for condom use and concurrency are as follows:

Level 1:
$$CumuLogCondUse_{ij} = \beta_{0j} + r_{ij}$$

Level 2: $\beta_{0j} = \gamma_{00} + u_{0j}$
 $r_{ij} \sim N (0, \sigma^2)$
 $u_{0j} \sim N (0, \tau_{00})$

Level 1:
$$LogCncr_{ij} = \beta_{0j} + r_{ij}$$

Level 2: $\beta_{0j} = \gamma_{00} + u_{0j}$
 $u_{0j} \sim N(0, \sigma^2)$

The intraclass correlation (ICC) was calculated as the within group variance (σ^2), over the total variance ($\sigma^2 + \tau_{00}^2$). The computed ICCs specified the proportion of the total variance in each sexual risk behavior which could be attributed to camp membership [98]. Using this ICC and estimates of within and between group variance from the null model as a point of comparison of the proportion of the variance explained by subsequent models, I continued my analysis by testing and comparing the fit of linear and non-linear models, as indicated by previous evidence of a non-linear relationship between depression and sexual risk [13-15].

Grand mean centering. Following imputation, I grand mean centered all continuous variables to be included in Aim 1 analyses, including anxiety and depression. Because baseline ICCs for these variables were low (0.046 and 0.047 for depression and anxiety, respectively), and as level 1 variation was of primary interest for this aim, continuous variables were centered using the grand mean rather than the camp mean.

Assessing collinearity. Because of the frequent comorbidity between anxiety and depression, I was concerned about a possible level of collinearity between measurements of their symptoms (the correlation between symptoms of anxiety and depression was 0.76 at baseline and 0.74 at the follow-up) which would limit the ability to interpret the independent effects of anxiety and depression when estimated simultaneously. For this reason, I tested three models for each sexual behavior dependent variable (for a total of six models): 1) change in anxiety symptoms only; 2) change in depression symptoms only; and 3) change in both anxiety and depression symptoms. I compared the parameter estimates and confidence intervals of models 1 and 2 to those in model 3 to assess the effects of the two correlated primary predictors.

Linear models. All models were fit using *proc glimmix*, quadrature estimation (using 15 quadrature points), and logit and cumulative logit link functions (for dichotomous concurrency, and three-level condom use, respectively). For all models the dependent variable was sexual risk (condom use or concurrency) measured at the follow-up controlling for baseline sexual risk and the covariates listed above. The primary predictor was change in anxiety or depression (baseline score subtracted from the follow-up score).

	Aim 1: Linear model equations						
	Condom use Concurrency						
	$CumuLogCondUse_{ij,t} = \gamma_{00} +$	$LogCncr_{ij,t} = \gamma_{00} +$					
Anxiety	$\gamma_{10}(Anx_{ij,t} - Anx_{ij,t-1}) +$	$\gamma_{10}(Anx_{ij,t} - Anx_{ij,t-1}) + \gamma_{20}Age_{ij,t-1} +$					
	$\gamma_{20}Age_{ij,t-1} + \gamma_{30}Edu_{ij,t-1} +$	$\gamma_{30}Edu_{ij,t-1} + \gamma_{40}SES_{ij,t-1} +$					
	$\gamma_{40}SES_{ij,t-1} + \gamma_{50}MaritalStatus_{ij,t-1} +$	γ_{50} MaritalStatus _{ij,t-1} + γ_{60} Cncr _{ij,t-1} +					
	$\gamma_{60}CondUse_{ij,t-1} + \gamma_{01}Tx_j + u_{0j} + r_{ij}$	$\gamma_{01}Tx_j + u_{0j} + r_{ij}$					
Depression	$CumuLogCondUse_{ij,t} = \gamma_{00} +$	$LogCncr_{ij,t} = \gamma_{00} +$					
	$\gamma_{10}(Dpr_{ij,t} - Dpr_{ij,t-1}) + \gamma_{20}Age_{ij,t-1} +$	$\gamma_{10}(Dpr_{ij,t} - Dpr_{ij,t-1}) + \gamma_{20}Age_{ij,t-1} +$					
	$\gamma_{30}Edu_{ij,t-1} + \gamma_{40}SES_{ij,t-1} +$	$\gamma_{30}Edu_{ij,t-1} + \gamma_{40}SES_{ij,t-1} +$					
	$\gamma_{50}MaritalStatus_{ij,t-1} +$	γ_{50} MaritalStatus _{ij,t-1} + γ_{60} Cncr _{ij,t-1} +					
	$\gamma_{60}CondUse_{ij,t-1} + \gamma_{01}Tx_j + u_{0j} + r_{ij}$	$\gamma_{01}Tx_j + u_{0j} + r_{ij}$					

Non-linear models. As indicated by previous studies of the relationship between depression and sexual risk [13-15], I tested non-linear specifications of mental health predictors for all hypothesized relationships to compare to the linear models described above. As prior evidence indicated a concave curvilinear relationship, I tested a quadratic specification of the mental health predictors. As these quadratic terms were not statistically significant, they were excluded from the final models.

Limitations. Respondents' willingness to report sexual risk behaviors may have been associated with the social acceptability of such an acknowledgment. In addition, reports of sexual risk behaviors were retrospective, and thus recall bias may have been an issue. However, the coding of the condom use and concurrency variables were designed to minimize the influence of recall bias, as discussed above. The HSCL-25 scale which was used to measure anxiety and depression was developed in a different cultural context than the study setting. Given that this scale has been previously validated in Tanzania as mentioned above [51, 52], I felt comfortable with the theoretical validity of the scale's measurement of anxiety and depression symptoms in this population. Further, the two time points as measured a year apart did not allow me to establish whether affective manifestations of anxiety and depression directly precede decisions to engage in risky sexual behaviors. Though there is an established temporal

relationship between changes in internalizing symptoms and in sexual risk in this study, event-level data are needed to establish the precise temporal relationship [83]. Within the limitations of the data available, the hypotheses as tested allowed me to establish whether changes in anxiety and depression were associated with sexual behaviors in this population.

Aim 2 analysis

Aim 2, Hypothesis 2a

Young men's sexual risk behaviors will be more correlated with those of friends than with those of other members of their camp network.

Aim 2, Hypothesis 2b

The relationship between mental health and sexual risk will be moderated by peer sexual risk behavior norms such that the relationship will be stronger for young men exposed to riskier peer norms than for men exposed to less risky peer norms.

In preliminary analyses for Aim 2 I described the structural measures of camp networks and further described the size, average in/out degree, and average degree by relationship type (acquaintance, friend, someone I don't get along with) of young men's friendship networks (ego-networks) at baseline.

Univariate network correlations

Next I tested network correlations for each dependent variable of interest (condom use and concurrency). I tested these correlations within each of the 59 networks in using a Quadratic Assignment Procedure (QAP). QAP is a non-parametric procedure for significance testing which is used to infer social influence by relating measures of structural similarity (in this case, structural similarity is defined by the existence of friendship tie) between two network members to a measure of their similarity on variables of interest [77]. QAP runs the specified model through multiple iterations by randomly permuting the

rows and columns of model matrices each time. This process produces an empirical distribution of the model coefficients against which to compare model estimates for hypothesis testing. Given a significance level of 0.05, an observed parameter is determined to be statistically significant if it is greater than 95 percent of the values in the empirical distribution based on 2000 iterations [77]. QAP is an extremely robust procedure because its assumptions are only those of the data generation process; all biases in the data are also seen in the permuted population [77].

Interactions between mental health and peer norms

To test the hypothesis that the relationship between mental health and sexual risk is moderated by the influence of peer sexual risk behavior norms, I built upon the main effect models used to assess hypothesis 1, and continued to utilize multilevel modeling to adjust estimates for the nested data structure by camp. For each model from Aim 1, I assessed the interaction between mental health and peer norm scores. Where these interaction terms were significant, I probed each significant interaction found by modeling the focal effect at the mean, one standard deviation above the mean, and one standard deviation below the mean of the moderator variable scores [99].

To adjust standard errors due to the dependence of observations in models including norms based on friends' self-reported behaviors, I included controls for subgroup clustering within each camp. As the same person could be nominated by multiple individuals as a friend, friendship cliques within the camps could lead to unobserved dependence within the data. To identify clique subgroups, I created a distance matrix for each camp where the cell values were the number of steps between each dyad. With each of these matrices I performed a principal components analysis to detect structurally significant subgroups within the camp, calculated the factor loading of each individual on each of the principal components, and included these terms in the regression model as a control [100]. Models to test these hypotheses are specified in the formulas below:

	Aim 2: Interaction model equations*					
	Condom use	Concurrency				
Anxiety	$CumuLogCondUse_{ij,t} = \gamma_{00} +$	$LogCncr_{ij,t} = \gamma_{00} +$				
	$\gamma_{10}(Anx_{ij,t} - Anx_{ij,t-1}) +$	$\gamma_{10}(Anx_{ij,t} - Anx_{ij,t-1}) +$				
	$\gamma_{20}(Norm_{ij,t} - Norm_{ij,t-1}) +$	$\gamma_{20}(Norm_{ij,t} - Norm_{ij,t-1}) +$				
	$\gamma_{30}(Anx_{ij,t} - Anx_{ij,t-1})(Norm_{ij,t} - Norm_{ij,t-1})$	$\gamma_{30}(Anx_{ij,t} - Anx_{ij,t-1})(Norm_{ij,t} - Norm_{ij,t-1})$				
	$\gamma_{40}Age_{ij,t-1} + \gamma_{50}Edu_{ij,t-1} + \gamma_{60}SES_{ij,t-1} +$	$\gamma_{40}Age_{ij,t-1} + \gamma_{50}Edu_{ij,t-1} + \gamma_{60}SES_{ij,t-1} +$				
	γ_{70} MaritalStatus _{ij,t-1} +	γ_{70} MaritalStatus _{ij,t-1} + γ_{80} Cncr _{ij,t-1} +				
	$\gamma_{80}CondUse_{ij,t-1} + \gamma_{01}Tx_j + u_{0j} + r_{ij}$	$\gamma_{01}Tx_j + u_{0j} + r_{ij}$				
	$CumuLogCondUse_{ij,t} = \gamma_{00} +$	$LogCncr_{ij,t} = \gamma_{00} +$				
~	$\gamma_{10}(Dpr_{ij,t} - Dpr_{ij,t-1}) +$	$\gamma_{10}(Dpr_{ij,t} - Dpr_{ij,t-1}) +$				
Depressior	$\gamma_{20}(Norm_{ij,t} - Norm_{ij,t-1}) +$	$\gamma_{20}(Norm_{ij,t} - Norm_{ij,t-1}) +$				
	$\gamma_{30}(Dpr_{ij,t} - Dpr_{ij,t-1})(Norm_{ij,t} - Norm_{ij,t-1})$	$\gamma_{30}(Dpr_{ij,t} - Dpr_{ij,t-1})(Norm_{ij,t} - Norm_{ij,t-1})$				
	$\gamma_{40}Age_{ij,t-1} + \gamma_{50}Edu_{ij,t-1} + \gamma_{60}SES_{ij,t-1} +$	$\gamma_{40}Age_{ij,t-1} + \gamma_{50}Edu_{ij,t-1} + \gamma_{60}SES_{ij,t-1} +$				
	γ_{70} MaritalStatus _{ij,t-1} +	γ_{70} MaritalStatus _{ij,t-1} + γ_{80} Cncr _{ij,t-1} +				
	$\gamma_{80}CondUse_{ij,t-1} + \gamma_{01}Tx_j + u_{0j} + r_{ij}$	$\gamma_{01}Tx_j + u_{0j} + r_{ij}$				

*models testing moderation by peer-reported behaviors include additional covariates reflecting loadings on principle components derived from camp distance matrices: $\sum_k \lambda_{ik} \eta_k$

Aim 3 analysis

Aim 3, Hypothesis 3a	
Young men's centrality within their camp s relationship between mental health and se relationship will be weaker for men with h lower centrality.	ocial network will moderate the exual risk behaviors such that the igher centrality than for those with
Aim 3, Hypothesis 3b	
Network social cohesion will moderate the and sexual risk behaviors such that the rel in camps with higher levels of social cohes levels of social cohesion.	relationship between mental health ationship will be weaker for young men ion than for those in camps with lower

To test these hypotheses, I built off of the main effects models from Aim 1 and include measures of

centrality, social cohesion, and respective interaction terms between these network measures and

anxiety or depression to test the hypothesized moderating effects. Baseline measures of centrality and

social cohesion were used in these models.

Hypothesis 3a. Building upon the main effect models used to assess hypothesis 1, to test the individual level moderator of centrality I added the centrality variable to the model. Following this, I added an interaction term for the mental health predictor and centrality variable to the model.

Hypothesis 3b. To evaluate moderation by camp social cohesion of the relationship between mental health and sexual risk, I fit two-level, slopes as outcomes models including camp social cohesion and the interaction between mental health and social cohesion as predictors.

If a significant interaction was found in testing either hypothesis 3a or 3b, I intended to probe each significant interaction by modeling the main effect at the mean, one standard deviation above the mean, and one standard deviation below the mean of the moderator variable scores [99]. No such significant interaction was found. All models were fit using *proc glimmix*, quadrature estimation (using 15 quadrature points), and logit and cumulative logit link functions (for dichotomous concurrency, and three-level condom use, respectively). Models to test these hypotheses are specified below:

	Aim 3, Hypothesis 3a: Centrality interaction model equations							
	Condom use Concurrency							
	$CumuLogCondUse_{ij,t} = \gamma_{00} + $	$LogCncr_{ij,t} = \gamma_{00} +$						
Anxiety	$\gamma_{10}(Anx_{ij,t} - Anx_{ij,t-1}) + \gamma_{20}Centr_{ij,t-1} +$	$\gamma_{10}(Anx_{ij,t} - Anx_{ij,t-1}) + \gamma_{20}Centr_{ij,t-1} +$						
	$\gamma_{30}Centr_{ij,t-1}(Anx_{ij,t} - Anx_{ij,t-1}) +$	$\gamma_{30}Centr_{ij,t-1}(Anx_{ij,t} - Anx_{ij,t-1}) +$						
	$\gamma_{40}Age_{ij,t-1} + \gamma_{50}Edu_{ij,t-1} + \gamma_{60}SES_{ij,t-1} +$	$\gamma_{40}Age_{ij,t-1} + \gamma_{50}Edu_{ij,t-1} + $						
	$\gamma_{70}MaritalStatus_{ij,t-1} +$	$\gamma_{60}SES_{ij,t-1} + \gamma_{70}MaritalStatus_{ij,t-1} +$						
	$\gamma_{80} CondUse_{ij,t-1} + \gamma_{01}Tx_j + u_{0j} + r_{ij}$	$\gamma_{80}Cncr_{ij,t-1} + \gamma_{01}Tx_j + u_{0j} + r_{ij}$						
	$CumuLogCondUse_{ij,t} = \gamma_{00} +$	$LogCncr_{ij,t} = \gamma_{00} +$						
Depression	$\gamma_{10}(Dpr_{ij,t} - Dpr_{ij,t-1}) + \gamma_{20}Centr_{ij,t-1} +$	$\gamma_{10}(Dpr_{ij,t} - Dpr_{ij,t-1}) + \gamma_{20}Centr_{ij,t-1} +$						
	$\gamma_{30}Centr_{ij,t-1}(Dpr_{ij,t} - Dpr_{ij,t-1}) +$	$\gamma_{30}Centr_{ij,t-1}(Dpr_{ij,t} - Dpr_{ij,t-1}) +$						
	$\gamma_{40}Age_{ij,t-1} + \gamma_{50}Edu_{ij,t-1} + \gamma_{60}SES_{ij,t-1} +$	$\gamma_{40}Age_{ij,t-1} + \gamma_{50}Edu_{ij,t-1} + $						
	$\gamma_{70}MaritalStatus_{ij,t-1} +$	$\gamma_{60}SES_{ij,t-1} + \gamma_{70}MaritalStatus_{ij,t-1} +$						
	$\gamma_{80}CondUse_{ij,t-1} + \gamma_{01}Tx_j + u_{0j} + r_{ij}$	$\gamma_{80}Cncr_{ij,t-1} + \gamma_{01}Tx_j + u_{0j} + r_{ij}$						

	Aim 3, Hypothesis 3b: Camp cohesion interaction model equations						
	Condom use	Concurrency					
	$CumuLogCondUse_{ij,t} = \gamma_{00} +$	$LogCncr_{ij,t} = \gamma_{00} +$					
Anxiety	$\gamma_{10}(Anx_{ij,t} - Anx_{ij,t-1}) + \gamma_{20}Cohes_{ij,t-1} +$	$\gamma_{10}(Anx_{ij,t} - Anx_{ij,t-1}) + \gamma_{20}Cohes_{ij,t-1} +$					
	$\gamma_{30}Cohes_{ij,t-1}(Anx_{ij,t} - Anx_{ij,t-1}) +$	$\gamma_{30}Cohes_{ij,t-1}(Anx_{ij,t} - Anx_{ij,t-1}) +$					
	$\gamma_{40}Age_{ij,t-1} + \gamma_{50}Edu_{ij,t-1} + \gamma_{60}SES_{ij,t-1} +$	$\gamma_{40}Age_{ij,t-1} + \gamma_{50}Edu_{ij,t-1} + $					
	$\gamma_{70}MaritalStatus_{ij,t-1} +$	$\gamma_{60}SES_{ij,t-1} + \gamma_{70}MaritalStatus_{ij,t-1} +$					
	$\gamma_{80}CondUse_{ij,t-1} + \gamma_{01}Tx_j + u_{0j} + r_{ij}$	$\gamma_{80}Cncr_{ij,t-1} + \gamma_{01}Tx_j + u_{0j} + r_{ij}$					
	$CumuLogCondUse_{ij,t} = \gamma_{00} +$	$LogCncr_{ij,t} = \gamma_{00} +$					
on	$\gamma_{10}(Dpr_{ij,t} - Dpr_{ij,t-1}) + \gamma_{20}Cohes_{ij,t-1} +$	$\gamma_{10}(Dpr_{ij,t} - Dpr_{ij,t-1}) + \gamma_{20}Cohes_{ij,t-1} +$					
pressi	$\gamma_{30}Cohes_{ij,t-1}(Dpr_{ij,t} - Dpr_{ij,t-1}) +$	$\gamma_{30}Cohes_{ij,t-1}(Dpr_{ij,t} - Dpr_{ij,t-1}) +$					
	$\gamma_{40}Age_{ij,t-1} + \gamma_{50}Edu_{ij,t-1} + \gamma_{60}SES_{ij,t-1} +$	$\gamma_{40}Age_{ij,t-1} + \gamma_{50}Edu_{ij,t-1} + $					
Ď	$\gamma_{70}MaritalStatus_{ij,t-1} +$	$\gamma_{60}SES_{ij,t-1} + \gamma_{70}MaritalStatus_{ij,t-1} +$					
	$\gamma_{80} CondUse_{ij,t-1} + \gamma_{01}Tx_j + u_{0j} + r_{ij}$	$\gamma_{80}Cncr_{ij,t-1} + \gamma_{01}Tx_j + u_{0j} + r_{ij}$					

3.5 Sample size and power

Using baseline data, I computed the intraclass correlation (ICC) for condom use and concurrency by camp (0.060 and 0.034, respectively). Using these estimates and an average camp size of 21 men, simulation studies were conducted to assess the power of the proposed study for Aim 1. For a significance level of 0.05, there was 80% power to detect an association between mental health and condom use provided the true odds ratio of always using condoms was 0.60 or lower (further from the null) for men categorized as displaying depression/anxiety versus men who are not displaying depression/anxiety. There was also 80% power to detect an association between mental health and concurrency provided the true odds ratio of concurrency is 1.65 or higher (further from the null) for men categorized as displaying depression/anxiety versus men who are not displaying the null) for men categorized as displaying depression/anxiety versus men who are not displaying the null) for men categorized as displaying the true odds ratio of concurrency is 1.65 or higher (further from the null) for men categorized as displaying depression/anxiety versus men who are not displaying depression/anxiety.

For the correlation hypotheses related to Aim 2, at a significance level of 0.05 there was 80% power to detect correlations as low as 0.12 and 0.13 for condom use and concurrency, respectively. For the moderation hypotheses related to Aims 2 and 3, statistical power was expected to be sufficient based on the simulation results of the power of cross-level interaction tests in multilevel studies presented by Mathieu et al [101]. Mathieu et al. found that larger level one sample sizes (greater than

18) along with larger level two sample sizes (greater than 35), often surpassed power of 80% in simulations with varying magnitudes of the direct cross-level effects as well as standard deviations of the level one slope coefficients [101].

CHAPTER 4: PAPER 1

CHANGES IN ANXIETY AND DEPRESSION SYMPTOMS PREDICT SEXUAL RISK BEHAVIORS AMONG YOUNG MEN LIVING IN DAR ES SALAAM, TANZANIA

4.1 Background

There are 4 million young people ages 15-24 living with HIV [102], 85% of whom live in sub-Saharan Africa [103]. AIDS-related deaths among youth rose by 50% between 2005 and 2012 and adolescents and young adults account for a growing proportion of African populations [103], making the need to target and engage youth in HIV prevention increasingly important. 40 percent of new infections in Tanzania occur among 16-24 year olds [1], and 17 percent of women and 8 percent of men ages 15-19 years are HIV-infected in Tanzania [3]. Young men are important targets in preventing HIV transmission in this context as gender norms position them to control the terms and conditions of sexual relationships [32], and encourage them to engage in high risk sexual behaviors including inconsistent condom use and sexual partner concurrency [1, 33-36]. Among young men in Dar es Salaam, the commercial capital of Tanzania, the proportion of young men reporting condom use at last sex seems to decline over adolescence (92% of 15-year-olds compared to 70% of 19-year-olds) [29]. and the proportion reporting concurrent sexual partnerships increases with age, from 5 percent at age 15 to 44% by age 19 [29]. This trend is consistent with national studies in Tanzania; half of sexually active youth ages 15-24 surveyed in 2010 did not use a condom at last intercourse, and nearly one-third reported concurrent partners [4].

Anxiety and depression are important correlates of sexual risk [5, 6, 8-10, 104], and may influence risk through multiple mechanisms including substance use [10], maladaptive coping to deal

with stress [47], and impaired decision making [48]. Cross-sectional associations have repeatedly been found between both anxiety and depression, and sexual risk behaviors [5-7, 9, 10], but we have little understanding of the longitudinal association between these anxiety, depression, and sexual risk [8, 11, 12, 49, 105], particularly from low-income or high HIV-prevalence contexts as the bulk off the extant evidence comes from high income contexts [5, 8, 10-15]. Due to its prevalence and frequent comorbidity with depression [106], understanding anxiety as a predictor of sexual risk is particularly needed in this context, as there has been only one study examining anxiety as a predictor of sexual risk in a sub-Saharan Africa to our knowledge [7]. Finally, given the importance of young men in HIV prevention we need better evidence of these risk relationships among young men who have sex with women, a literature which is lacking. To address these gaps in the literature, we assessed the longitudinal relationship between symptoms of anxiety and depression and sexual risk behaviors among a population of socioeconomically marginalized young men in Dar es Salaam. The results of this study provide much needed information for the development of effective HIV and sexual risk prevention interventions relevant to the growing population of male youth in sub-Saharan urban areas.

4.2 Methods

Study context

This study was conducted in the context of an ongoing HIV prevention trial, *A Multilevel Intervention to Reduce HIV Risk Among Networks of Men in Tanzania* (PI: Suzanne Maman; 1R01MH098690-01), a cluster randomized trial of a microfinance and health leadership intervention to prevent sexually transmitted infections and intimate partner violence [107]. Participants in this trial are members of venues known as "camps" in Dar es Salaam where men socialize and engage in small scale enterprise [28]. Most camp members are not formally employed and who spend several hours each day at their camp [28].

Sampling and data collection

We identified camps for inclusion in the trial in four wards (equivalent to U.S. census tracts) of Dar es Salaam (Manzese, Tandale, Mwananyamala, and Mabibo) using an adaptation of PLACE (Priorities for Local AIDS Control Efforts) methodology [108]. To be eligible for inclusion in the trial, camps had to have between 20 and 80 members, have been in existence for at least one year prior to the baseline assessment, and report no violent incidents with weapons in the past 6 months. A total of 303 camps were verified, of which 205 were eligible. 60 camps were randomly selected for inclusion in the study. Through member rosters completed by the leaders of these camps, 1581 male members were identified and assessed for eligibility. To be eligible men had to: 1) be a registered camp member for at least three months; 2) plan on residing in Dar es Salaam for the next 30 months; 3) be 15 years or older; 4) visit the camp at least once per week; and 5) be willing to provide contact information for self and two family members or friends.

Eligible participants were asked to provide written consent, and consenting participants completed the baseline assessment in fall 2013 and a follow up assessment 12 months after the launch of the intervention. Both questionnaires were administered by trained Tanzanian interviewers in Swahili using computer assisted personal interviewing (CAPI). The interviews lasted approximately 60-90 minutes. A total of 1,258 men completed the baseline behavioral assessment, after which camp members from one camp (n=9) were removed from the study because of new information rendering them ineligible for participation, resulting in a final baseline sample of 1,249 men within 59 camps. 978 men in the 59 camps completed the follow-up assessment for 78% retention.

Measures

Symptoms of anxiety and depression were measured using a version of the Hopkins Symptom Checklist-25 (HSCL-25)[85] that had previously been translated and validated in Tanzania [86, 87]. Participants

rated a total of 25 symptoms (10 related to anxiety, 15 to depression) rated on a 4-point Likert-type scale. Anxiety and depression scores were calculated by taking the mean of the corresponding 10 and 15 items, respectively. In the baseline sample both subscales showed good internal consistency (a = 0.94 for anxiety, a = 0.91 for depression). Where depression and anxiety are described dichotomously in the results, the recommended cutoff of 1.75 was used for clinical caseness [88].

Condom use was measured as an ordered-categorical variable created from men's self-report of condom use over the 3 most recent sexual partners (number of sex acts and number of times condoms used over the most recent month of the relationship with each partner). Using the proportion of reported sex acts where condoms were used, participants were assigned to one of three categories: "never use" (0% use), "some use" (greater than 0%, less than 100%), or "always use" (100%). This categorical approach was preferred to a continuous variable in the interest of minimizing the effect of recall bias, and over a dichotomous variable per the recommendations of Noar et al [83]. Sexual concurrency was evaluated by self-report of any overlapping sexual partnerships. Participants were asked to enumerate their 3 most recent relationships over the past 12 months and to report if they had sex with anyone else during any of these partnerships. This measure was developed following best practices recently recommended by USAID [84].

Covariates included in all analyses include age, education level, economic status, marital status, and treatment condition. Age was calculated based on reported date of birth, or when not available the reported age in years. Participants reported the highest level of education they had completed and education responses were collapsed into three categories: primary school or less (no education, Standard 4 or less, or Standard 5-7); some secondary school (Form 1, Form 2, or Form 3); or secondary school completed or greater (Form 4 or Greater than Form 4). Socioeconomic status (SES) was evaluated through the Filmer Pritchett Wealth Index [91]. Participants were asked to indicate which of 10 possible assets they owned [92] and a composite score was created by weighting each asset by its factor loading

on the first component in a principle components analysis [91]. The weighted composite score for each participant was categorized into terciles based on the entire baseline sample (the lowest 33% as "lowest SES", the highest 33% as "highest SES," and the remainder as "middle SES"). Marital status was evaluated by asking men if they had ever been married. Finally, as the intervention being evaluated was designed to affect condom use and sexual partner concurrency, treatment condition was included as a covariate to account for this design effect in all analyses.

Analysis

All statistical analyses were conducted in SAS v 9.4 and used a 2-sided significance level of 0.05. First we described demographic characteristics of the participants and levels of variables of interest at baseline and at the follow-up. To address missing data (primarily due to attrition), we applied sequential multiple imputation using the fully conditional specification in *proc MI*. Forty imputations were created using a linear regression specification for continuous variables and a logistic regression specification for categorical variables. To test the primary hypothesis while accounting for dependence due to clustering within camps, we fit multilevel models with a generalized link function. All models were fit using *proc glimmix* with quadrature estimation, random intercepts, and logit and cumulative logit link functions (for dichotomous concurrency, and three-level condom use, respectively).

All models assessed the association between the dependent variable of reported sexual behavior at the follow-up (condom use or concurrency) and the independent variable of within-person change in symptoms of anxiety or depression, controlling for baseline sexual behavior and the covariates listed above. Change scores of symptoms of anxiety and depression were chosen to increase the plausibility of causal inferences about the observed temporal associations between men's symptoms and sexual risk behaviors. Because of the frequent comorbidity between anxiety and depression, we were concerned about a possible level of collinearity between measurements of their symptoms (the

correlation between symptoms of anxiety and depression was 0.76 at baseline and 0.74 at the followup). For this reason, we tested three models for each sexual behavior dependent variable (for a total of six models): 1) Change in anxiety symptoms only; 2) Change in depression symptoms only; and 3) Change in both anxiety and depression symptoms. We compared the parameter estimates and confidence intervals of models 1 and 2 to those in model 3 to assess the effects of our two primary predictors.

Ethical Review

The study was approved by the ethical review committees at the University of North Carolina at Chapel Hill and Muhimbili University of Health and Allied Sciences in Dar es Salaam, Tanzania. Individual written informed consent was obtained from all study participants.

4.3 Results

1249 men were interviewed at baseline, of whom 1113 (89%) reported being sexually active. These 1113 men were included in the analyses presented below.

Participant characteristics

Sexually active men interviewed at baseline had an average age of 27 years (range: 15 to 59). Over half had a primary school education or less (59%), nearly a third had graduated from secondary school (31%), and the remaining 11% had some secondary school but had not graduated. A quarter of the men had ever been married (25%) and 38% had children.

	Total
Age (y)***	26.8 ± 7.1
Currently in school	97 (8.7%)
Education level (ref = less than primary school completed)	
Primary school or less	652 (58.7%)
Some secondary school	116 (10.5%)
Secondary school completed or greater	342 (30.8%)
Socioeconomic status	
Lowest	291 (26.2%)
Middle	435 (39.1%)
Highest	386 (34.7%)
Ever married	277 (25.0%)
Has children	423 (38.0%)

Table 4. Baseline Participant Characteristics (n=1113)^a

^aData are expressed as No.(%) or Mean ± SD

Change in symptoms of anxiety and depression and sexual behaviors

Men had an average score of 1.4 (range 1 to 4 with higher scores indicating more severe symptoms) for both anxiety and depression symptoms at both baseline and at the follow-up assessment (Table 5). At baseline 21% reported clinically significant symptoms of depression and 19% reported clinically significant symptoms of anxiety based on common clinic cutoffs (score \geq 1.75) [90]. At the follow-up 20% and 16% of respondents met the criteria for depression and anxiety, respectively. Fifteen percent of men showed symptoms of both anxiety and depression at baseline compared to 12% of respondents at the follow-up. The mean within-person change in depression and anxiety scores were both close to zero, but there was substantial variation in change scores (-0.03 ± 0.76 for depression and -0.03 ± 0.72 for anxiety). The within-person correlation between baseline and follow-up symptoms was 0.08 for anxiety and 0.13 for depression. The correlation between symptoms of anxiety and depression was high at both time points: 0.76 at baseline and 0.74 at the follow-up. Furthermore, the correlation between withinperson change in anxiety and depression symptoms was 0.73.

At the sample level, there were relatively similar levels of condom use and concurrency across the two time points; about half of men reported never using condoms at both baseline and the followup (53% and 52%, respectively). More men reported always using condoms at baseline than at the follow-up (33% and 26%, respectively). 14% of men reported sometimes using condoms at baseline, and this proportion increased to 22% at the follow-up. Within individuals there were significant changes in condom use; only 325 (48%) of men remained in the same condom use category at the follow-up as compared to baseline (Table 6). Men who reported no condom use at baseline were the most likely to maintain the same use status at the follow-up (216 men, 61% of baseline non-users). Only 26% of men with some use at baseline reported the same level of use at the follow-up, with the greatest number changing to not using condoms (54 men, 53% of baseline sometimes users). Among baseline always users, nearly as many men reported no condom use at the follow-up (82 men, 37% of baseline always users) as reported still always using condoms (83 men, 38% of baseline always users). The within-person correlation between baseline and follow-up condom use was 0.21. Many more men reported concurrency at the follow-up (262, 32%) than at baseline (193, 20%; Table 5). 30% of men who were not concurrent at baseline reported concurrency at the follow-up (Table 7) while half (49%) of men who were concurrent at baseline reported concurrency again at the follow-up. The within-person correlation between baseline and follow-up concurrency was 0.16.

	Baseline	Follow-up
Depression score	1.43 ± 0.57	1.42 ± 0.56
Anxiety score	1.38 ± 0.51	1.36 ± 0.53
Depression (score \geq 1.75)	237 (21.3%)	174 (20.0%)
Anxiety (score ≥ 1.75)	206 (18.5%)	141 (16.2%)
Both depression and anxiety (both scores \ge 1.75)	164 (14.7%)	101 (11.6%)
Condom use		
Never	491 (52.7%)	408 (51.6%)
Sometimes	133 (14.3%)	176 (22.3%)
Always	308 (33.1%)	207 (26.2%)
Concurrency	193 (20.2%)	262 (32.1%)

Table 5. Cross-sectional description of symptoms of anxiety and depression and sexual risk^a

^aData are expressed as No. (%) or Mean ± SD

Table 6. Condom use change

		Follow-up condom use Table 7. C		oncurrency change				
		None	Some	Always			Follow-up co	oncurrency
							No	Yes
Baseline condom use	None	216 (59.7%)	73 (20.2%)	73 (20.2%)	rency		400	172
	me	54	26	21	oncuri	Baseline concurr Yes	(69.9%)	(30.1%)
	So	(53.5%)	(25.7%)	(20.8%)	ine co		76	73
	Always	82 (37.3%)	55 (25.0%)	83 (37.7%)	Basel		(51.0%)	(49.0%)

Anxiety models

The random intercepts models for anxiety (Table 8; Model 1) indicated that changes in anxiety symptoms were significantly associated with both condom use and concurrency at the follow-up, controlling for baseline behavior and covariates listed in the Methods. Specifically, men who experienced an increase in anxiety symptoms reported lower condom use levels and had higher odds of

concurrency at the follow-up than did men with decreasing anxiety symptoms. For each one-unit increase in anxiety symptoms, there was an estimated 18% decrease in the odds of a higher level of condom use (aOR=0.82; 95% CI: 0.68, 0.99). For each one-unit gain in anxiety symptoms, there was an estimated 67% increase in the odds of sexual concurrency (aOR=1.67; 95% CI: 1.34, 2.08).

Depression models

The random intercepts models for depression (Table 8; Model 2) indicated that changes in depression symptoms were also significantly associated with both condom use and concurrency at the follow-up, controlling for baseline behavior and covariates listed in the Methods. Specifically, men who experienced an increase in depression symptoms reported lower condom use levels and had higher odds of concurrency at the follow-up than did men with decreasing depression symptoms. For each one-unit increase in depression symptoms, there was an estimated 24% decrease in the odds of a higher level of condom use (aOR=0.76; 95% CI: 0.64, 0.90). For each one-unit gain in depression symptoms, there was an estimated 60% increase in the odds of sexual concurrency (aOR=1.60, 95% CI: 1.30, 1.97).

Independent effects of anxiety and depression

To evaluate the relative strength of the association of anxiety and depression with each sexual behavior, we tested additional models for condom use and concurrency including both anxiety and depression change as predictors in the same models (Table 8; Model 3). The width of the confidence intervals for the estimated effects of anxiety and depression on condom use showed little change between the separate and combined models, indicating that collinearity had minimal effect on model fit. There was a larger widening of the confidence intervals in the combined concurrency model, indicating that precision was negatively but modestly affected by collinearity between anxiety and depression. When the effects of anxiety and depression were estimated together, only depression was significantly associated with
condom use, and not anxiety. For concurrency, when anxiety and depression effects were estimated together, anxiety and depression change predictors remained significant or marginally significant. This indicates that both anxiety and depression symptoms may independently affect concurrency, but the effect of each is attenuated when controlling for the other. On the other hand, observed association between anxiety symptoms and condom use may be explained by anxiety's high level of covariance with depression.

	Condom use		Concurrency	
Model	aOR (95%CI)	р	aOR (95%CI)	р
1. Anxiety change only	0.82 (0.68, 0.99)	0.040	1.67 (1.34, 2.08)	<.0001
2. Depression change only	0.76 (0.64, 0.90)	0.002	1.60 (1.30, 1.97)	<.0001
3. Both anxiety & depression				
Anxiety change	1.02 (0.77, 1.15)	0.873	1.40 (1.04, 1.89)	0.026
Depression change	0.75 (0.58, 0.97)	0.026	1.27 (0.96, 1.67)	0.095

Table 8. Aim 1 model results

Note: Covariates in all models include treatment condition, age, education level, SES, and ever having been married.

4.4 Discussion

In this sample of young male members of "camps" in Dar es Salaam, we found that many men experienced clinically significant symptoms of anxiety and depression (19% and 21%, respectively at baseline), and reported significant levels of partner concurrency and inconsistent condom use from an HIV prevention perspective. We found that changes in anxiety and depression were significantly associated with both condom use and concurrency at the follow-up controlling for baseline behavior. A secondary model indicated that anxiety and depression may independently affect concurrency but depression may be the primary predictor of condom use, with the observed relationship between anxiety and condom use being likely attributable to the covariance between anxiety and depression symptoms. These findings provide important evidence of the longitudinal association between symptoms of anxiety and depression and HIV risk a primarily heterosexual population of young men in a sub-Saharan urban setting.

Though the present study did not assess potential mechanisms explaining the observed relationships, there are multiple theorized pathways through which negative affect in the form of symptoms of anxiety and depression may lead to sexual risk behaviors. Negative affective states may hamper one's ability to self-regulate and change thoughts, emotions, and behaviors [109, 110]. Without this ability to self-regulate, individuals experiencing anxiety and depression may face greater challenges in avoiding risky sexual behaviors. Furthermore, when faced with negative emotions they may be motivated to act to relieve the negative emotion [111, 112], which can be achieved through self-gratifying behaviors (e.g. sex with a secondary partner) which serve as a distraction from the negative emotions [112-118]. Finally, people experiencing depressed emotions may have decreased motivation to care for themselves and to avoid risky behaviors [119]. Future research is needed to determine which of these mechanisms is in operation in this population.

These mechanisms may help to explain our finding that depression and not anxiety seemed to be independently associated with condom use. Because of the hopelessness, sense of futility, and tendency toward self-destructive behaviors associated with depression [120], men experiencing greater levels of depression may be less likely to use condoms. On the other hand, men with symptoms of anxiety may have a range of experiences of anxiety that affect condom use differently. While some individuals may experience excessive worry [121], leading to vigilance about condom use, others may experience impulsivity [122] that could lead to disregard for condom use. If at the individual level feelings of anxiety encourage or discourage condom use differently for different men, at the population level there may be no consistent observable relationship between general symptoms of anxiety and

condom use. Future studies should seek to understand how different anxious tendencies (e.g. worrytype vs. impulsive-type) relate to condom use.

Men in this study experienced substantial changes in their anxiety and depression symptoms. Substantial fluctuations in symptoms of anxiety and depression have been frequently observed, particularly among adolescents and young adults [123-125], and may be dependent upon the extent to which a person's anxiety and depression is related to genetic heritability rather than environmental influences [125]. Further research is needed to determine the extent to which sexual risk behaviors are associated with enduring traits of anxiety or depression as opposed to shorter-term states of anxiety or depression which may be highly contextually influenced. To further elucidate these relationships, eventlevel data such as ecological momentary assessment (EMA) data will be needed [126]. No such study has been undertaken in sub-Saharan Africa to our knowledge. Future studies employing event-level data should include a focus on the relationship between anxiety and partner concurrency, as there has been an overwhelming focus on depression in this line of research to date.

Limitations

There are important limitations to this study. Respondents' willingness to report sexual risk behaviors may have been associated with the social acceptability of such an acknowledgment. In addition, reports of sexual risk behaviors were retrospective, and thus recall bias may be an issue. The coding of the condom use and concurrency variables were designed to minimize the influence of recall bias, as discussed in the Methods. The HSCL-25 scale which was used to measure anxiety and depression was developed in a different cultural context than the study setting, though the scale version used has been previously validated in Tanzania [86, 87]. Further, the two time points as measured a year apart do not allow us to establish whether affective manifestations of anxiety and depression directly precede decisions to engage in risky sexual behaviors. Event-level data is needed to establish a precise temporal

relationship between anxiety and depression and sexual risk [126]. Within the limitations of the data available, the hypotheses as tested allow us to establish that changes in anxiety and depression are associated with sexual behaviors in this population.

Future directions

The results of this study indicate the need to address anxiety and depression among young men in HIV prevention efforts. Previous HIV prevention trials for women have been more effective when they have also been able to improve mental health [127]. While there are promising interventions to address depression in this context [128, 129], we need evidence-based interventions for anxiety in low resource settings. Promising models of mental health promotion which may be scalable in this setting include training nurses to implement mental health screening and treatment [129, 130], or training lay people to screen for common mental disorders and make referrals to counseling [131-133]. To be able to target such interventions, we need to understand which men are most at risk for anxiety and depression. More population-based research is needed in sub-Saharan Africa, but the few studies from this context indicate that depression may be associated with age [134, 135], marital status [134, 136, 137], education [136, 138], employment and income [135, 136, 138], and gender [134, 136]. Though studies have observed a higher risk of depression among women, the substantial levels of clinically significant symptoms of anxiety and depression among men in this study should not be discounted. Research specific to men will be needed as the determinants of anxiety and depression are likely to be different among men than among women [134, 136].

Conclusion

This study contributes to our understanding of the association between mental health and sexual risk behaviors among young men in an African urban setting. The results indicate that changes in anxiety and

depression were associated with both condom use and concurrency, but that anxiety may only independently affect concurrency and not condom use. As we found clinically significant symptoms of anxiety and depression among one out of five men in this setting, efforts to screen and provide treatment for depression and anxiety disorders are indicated as a potentially important component of primary HIV prevention in this setting. The results further indicate the development of effective HIV prevention interventions targeting young men living with anxiety and depression.

CHAPTER 5: PAPER 2

MODERATION BY OBSERVED AND PERCEIVED BEHAVIORAL NORMS OF THE RELATIONSHIP BETWEEN MENTAL HEALTH AND SEXUAL RISK AMONG YOUNG MEN LIVING IN DAR ES SALAAM, TANZANIA

5.1 BACKGROUND

There are 4 million young people ages 15-24 living with HIV [102], 85% of whom live in sub-Saharan Africa [103]. AIDS-related deaths among youth rose by 50% between 2005 and 2012 and adolescents and young adults account for a growing proportion of African populations [103], making the need to target and engage youth in HIV prevention increasingly important. In Tanzania specifically, 40 percent of new infections occur among 16-24 year olds [1]. Young men are important targets in HIV prevention in this context as gender norms position them to control the terms and conditions of sexual relationships [32], and encourage them to engage in high risk sexual behaviors including inconsistent condom use and sexual partner concurrency [1, 33-36]. Half of sexually active Tanzanian youth ages 15-24 did not use a condom at last intercourse in 2010, and nearly one-third reported concurrent partners [4].

HIV risk in eastern and southern Africa has been observed to cluster in informal urban settlements [139-141]. The majority of urban residents in southern and eastern Africa live in these settlements, informally known as slums [22]. The populations of these settlements are growing due to high levels with rapid urbanization. In Dar es Salaam, Tanzania's commercial capital, the current population of 4.4 million [19] is growing at an annual rate of 4.3% [142]. The city's growing population has led to a high demand for infrastructure which the government has been unable to meet, and as a result an estimated 70% of the city's population lives in informal settlements [22]. People living in these settlements lack access to sufficient housing and basic services such as water and sanitation [142]. The prevalence of these living conditions is also explained by the high levels of poverty [143] and unemployment (21.5%) in the city [144]. Residence in contexts like these settlements is linked to HIV risk through a number of mechanisms. Both the state of poverty itself and the context of disadvantage in impoverished neighborhoods promote sexual risk behaviors [145-148]. As there are often poor education and employment options for youth in these settings [149, 150], youth often lack hope and positive orientation to the future [151], which is in turn linked to risk behavior [152, 153]. Neighborhood physical disorder also contributes to psychological distress [154] by providing a constant reminder of one's status in poverty, and can cause the individual to feel trapped in their current state [155]. The living environment can shape mental health through other stressors, such as violence, poor housing, noise, and crowding [156, 157]. These same stressors can lead to substance use [148, 158]. Both poor mental health and substance use can in turn shape sexual risk [145, 159].

An impoverished living context has also been associated with riskier norms related to sexual risk behavior. Communities have distinct social norms [160, 161], and the social isolation of impoverished neighborhoods can create a context in which distinct norms emerge and sustain themselves [155, 162]. Neighborhood disorder is also associated with the perceived prevalence of sexual risk behaviors [163]. The potential clustering of risky social norms is of concern for HIV prevention as norms can affect individuals' perceptions of acceptable behavior, and are particularly influential for youth who are more strongly influenced by the attitudes and behaviors of their peers that older adults [164]. Normative influence may take the form of active peer pressure [165], leading to perceptions of what actions are considered acceptable or unacceptable to peers known as *injunctive norms* [166]. Influence can also occur through less direct social learning processes [167, 168] whereby peers both transmit and reinforce norms [169], shaping individual perceptions of what peers are actually doing, known as *descriptive norms* [166].

In disadvantaged contexts, youth are exposed to multiple susceptibilities including normative risk behaviors and poor mental health. It is important to understand if these coexisting susceptibilities simply add to one another or compound in shaping young men's predisposition to sexual risk behaviors. Knowledge of such a compounding (or interaction) effect would indicate the need to address these susceptibilities in tandem and not in separate, targeted programs. Because of previous evidence from the study population of the relationship between anxiety, depression, and sexual risk [30], and preliminary evidence of peer normative influence on sexual risk behaviors in this population [170], we have chosen to focus on mental health and social norms to assess the extent to which these susceptibilities act together to shape sexual risk. In a previous study with the same population of young men living in Dar es Salaam, we found that changes in depression over a one-year period were independently associated with condom use and concurrency, and anxiety was independently associated with concurrency (see Chapter 4: Paper 1). We also previously found a significant clustering of condom use, partner concurrency, and attitudes toward these behaviors within subgroups of the study population (i.e. a significant proportion of the variance in these behaviors was attributable to group membership) [170]. Qualitative evidence from the study population further indicates normative influence on men's sexual behaviors, indicating that men pressure and encourage each other to engage in sexual risk behaviors [28].

Behavioral norms and mental health interact to shape drinking behaviors among youth in the US [171], but there are no studies to our knowledge assessing a similar relationship for sexual behaviors. To address this gap, we aim to understand if social norms moderate the relationship between mental health and sexual risk. We hypothesize that higher levels of risk behaviors among peers and perceptions of riskier norms among peers will serve to magnify this individual-level risk. In other words, we theorize that men with higher levels of anxiety or depression may have a higher inclination toward risk behaviors, and the influence of peer norms may increase the likelihood that this inclination manifests in actual risk

behaviors. Evidence of such an interaction between mental health and peer norms would indicate that there may be something social or learned about sexual risk as a response to poor mental health. Further, we aim to gain specific understanding of the role of different types of social norms in men's sexual behaviors, and do so by additionally assessing the direct effect of observed peer behaviors and perceived behavioral norms on men's sexual behaviors. The results of this study offer implications for future interventions aiming to reduce HIV risk among young men living in urban informal settlements in east Africa.

5.2 METHODS

Study context

This study was conducted in the context of an ongoing HIV prevention trial, *A Multilevel Intervention to Reduce HIV Risk Among Networks of Men in Tanzania* (PI: Suzanne Maman; 1R01MH098690-01), a cluster randomized trial of a microfinance and health leadership intervention to prevent sexually transmitted infections and intimate partner violence [107]. The setting and design of this trial have been previously described in detail [107]. The study is taking place in the country's commercial capital, Dar es Salaam, where 7% of the population is living with HIV [2].

For this trial, we identified social networks of young men occupying designated venues known as "camps" in Dar es Salaam where men socialize and engage in small scale enterprise [28]. Participants in the present study are 1113 sexually active men who are members of 59 of these camps. Camps are stable social units; those included in the trial have been in existence for an average of eight years. Men typically belong to only one camp and pay membership fees to belong to that camp [28]. Camps are made up of youth who frequent the venue for its supportive social environment [28]. Most men in the camps are not formally employed and spend several hours each day at their camp [28]. These camps

provide unique access to social networks of high risk young men who are otherwise difficult to reach, as they are largely neither in school nor formally employed [28].

Sampling and data collection

We identified camps for inclusion in the trial in four wards (which are equivalent to U.S. census tracts) of Dar es Salaam (Manzese, Tandale, Mwananyamala, and Mabibo) using an adaptation of PLACE (Priorities for Local AIDS Control Efforts) methodology, a rapid assessment designed to locate venues where people meet new sexual partners through community informant and camp verification interviews [108]. To be eligible for inclusion in the trial, camps had to meet a number of eligibility requirements including having between 20 and 80 members, having been in existence for at least one year prior to the baseline assessment, and reporting no violent incidents during which a weapon was used within the past 6 months. A total of 303 camps were verified, of which 98 camps did not meet the trial's inclusion criteria. Of the 205 eligible camps identified, 60 camps were randomly selected for inclusion in the trial.

All members of camps selected in to the trial were preliminarily eligible for inclusion in the study. Following random selection, camp leaders were given a camp roster to fill with the names and basic information about each camp member. 1581 men were identified as camp members and assessed for eligibility. To be eligible for participation in the trial at baseline, participants had to: 1) be a registered camp member for at least three months; 2) plan on residing in Dar es Salaam for the next 30 months; 3) be 15 years or older; 4) visit the camp at least once per week; and 5) be willing to provide contact information for themselves and two family members or friends. Based on these criteria, 86 men (5.4%) were ineligible, and 25 (1.6%) refused to participate. We contacted but were unable to schedule appointments with 141 men (8.9%) and were unable to contact 71 individuals after three attempts (4.5%). A total of 1,258 men completed the baseline behavioral assessment. Soon after baseline data collection, camp members from one camp (n=9) requested to be removed from the study because their

leader falsified information regarding the camp's eligibility. This camp was removed, resulting in a final baseline sample of 1,249 men within 59 camp networks. 978 men in the 59 camps completed the follow-up assessment for 78% retention.

Measures

Both sexual behavior variables of condom use and concurrency were measured through self-report. An ordered-categorical condom use variable was calculated from a self-report of condom use over the 3 most recent sexual partners (number of sex acts over the most recent month of the relationship, and number of times condoms used for each partner). Using the proportion of reported sex acts where condoms were used (expressed as a percent, participants were assigned to one of three categories: "never use" (0% use), "some use" (greater than 0%, less than 100%), or "always use" (100%). Because of the possibility of some degree of recall bias over the course of the three most recent relationships, use of a continuous variable of condom use is not indicated as each unit increase would not necessarily indicate an important difference in condom use. Dichotomizing condom use, however, is not recommended by Noar et al. [83], thus use of a three-level categorical condom use variable preserves information in differences in use that can be reliably self-reported while reducing measurement error due to recall bias. Sexual concurrency was evaluated by self-report of any overlapping sexual partnerships for an individual's past 3 partners. Participants were asked to enumerate current and past sexual relationships and to report if they had sex with anyone else during any of these partnerships. Participants reporting any instance of simultaneous sexual relationships either currently or in the past 12 months were coded as displaying concurrency. This measure was developed following best practices recently recommended by USAID [84].

Four types of behavioral norms were measured for both condom use and concurrency: friends' reported behaviors, perceived descriptive norms, perceived injunctive norms, and reported friend

encouragement of the behavior. Men were asked to name their three closest friends in their camp by selecting names from a list of all camp members. The name of each nominated friend was associated with a unique identifier which was used to link this nomination record to the friend's reported behavior according to their own completed questionnaire. Taking the reported behavior of all three friends for each individual, we created an average percent condom use and a percent of friends reporting concurrency measure for each participant. To assess perceived descriptive norms, for condom use men were asked for each of these three closest friends, "*Do you think Friend X uses condoms all the time?*" For injunctive norms men were asked "*Do you think Friend X thinks that he should be using condoms all the time?*" Finally, to assess friends' encouragement of behaviors men were asked "*Has Friend X encouraged you to use condoms all the time?*" Men were asked analogous questions for partner concurrency and provided yes/no answers. We created a measure of the proportion of friends for which the respondent answered "yes" for each item.

Symptoms of anxiety and depression were measured using the Hopkins Symptom Checklist-25 (HSCL-25) [85], specifically a version of the scale previously translated into Kiswahili and tested in Tanzania [86]. This measure inventories symptoms of anxiety and depression in 25 items (10 related to anxiety, 15 to depression) rated on a 4-point Likert-type scale. The scale has been previously validated in Tanzania [86, 87]. Anxiety scores were calculated by taking the mean of the 10 anxiety-related items in the HSCL-25. In the sample the anxiety subscale showed good internal consistency at baseline (Cronbach's alpha = 0.94). Depression scores were calculated by taking the mean of the 15 depression-related items in the HSCL-25. At baseline the depression subscale also showed good internal consistency (a = 0.91).

Covariates included in all analyses include age, education level, economic status, marital status, and treatment condition. Age was calculated based on reported date of birth. In cases where birthdate was not reported, the participant's reported age in years was used. Participants reported the highest

level of education they had completed based on eight possible categories (no education, Standard 4 or less, Standard 5-7, Form 1, Form 2, Form 3, Form 4, Greater than Form 4). For analysis, education responses were collapsed into three categories: primary school or less (responses: no education, Standard 4 or less, Standard 5-7); some secondary school (responses: Form 1, Form 2, Form 3); or secondary school completed or greater (responses: Form 4, Greater than Form 4). Socioeconomic status (SES) was evaluated through an asset index, the Filmer Pritchett Wealth Index [91]. Participants were asked to indicate which of 10 possible assets they owned (e.g. cellphone, television) [92], and a composite score was created by weighting each asset by its loading on the first component in a principle components analysis and then placing individuals on a continuous scale of relative wealth [91]. The composite score for each participant was categorized into terciles based on the entire sample of men and women in our baseline dataset (the lowest 33% of participants were classified as "lowest SES", the highest 33% were classified as "highest SES" and the remainder were classified as "middle SES"). Marital status was evaluated through a single yes/no item in which men were asked if they had ever been married. Finally, as the intervention being evaluated was designed to affect condom use and sexual partner concurrency, treatment condition (treatment or control) was included as a covariate to account for this design effect in all analyses.

Analysis

All statistical analyses were conducted in SAS v 9.4 and used a 2-sided significance level of 0.05. First, we described demographic characteristics of the participants and levels of variables of interest at baseline and at the follow-up. To address missing data from the behavioral survey (primarily due to attrition), we applied sequential multiple imputation using the fully conditional specification in *proc MI*. Forty imputations were created using a linear regression specification for continuous variables and a logistic regression specification for categorical variables. To account for missing network ties in the quadratic

assignment procedure discussed below, we conducted imputation by reconstruction as recommended by Huisman [97]. Specifically, we used random imputation of ties proportional to the observed network density (i.e., the probability of a tie is equal to the observed density of each network), as described by Huisman as an extension of Stork and Richard's method of reconstruction [97].

In a previous study we found that that within-person change in anxiety and depression between baseline and the follow-up assessment predicted follow-up risk behavior controlling for baseline risk behavior (see Chapter 4: Paper 1). In the present study we hypothesized that within-person change in behavioral norms would predict follow-up risk behavior controlling for baseline risk behavior, and that change in norms would moderate the relationship between changes in anxiety and depression, and sexual risk. To account for dependence due to clustering within camps, we fit multilevel models with a generalized link function. All models were fit using *proc glimmix* with quadrature estimation (using 15 quadrature points), random intercepts, and logit and cumulative logit link functions (for dichotomous concurrency, and three-level condom use, respectively). For all models the dependent variable was sexual risk (condom use or concurrency) measured at the follow-up, controlling for baseline sexual risk and the covariates listed above. The primary predictors were change in each behavioral norm score, change in anxiety or depression (baseline score subtracted from follow-up score), and their interaction.

Anxiety and depression are highly comorbid, and this was observed at the symptom-level in our sample (the correlation between symptoms of anxiety and depression was 0.76 at baseline and 0.74 at the follow-up). We were concerned that collinearity between measurements of their symptoms would limit the ability to interpret the independent effects of anxiety and depression when estimated simultaneously. For this reason, we tested three models for each combination of the four social norm moderators and two sexual behavior dependent variables (for a total of 24 models): 1) inclusion of change in anxiety symptoms only; 2) change in depression symptoms only; and 3) change in both anxiety and depression symptoms. For each model, we assessed the interaction between mental health and

peer norm change scores. Where these interaction terms were significant, we probed each significant interaction found by modeling the focal effect at the mean, one standard deviation above the mean, and one standard deviation below the mean of the moderator variable score [99].

For models including friend-reported behaviors as a predictor, to adjust standard errors due to the dependence of observations we included controls for subgroup clustering within each camp. As the same person could be nominated by multiple individuals as a friend, friendship cliques within the camps could lead to unobserved dependence within the data. To identify clique subgroups, we created a distance matrix for each camp where the cell values were the number of steps between each dyad. With each of these matrices we performed a principal components analysis to detect structurally significant subgroups within the camp, calculated the factor loading of each individual on each of the principal components, and included these factor loading terms in the models as a control [100].

In a separate modeling approach to further assess the role of peer influence on condom use and currency in men's full camp networks (as opposed to among men's closest friends), we tested network correlations for condom use and concurrency, controlling for factors associated with friendship formation and the aforementioned demographic and treatment condition control variables. We tested these correlations within each of the 59 camp networks using a Quadratic Assignment Procedure (QAP). QAP is a non-parametric procedure for significance testing which is used to infer social influence by relating measures of structural similarity (in this case, structural similarity is defined by the existence of friendship tie) between two network members to a measure of their similarity on variables of interest [77]. QAP produces an empirical distribution of the model coefficients through random permutation of model matrices against which to compare parameter estimates for hypothesis testing [77]. Given a significance level of 0.05, an observed parameter is determined to be statistically significant if it is greater than 95 percent of the values in the empirical distribution based on 2000 iterations [77].

Ethical Review

The study was approved by the ethical review committees at the University of North Carolina at Chapel Hill and Muhimbili University of Health and Allied Sciences in Dar es Salaam, Tanzania. Individual written informed consent was obtained from all study participants.

5.3 RESULTS

1249 men were interviewed at baseline, of whom 1113 (89%) reported being sexually active. 871 of these 1113 men (78%) participated at the follow-up. Men who did not participate at the follow-up did not significantly differ in their risk behavior from men who did (condom use: $\chi^2 = 0.31 \text{ p} = 0.58$; concurrency: OR = 0.94 p = 0.75). The 1113 men who reported being sexually active at baseline were included in the analyses presented below.

Participant characteristics

Sexually active men interviewed at baseline had an average age of 27 years (range: 15 to 59). Over half had a primary school education or less (59%), nearly a third had graduated from secondary school (31%), and the remaining 11% had some secondary school but had not graduated. A quarter of the men had ever been married (25%) and 38% had children.

	N(%) or Mean ± SD
Age (y)***	26.8 ± 7.1
Currently in school	97 (8.7%)
Education level (ref = less than primary school completed)***	
Primary school or less	652 (58.7%)
Some secondary school	116 (10.5%)
Secondary school completed or greater	342 (30.8%)
Socioeconomic status	
Lowest	291 (26.2%)
Middle	435 (39.1%)
Highest	386 (34.7%)
Ever married	277 (25.0%)
Has children	423 (38.0%)

Table 9. Baseline Participant Characteristics (n=1113)^a

Cross-sectional description and change in key variables

Men had an average score of 1.4 (range 1 to 4 with higher scores indicating more severe symptoms) for both anxiety and depression symptoms at both baseline and at the follow-up assessment (Table 10). Though mean change in depression and anxiety scores were both close to zero, there was substantial variation in change scores (-0.03 ± 0.76 for depression and -0.03 ± 0.72 for anxiety). At baseline 21% reported clinically significant symptoms of depression and 19% reported clinically significant symptoms of anxiety based on common clinic cutoffs (score ≥ 1.75 [90]). At the follow-up, 20% and 16% of respondents met the criteria for depression and anxiety, respectively. The within-person correlation between baseline and follow-up symptoms was 0.08 for anxiety and 0.13 for depression. The correlation between symptoms of anxiety and depression was high at both time points: 0.76 at baseline and 0.74 at the follow-up. Furthermore, the correlation between within-person change in anxiety and depression symptoms was 0.73. At the sample level, there were relatively similar levels of condom use and concurrency across the two time points. About half of men reported never using condoms at both baseline and the followup (53% and 52%, respectively). More men reported always using condoms at baseline than at the follow-up (33% and 26%, respectively). 14% of men reported sometimes using condoms at baseline, and this proportion increased to 22% at the follow-up. The within-person correlation between baseline and follow-up percent condom use was 0.21. Many more men reported concurrency at the follow-up (262, 32%) than at baseline (193, 20%). The within-person correlation between baseline and follow-up concurrency was 0.16.

By friends' own self report, on average, men's three closest friends used condoms 43% of the time at baseline and 38% of the time at the follow-up. Condom use by men's friends decreased by an average of 6% by the follow-up. At baseline, on average, 21% of men's friends reported concurrency, and 30% did so at the follow-up for an average increase of 10%. In comparison, at baseline men perceived on average that 42% of their friends used condoms all the time, and that 24% of their friends had concurrent sexual partners (perceived descriptive norms). They also perceived that 59% of their friends thought that they should use condoms all the time and that only 18% would approve of them having concurrent sexual partners (perceived injunctive norms). Participants reported on average that 49% of friends had encouraged them to use condoms all the time, while 39% had discouraged them from having concurrent sexual partners. Each of these perceptions increased by a factor of 8% to 22% by the follow-up, with the greatest increase occurring in injunctive condom use norms.

	Baseline	Follow-up	Within- person change
Internalizing symptoms			
Depression score	1.43 ± 0.57	1.42 ± 0.56	-0.03 ± 0.76
Anxiety score	1.38 ± 0.51	1.36 ± 0.53	-0.03 ± 0.72
Depression (score \geq 1.75)	237 (21.3%)	174 (20.0%)	
Anxiety (score ≥ 1.75)	206 (18.5%)	141 (16.2%)	
Both depression and anxiety (both scores ≥ 1.75)	164 (14.7%)	101 (11.6%)	
Sexual risk			
Condom use			
Never	491 (52.7%)	408 (51.6%)	
Sometimes	133 (14.3%)	176 (22.3%)	
Always	308 (33.1%)	207 (26.2%)	
Concurrency	193 (20.2%)	262 (32.1%)	
Observed peer norms			
Condom use	0.43 ± 0.37	0.38 ± 0.33	-0.06 ± 0.50
Concurrency	0.21 ± 0.30	0.30 ± 0.34	0.10 ± 0.44
Perceived peer norms			
Descriptive condom use	0.42 ± 0.46	0.54 ± 0.42	0.13 ± 0.57
Injunctive condom use	0.59 ± 0.46	0.81 ± 0.32	0.22 ± 0.52
Condom use encouraged	0.49 ± 0.44	0.66 ± 0.39	0.17 ± 0.55
Descriptive concurrency	0.24 ± 0.38	0.36 ± 0.39	0.11 ± 0.51
Injunctive concurrency	0.18 ± 0.33	0.26 ± 0.38	0.08 ± 0.50
Concurrency discouraged	0.39 ± 0.43	0.48 ± 0.41	0.09 ± 0.57

Table 10. Cross-sectional description of key variables^a

^aData are expressed as No. (%) or Mean ± SD

Association between observed and perceived predictors

In comparing friends' reported behaviors and men's perceptions of behavioral norms among their friends (Table 11), none of the correlations surpassed an absolute value of 0.04, and none of the correlations were statistically significant. Friends' own reports of their condom use and concurrency

were not associated with men's perceptions of their friends' behaviors, their approval of these behaviors, or reports of their friends' encouragement of these behaviors.

Observed measure	Perceived measure			
	Perceived descriptive norm	Perceived injunctive norm	Behavior encouraged	
Condom use	-0.002	0.037	0.038	
Concurrency	0.000	-0.013	-0.034	

Table 11. Baseline correlations between observed and perceived norms

Peer similarity in behaviors, anxiety, and depression

The results of the QAP analyses are presented in Figure 4 below, presented as box plots of the distribution of the individual results for each of the 59 camps. Overall, there was little evidence for peer similarity in concurrency or condom use at the camp level. On average across all camps, friends had 0.92 times the odds of having the same concurrency as non-friends (range: 0.43, 2.38; β (log odds)= -0.08), but the estimated association was only significant at a=.05 level for 9 (15%) of camps. Friends had 0.07 greater similarity in percent condom use than non-friends (range: -0.10, 0.40), but the estimated association was only significant for 7 (12%) of camps.



Figure 4. Box plot distributions of QAP correlations by camp

Condom use models

In assessing moderation by condom use norms of the relationship between mental health and men's own condom use, we found that change in perceived descriptive condom use was significantly associated with men's condom use (aOR 1.63; 95% CI: 1.22, 2.26). Changes in perceived descriptive condom use also significantly interacted with changes in anxiety symptoms in association with condom use (aOR 1.95; 95% CI: 1.10, 3.47). Changes in perceived injunctive norms around condom use were not significantly related to condom use, nor did they significantly interact with anxiety or depression change. Changes in friend encouragement of condom use were significantly associated with men's condom use (aOR 1.54; 95% CI: 1.15, 2.05). Changes in friends' self-reported condom use were not significantly associated with men's own condom use (aOR 0.81; 95% CI: 0.55, 1.19) and did not significantly interact with anxiety or depression change. The estimates presented above are taken from the third series of models which included both changes in anxiety and depression as predictors, with very similar estimates seen in this series of models as compared to series 1 (anxiety predictor only) or series 2 (depression predictor only). Significant interactions are presented in Figure 5 and discussed below.

	Moderator			
	Descriptive	Injunctive	Encourage	Observed
	condom use	condom use	condom use	condom use
	aOR (95%CI)	aOR (95%CI)	aOR (95%CI)	aOR (95%CI)
1. Anxiety and peer co	ondom use			
Main effect model				
ΔAnxiety	0.87 (0.71, 1.07)	0.84 (0.67, 1.03) ⁺	0.86 (0.70, 1.06)	0.83 (0.65, 1.06)
ΔMod	1.66 (1.22, 2.26)**	1.17 (0.87, 1.59)	1.56 (1.17, 2.07)**	0.81 (0.56, 1.18)
Interaction model				
ΔAnxiety	0.82 (0.66, 1.01)	0.78 (0.61, 0.99)*	0.81 (0.65, 1.01)	0.84 (0.65, 1.08)
ΔMod	1.62 (1.19, 2.20)**	1.15 (0.85, 1.56)	1.53 (1.15, 2.05)**	0.81 (0.55, 1.18)
∆Anxiety*∆Mod	1.56 (1.08, 2.24)*	1.30 (0.86, 1.95)	$1.40~(0.98,~2.01)^{\dagger}$	1.10 (0.61, 1.97)
2. Depression and per	er condom use			
Main effect model				
∆Depression (0.77 (0.63, 0.94)*	0.74 (0.60, 0.90)**	0.76 (0.62 <i>,</i> 0.92) [*]	0.73 (0.57, 0.94)*
ΔMod	1.61 (1.19, 2.19)**	1.12 (0.83, 1.52)	1.52 (1.14, 2.03)**	0.82 (0.56, 1.19)
Interaction model				
∆Depression (0.74 (0.60, 0.91)**	0.69 (0.55, 0.87)**	0.74 (0.60, 0.91)**	0.74 (0.57, 0.95)*
ΔMod	1.60 (1.17, 2.17)**	1.11 (0.82, 1.51)	1.51 (1.13, 2.02)**	0.81 (0.56, 1.19)
∆Depression*∆Mod	1.25 (0.90, 1.74)	1.29 (0.88, 1.90)	1.14 (0.83, 1.57)	1.06 (0.62, 1.82)
3. Anxiety, depression, and peer condom use				
Main effect model				
ΔAnxiety	1.14 (0.86, 1.51)	1.04 (0.76, 1.47)	1.14 (0.86, 1.51)	1.09 (0.79, 1.51)
∆Depression (0.71 (0.54, 0.93) [*]	0.69 (0.53, 0.91)**	$0.70~{(0.53, 0.91)}^{*}$	0.69 (0.50, 0.96)*
ΔMod	1.63 (1.20, 2.22)**	1.14 (0.84, 1.55)	1.54 (1.15, 2.05)**	0.81 (0.55, 1.19)
Interaction model				
ΔAnxiety	1.07 (0.79, 1.44)	1.11 (0.83, 1.43)	1.04 (0.77, 1.40)	1.09 (0.78, 1.54)
ΔDepression	0.71 (0.53 <i>,</i> 0.94) [*]	0.67 (0.49, 0.91)**	0.72 (0.54 <i>,</i> 0.95) [*]	0.69 (0.50 <i>,</i> 0.97) [*]
ΔMod	1.58 (1.16, 2.15)**	1.11 (0.82, 1.52)	1.52 (1.14, 2.04)**	0.81 (0.55, 1.18)
ΔAnxiety *ΔMod	1.95 (1.10, 3.47)*	1.25 (0.71, 2.20)	1.75 (1.01 <i>,</i> 3.02) [*]	1.03 (0.45, 2.37)
∆Depression*∆Mod	0.79 (0.46, 1.33)	1.10 (0.64, 1.89)	0.78 (0.48, 1.26)	1.05 (0.48, 2.26)

+ p<0.1; *p<0.05; **p<0.01; ***p<0.001;

Note: All models include controls for condition, age, education level, SES, and ever having been married.

Looking specifically at the interaction between anxiety symptoms and descriptive condom use in association with men's condom use, we present plots of the simple intercepts and slopes of the association between anxiety change and condom use, at the mean, at one standard deviation above, and at one standard deviation below the mean of each condom use norm change score (Figure 5). Here were present simple slopes as beta coefficients based upon the estimates from model series 1 (anxiety predictor only) for the sake of the interpretability of the observed interaction.

At the mean level of change in perceived descriptive condom use, men with greater increases in anxiety symptoms reported lower levels of condom use (simple slope= -0.20, p= 0.035). This association was amplified with decreasing/worsening perceived descriptive norms for condom use (simple slope at one standard deviation below the mean= -0.44, p= 0.004), and attenuated by increasing/improving perceived descriptive norms for condom use (simple slope at one standard deviation above the mean = 0.03, p=0.776).

Similarly, at the mean level of change in condom use encouragement, men with greater increases in anxiety symptoms reported lower levels of condom use (simple slope= -0.21, p= 0.030). This association was amplified with decreasing encouragement of condom use (simple slope at one standard deviation below the mean= -0.39, p= 0.009), and attenuated by increasing encouragement of condom use (simple slope at one standard deviation above the mean = -0.03, p= 0.761).

Anxiety and descriptive condom use norm change interaction plot

Anxiety and encouragement of condom use norm change interaction plot



Figure 5. Interaction plots for condom use models

Concurrency models

In testing the hypotheses regarding moderation by concurrency norms of the relationship between mental health and partner concurrency, changes in perceived descriptive concurrency norms (aOR 1.42; 95% CI: 1.02, 2.00) and injunctive concurrency norms (aOR 1.50; 95% CI: 1.08, 2.10) were significantly associated with men's concurrency. However, changes in levels of concurrency discouragement and observed friend concurrency were not significantly associated with concurrency. None of these hypothesized moderators significantly interacted with either change in anxiety or depression in association with concurrency. The estimates presented above are taken from the third series of models which included both changes in anxiety and depression as predictors, with very similar estimates seen in this series of models as compared to series 1 (anxiety predictor only) or series 2 (depression predictor only).

	Moderator			
	Descriptive	Injunctive	Discourage	Observed
	concurrency	concurrency	concurrency	concurrency
	aOR (95%CI)	aOR (95%CI)	aOR (95%CI)	aOR (95%CI)
1. Anxiety and peer	concurrency			
Main effect model				
ΔAnxiety	1.58 (1.25, 2.00)***	1.59 (1.26, 2.01)***	1.57 (1.24, 1.99)***	1.59 (1.21, 2.10)**
ΔMod	1.43 (1.02, 2.00)*	1.53 (1.09, 2.13)*	0.81 (0.61, 1.08)	0.72 (0.44, 1.17)
Interaction model				
ΔAnxiety	1.67 (1.31, 2.14)***	1.63 (1.29, 2.07)***	1.56 (1.24, 1.97) ^{***}	1.59 (1.19 <i>,</i> 2.12) ^{**}
ΔMod	1.49 (1.06, 2.11)*	1.58 (1.12 <i>,</i> 2.22) ^{**}	0.80 (0.60, 1.06)	0.72 (0.44, 1.17)
∆Anxiety*∆Mod	$0.67~(0.43,~1.05)^{+}$	0.75 (0.47, 1.20)	1.19 (0.77, 1.83)	1.04 (0.53, 2.02)
2. Depression and pe	eer concurrency			
Main effect model				
ΔDepression	1.54 (1.24, 1.91)***	1.53 (1.24, 1.90)***	1.53 (1.24, 1.90)***	1.57 (1.22, 2.02)***
ΔMod	1.42 (1.01, 2.00)*	1.48 (1.06, 2.07)*	0.80 (0.60, 1.07)	0.74 (0.46, 1.21)
Interaction model				
ΔDepression	1.62 (1.29, 2.02)***	1.57 (1.26, 1.96)***	1.53 (1.24, 1.90)***	1.54 (1.19 <i>,</i> 1.99) ^{**}
ΔMod	1.48 (1.06, 2.08)*	1.54 (1.10, 2.16)*	0.79 (0.59, 1.05)	0.71 (0.44, 1.17)
$\Delta Depression^*\Delta Mod$	0.72 (0.47, 1.10)	0.74 (0.47, 1.16)	1.22 (0.84, 1.79)	1.30 (0.67, 2.50)
3. Anxiety, depression, and peer concurrency				
Main effect model				
ΔAnxiety	1.31 (0.94, 1.81)	1.33 (0.96, 1.85)	1.29 (0.93, 1.79)	1.31 (0.90, 1.91)
ΔDepression	1.29 (0.95, 1.74)	1.27 (0.94, 1.72)	1.29 (0.95, 1.75)	1.31 (0.93, 1.84)
ΔMod	1.42 (1.02, 2.00)*	1.50 (1.08, 2.10)*	0.82 (0.61, 1.09)	0.73 (0.45,1.18)
Interaction model				
ΔAnxiety	1.36 (0.95, 1.95)	1.34 (0.95, 1.88)	1.28 (0.92, 1.78)	1.33 (0.90, 1.96)
ΔDepression	1.31 (0.95, 1.81)	1.30 (0.94, 1.78)	1.30 (0.96, 1.76)	1.28 (0.90, 1.81)
ΔMod	$1.50 (1.06, 2.11)^{*}$	1.57 (1.12 <i>,</i> 2.20) ^{**}	0.79 (0.60, 1.06)	0.71 (0.44,1.17)
ΔAnxiety *ΔMod	0.73 (0.37, 1.47)	0.90 (0.46, 1.77)	1.05 (0.55, 2.03)	0.76 (0.29, 2.01)
$\Delta Depression^*\Delta Mod$	0.90 (0.46, 1.75)	0.79 (0.42, 1.51)	1.17 (0.66, 2.07)	1.54 (0.60, 3.95)

 Table 13. Concurrency peer norm interaction model results

+ p<0.1; *p<0.05; **p<0.01; ***p<0.001;

Note: All models include controls for condition, age, education level, SES, and ever having been married.

5.4 DISCUSSION

The results of this study indicate that changes in perceived norms of behaviors among men's closest

friends are associated with men's own condom use and concurrency behaviors. Specifically, perceived

descriptive norms were associated with both condom use and concurrency, perceived injunctive norms

were associated with concurrency, and direct encouragement was associated with condom use.

Changes in perceived condom use norms (descriptive norms and reported encouragement) also interacted with changes in anxiety symptoms to shape condom use. Perceived norms and the observed behaviors of men's closest friends were not significantly correlated. Changes in the observed behaviors of men's closest friends were also not associated with men's own behaviors. Further, there was little evidence of significant observed similarity in these behaviors in men's larger camp friendship networks.

We found multiple significant associations between men's perceptions of behavioral norms among their closest friends and their own behaviors. Such perceptions can affect behavior as a normative behavior is seen as the "correct" or wise thing to do in a given social context [172]. People may also anticipate social acceptance or increase in social status as a result of assuming a normative behavior [173, 174]. Motivation to adhere to the norm may even increase when peers directly reinforce perceptions of anticipated social rewards e.g. through direct encouragement [175]. Among men's closest friends, we found that both perceived descriptive norms and friends' encouragement were associated with men's condom use, and descriptive and injunctive norms were associated with concurrency. While it appears that men's perceptions of their friends' adoption of behaviors (descriptive norms) are related to both condom use and concurrency, future studies should attempt to understand why direct encouragement may be related to condom use but not concurrency, and why perceptions of what behaviors friends find acceptable (injunctive norms) may be related to men's concurrency but not condom use.

We also found that descriptive condom use norms and encouragement to use condoms interacted with anxiety to shape condom use. This interaction between mental health and peer norms can be understood through the fact that peer influence not only acts to directly shape behavior but peers may serve as a reference groups which individuals look to in behavioral decision-making [68]. In this way, higher levels of risk behaviors and approval thereof among direct peers will serve to magnify individual-level risks, such as those posed by poor mental health. Our results echo previous findings

related to mental health, behavioral norms, and alcohol use. One study found that worse mental health in combination with risky peer drinking norms led to more negative outcomes related to drinking behavior [171]. Similarly, a study among college students found that perceived peer drinking norms moderated the association between social anxiety and alcohol use [176], and another study found a similar relationship for depressive symptoms [177]. Building upon this literature, our results provide novel evidence of the interaction between mental health and sexual behavior norms. As we only found interactions in the relationship between anxiety and condom use, future studies should seek to understand why perceived norms might not interact with depression to shape young men's sexual behaviors, and why norms related to concurrency may not magnify the relationship between mental health and concurrency as they seem to do for condom use.

While we found numerous associations between perceived norms and men's behavior, we found no support for the association between men's behavior and the self-reported behaviors of men's closest friends or their larger camp friendship networks. These results were surprising given previous studies which have found that individuals' sexual behaviors, including partner concurrency [58] and condom use [59-62], reflect their peers' behavior. Specifically, these studies found that youth exposed to higher levels of concurrency or lower levels of condom use among their peers were more likely to display concurrency or lower levels of condom use themselves. It is possible that men have other important friendships outside of their camps which may be more relevant to their own sexual behavior than camp friendships. However, the fact that men's perceptions of camp friends' behaviors *were* associated with their own behaviors gives precedence rather to the hypothesis that norms among these friend groups were salient to men's behavioral decision making, but men's perceptions of their friends' behaviors is behaviors.

To this point, we found no significant correlations between men's perceptions of their friends' behaviors and their friends' own self-reported behaviors, similar to previous findings in this population

with regard to perceptions of friends' HIV testing [178]. Such a finding is not entirely surprising, as individuals' perceptions of norms are related to observable peer behaviors, but these perceptions rarely match what peers are actually doing as perceptions of norms are filtered through each individual's unique position and perspective [179]. Previous research has shown that individuals tend to underestimate their peers' protective behaviors and overestimate their risky behaviors [180]. Other studies have observed a false consensus effect, observing a tendency to misperceive peer behavior in a direction that is that is consistent with their own behavior [181, 182]. Further, as people have limited opportunity to observe their friends' behaviors, they often take mental shortcuts and unconsciously extrapolate ideas of descriptive norms using availability heuristics, interpreting norms from observations and stored memories that are most readily available [183]. For example, a peer who has a second concurrent partner may draw more interest or attention than peers who have only one partner, making the examples of concurrency more cognitively available.

Taken together, these perceptual biases could explain the fact that men's perceived norms did not correlate with directly observed reports of friends' behaviors, and may explain the observed relationship between perceived norms and men's own behaviors. Though the change predictors included in our models indicate that within-individual changes in perceived norms were associated with men's behaviors, future research is needed to better determine the temporality in these relationships, i.e. to determine if changes in perceived norms precede changes in behaviors or vice versa. It is also possible that norms and these behaviors are mutually deterministic, that norms shape behaviors and perceived norms may in turn be revised to align more closely with men's own behaviors. A better understanding of these relationships is needed to inform norm change interventions, such as popular opinion leader interventions which aim to leverage the normative influence of key community members often used to target HIV risk behaviors [184-186].

Limitations

There are important limitations to this study. Respondents' willingness to report sexual risk behaviors may have been associated with the social acceptability of such an acknowledgment. In addition, reports of sexual risk behaviors were retrospective, and thus recall bias may be an issue. However, the coding of the condom use and concurrency variables was designed to minimize the influence of recall bias, as discussed in the Methods. The HSCL-25 scale which was used to measure anxiety and depression was developed in a different cultural context than the study setting, though the scale used has been previously validated in Tanzania [86, 87]. We were unable to represent all of the normative influences in men's lives as we did not capture information about the behaviors and norms of peers outside of the camps, thus the results of this study can only represent the normative context within camps. Nevertheless, the relationships and associated norms captured in this study reflect an important peer group for men belonging to these camps. In nominating their closest friends, men were limited to three nominees which may have excluded other close friends who participants may consider to be just as close as those named. There was approximately 22% attrition in the analytic sample at the follow-up assessment. While loss to follow-up might be speculated to occur among men with the poorer mental health and more sexual risk, men who did not participate at the follow-up did not significantly differ in their risk behavior from men who did (condom use: $\chi^2 = 0.31 \text{ p} = 0.58$; concurrency OR = 0.94 p = 0.75). Furthermore, the application of multiple imputation served to minimize the influence of any potential bias from loss to follow-up. Finally, with the present data we cannot say if a changes in norms and mental health preceded changes in men's sexual behaviors. It is possible that the observed association between changes in perceived norms and men's behaviors are the result of a false consensus effect, i.e. that men's perceptions of norms changed to match their own behaviors. Experimental research with successful manipulation of perceived norms is needed to better determine the temporal direction of a potential causal relationship between perceived norms and these behaviors. With the current analysis

we are able to assess the association of changes in mental health and social norms with men's behaviors at the follow-up controlling for their baseline behavior.

Implications for future research and intervention

Future studies should seek to understand why different perceived descriptive, injunctive, and encouragement norms are more closely related to condom use or concurrency, and why perceived behavioral norms would interact with anxiety but not depression to shape sexual risk behaviors. A better understanding of these relationships and potential mechanisms explaining them would help to understand how to target different types of norms for different behaviors to optimize HIV prevention interventions. Scientists should also seek to understand what shapes men's perceptions of their friends' behaviors if not their friends' actual behaviors. Such an understanding would help to identify other potential important psychosocial targets for behavioral interventions. Future HIV prevention interventions should consider the importance of targeting perceived social norms and mental health simultaneously, as our results indicate that these factors act together to shape sexual behavior. In norms-focused interventions, it will be important to consider how participants' mental health, specifically anxiety, could affect the success of the intervention. In programs aiming to promote mental health, interventionists should consider the additional importance of condom use and concurrency norms, and aim to target perceptions of risky norms to prevent risk behaviors. These measures will be particularly important among marginalized populations similar to the camps in the study, as men in these contexts may be exposed to multiple susceptibilities to risk behavior, including poor mental health and risky social norms.

Conclusion

In this study of sexual risk behaviors among young men living in Dar es Salaam, we found that perceived norms affected young men's behaviors, and that these norms interacted with anxiety to shape sexual risk. We further found that perceived norms may not reflect actual peer behaviors, and found no evidence of peer influence in comparing observed behaviors in among men's closest friends and in their larger friendship networks. The results of this study provide novel evidence of the interacting effects of poor mental health and risky perceived norms among a hard to reach population of marginalized young men in Dar es Salaam. As such susceptibilities are common in the many informal settlements throughout eastern and southern Africa, our findings provide important information for future norms-based and mental health promotion interventions targeting HIV prevention in this key population.

CHAPTER 6: AIM 3 RESULTS

Originally the analyses for Aims 2 and 3 were intended to be included in one paper, however the entire results of both Aims were deemed to be too much to include in a single publishable paper by the Chair and the Student. Thus, the second manuscript (Chapter 5) includes the results relating to Aim 2 only. I conducted the analyses for Aim 3 and found no significant associations between cohesion or centrality and men's behavior. Further, none of the hypothesized interactions were significant. Therefore, I have included these results in this abbreviated chapter that includes background, methods, results, and a brief discussion. There is no plan to publish these results.

6.1 BACKGROUND

85% of the 4 million young people ages 15-24 living with HIV are in sub-Saharan Africa [103]. AIDSrelated deaths among youth rose by 50% between 2005 and 2012 and adolescents and young adults account for a growing proportion of African populations [103], making the need to target and engage youth in HIV prevention increasingly important. Condom use and sexual partner concurrency are important behavioral targets for HIV prevention among youth; half of sexually active Tanzanian youth ages 15-24 did not use a condom at last intercourse in 2010, and nearly one-third reported concurrent partners [4].

Anxiety and depression are important correlates of sexual risk [5, 6, 8-10, 104] and may influence risk through multiple mechanisms including substance use [10], maladaptive coping to deal with stress [47], and impaired decision making [48]. We need a better understanding of social resources which may serve to buffer the risk posed by poor mental health. Social support may play such a role;

Cohen's stress buffering theory indicates that social support can operate to intervene between the experience of stress (in this case, the experience of anxiety and/or depression) and maladaptive coping behaviors (such as sexual risk) by reducing the effect of the stressor or psychological processes which occur in response to the stressor [78]. Lazarus and Folkman's Transactional Model of Stress and Coping provides further support for social support's role in promoting positive rather than negative coping outcomes [79], again indicating network social support as a potential moderator between stress related to anxiety and depression and maladaptive coping outcomes (sexual risk behaviors). Lazarus and Folkman posited that social support buffers the relationship between stressors and health behaviors by acting upon coping efforts (actions and processes to deal with the effect of stress), including problem management and emotional regulation [79].

Through the mechanism of social support, social cohesion and greater social connections (centrality) may serve to reduce risk behaviors among people living with anxiety and depression. In her model of social network influence on health behaviors, Lisa Berkman postulates that macro-level network structure and an individual's connections within their networks will shape exposure to microlevel mechanisms, including social support [76]. Following this model, I posit that greater social cohesion and centrality increase opportunities for social support. While there is much interest and previous research on the health promoting effects of social support, a functional aspect of social relationships, little attention has been paid to the health impact of the structure of social relations. In this Aim, I assessed the role of network structure (cohesion) and position (centrality) as moderators of the relationship between mental health and sexual risk. To do this I assessed the following hypotheses:

Hypothesis 3a: Young men's centrality within their camp social network will moderate the relationship between mental health and sexual risk behaviors such that the relationship will be weaker for men with higher centrality than for those with lower centrality. **Hypothesis 3b**: Network social cohesion will moderate the relationship between mental health and sexual risk behaviors such that the relationship will be weaker for young men in camps with higher levels of social cohesion than for those in camps with lower levels of social cohesion.

6.2 METHODS

Study context

This study was conducted in the context of an ongoing HIV prevention trial, *A Multilevel Intervention to Reduce HIV Risk Among Networks of Men in Tanzania* (PI: Suzanne Maman; 1R01MH098690-01), a cluster randomized trial of a microfinance and health leadership intervention to prevent sexually transmitted infections and intimate partner violence [107]. The setting and design of this trial have been previously described in detail [107].The study is taking place in the country's commercial capital, Dar es Salaam, where 7% of the population is living with HIV [2]

For this trial, we identified social networks of young men occupying designated venues known as "camps" in Dar es Salaam where men socialize and engage in small scale enterprise [28]. Participants in the present study are 1113 sexually active men who are members of 59 of these camps. Camps are stable social units; those included in the trial have been in existence for an average of eight years. Men typically belong to only one camp and pay membership fees to belong to that camp [28]. Camps are made up of youth who frequent the venue for its supportive social environment [28]. Most men in the camps are not formally employed and spend several hours each day at their camp [28]. These camps provide unique access to social networks of high risk young men who are otherwise difficult to reach, as they are neither in school nor formally employed [28].

Sampling and data collection

We identified camps for inclusion in the trial in four wards (which are equivalent to U.S. census tracts) of Dar es Salaam (Manzese, Tandale, Mwananyamala, and Mabibo) using an adaptation of PLACE (Priorities for Local AIDS Control Efforts) methodology, a rapid assessment designed to locate venues where people meet new sexual partners through community informant and camp verification interviews [108]. To be eligible for inclusion in the trial, camps had to meet a number of eligibility requirements including having between 20 and 80 members, having been in existence for at least one year prior to the baseline assessment, and reporting no violent incidents during which a weapon was used within the past 6 months. A total of 303 camps were verified, of which 98 camps did not meet the parent study's inclusion criteria. Of the 205 eligible camps identified, 60 camps were randomly selected for inclusion in the study.

All members of camps selected in to the trial were preliminarily eligible for inclusion in the study. Following random selection, camp leaders were given a camp roster to fill with the names and basic information about each camp member. 1581 men were identified as members and assessed for eligibility. To be eligible for participation in the trial at baseline, participants had to: 1) be a registered camp member for at least three months; 2) plan on residing in Dar es Salaam for the next 30 months; 3) be 15 years or older; 4) visit the camp at least once per week; and 5) be willing to provide contact information for themselves and two family members or friends. Based on these criteria, 86 men (5.4%) were ineligible, and 25 (1.6%) refused to participate. We contacted but were unable to schedule appointments with 141 men (8.9%) and were unable to contact 71 individuals after three attempts (4.5%). A total of 1,258 men completed the baseline behavioral assessment. Soon after baseline data collection, camp members from one camp (n=9) requested to be removed from the study because their leader falsified information regarding the camp's eligibility. This camp was removed, resulting in a final

baseline sample of 1,249 men within 59 camp networks. 978 men in the 59 camps completed the follow-up assessment for 78% retention.

Measures

Both sexual behavior variables of condom use and concurrency were measured through self-report. An ordered-categorical condom use variable was calculated from a self-report of condom use over the 3 most recent sexual partners (number of sex acts over the most recent month of the relationship, and number of times condoms used for each partner). Using the proportion of reported sex acts where condoms were used (expressed as a percent, participants were assigned to one of three categories: "never use" (0% use), "some use" (greater than 0%, less than 100%), or "always use" (100%). Because of the possibility of some degree of recall bias over the course of the three most recent relationships, use of a continuous variable of condom use is not indicated as each unit increase would not necessarily indicate an important difference in condom use. Dichotomizing condom use, however, is not recommended by Noar et al. [83], thus use of a three-level categorical condom use variable preserves information in differences in use that can be reliably self-reported while reducing measurement error due to recall bias. Sexual concurrency was evaluated by self-report of any overlapping sexual partnerships for an individual's past 3 partners. Participants were asked to enumerate current and past sexual relationships and to report if they had sex with anyone else during any of these partnerships. Participants reporting any instance of simultaneous sexual relationships either currently or in the past 12 months were coded as displaying concurrency. This measure was developed following best practices recently recommended by USAID [84].

Measures of centrality of and social cohesion were calculated using baseline social network data. Using a roster list including all members of each camp, men were asked to nominate their friends who belonged to their camp by selecting their names from this list. Centrality measures, specifically in-
degree centrality, were calculated as the proportion of people in the individual's camp network who nominated the individual as a friend [95]. In-degree centrality was calculated within each network for each actor using in UCINET [93]. Standardized centrality scores were calculated as follows: Centrality = $InTie_{ij}/(n_j-1)$, where $InTie_{ij}$ is the number of ties received by individual *i* within camp *j*, and n_j is the total number of actors in camp *j*.

Each camp network received a structural and a perceived cohesion score. Structural social cohesion measures for each camp specifically captured the construct of structural cohesion, or the minimum number of nodes who, if removed, would disconnect the network, using methods describes by Moody and White [75]. Each network received a social cohesion score calculated as the average pairwise connectivity, or the mean of the total number of possible independent paths between all pairs of any two given individuals in the network [75]. We calculated these scores in UCINET [93]. We created matrices of pairwise connectivity scores for each dyad by camp. We converted these matrices to pairwise lists of connectivity values for each camp and used mean of these values as the structural cohesion for each camp to test hypothesis 3b. We further calculated a measure of perceived social cohesion for each camp by averaging men's responses to a five item scale at the camp level. We used an adapted measure of social cohesion developed by Sampson, Raudenbush, and Earls [94]. Participants responded to statements related to cohesion of camp members on a 4-point Likert-type scale ranging from 1 = strongly disagree to 4 = strongly agree. Statements included "people in my camp are willing to help each other" and "the members of my camp share the same values."

Symptoms of anxiety and depression were measured using the Hopkins Symptom Checklist-25 (HSCL-25) [85], specifically a version of the scale previously translated into Kiswahili and tested in Tanzania [86]. This measure inventories symptoms of anxiety and depression in 25 items (10 related to anxiety, 15 to depression) rated on a 4-point Likert-type scale. The scale has been previously validated in Tanzania [86, 87]. Anxiety scores were calculated by taking the mean of the 10 anxiety-related items in

the HSCL-25. In the sample the anxiety subscale showed good internal consistency at baseline (Cronbach's alpha = 0.94). Depression scores were calculated by taking the mean of the 15 depression-related items in the HSCL-25. At baseline the depression subscale also showed good internal consistency (a = 0.91).

Covariates included in all analyses include age, education level, economic status, marital status, and treatment condition. Age was calculated based on reported date of birth; in cases where birthdate was not reported, the participant's reported age in years was used. Participants reported the highest level of education they had completed based on eight possible categories (no education, Standard 4 or less, Standard 5-7, Form 1, Form 2, Form 3, Form 4, Greater than Form 4). For analysis, education responses were collapsed into three categories: primary school or less (responses: no education, Standard 4 or less, Standard 5-7); some secondary school (responses: Form 1, Form 2, Form 3); or secondary school completed or greater (responses: Form 4, Greater than Form 4). Socioeconomic status (SES) was evaluated through an asset index, the Filmer Pritchett Wealth Index [91]. Participants were asked to indicate which of 10 possible assets they owned (e.g. cellphone, television) [92] and a composite score was created by weighting each asset by its loading on the first component in a principle components analysis and then placing individuals on a continuous scale of relative wealth [91]. The composite score for each participant was categorized into terciles based on the entire sample of men and women in our baseline dataset (the lowest 33% of participants were classified as "lowest SES," the highest 33% were classified as "highest SES," and the remainder were classified as "middle SES"). Marital status was evaluated through a single yes/no item in which men were asked if they had ever been married. Finally, as the intervention being evaluated is designed to affect condom use and sexual partner concurrency, treatment condition (treatment or control) was included as a covariate to account for this design effect in all analyses.

Analysis

All statistical analyses were conducted in SAS v 9.4 and used a 2-sided significance level of 0.05. To address missing data from the behavioral survey (primarily due to attrition), we applied sequential multiple imputation using the fully conditional specification in *proc MI*. Forty imputations were created using a linear regression specification for continuous variables and a logistic regression specification for categorical variables. To account for missing network ties in the calculation of structural cohesion and in-degree centrality discussed above, we conducted imputation by reconstruction as recommended by Huisman [97]. Specifically, we used random imputation of ties proportional to the observed network density (i.e., the probability of a tie is equal to the observed density of each network), as described by Huisman as an extension of Stork and Richard's method of reconstruction [97].

To test the primary hypothesis while accounting for dependence due to clustering within camps, we fit multilevel models with a generalized link function. All models were fit using *proc glimmix* with quadrature estimation (using 15 quadrature points), random intercepts, and logit and cumulative logit link functions (for dichotomous concurrency, and three-level condom use, respectively). For all models the dependent variable was sexual risk (condom use or concurrency) measured at the follow-up, controlling for baseline sexual risk and the covariates listed above. The primary predictors were the baseline value of each hypothesized moderator, change in anxiety or depression (baseline score subtracted from follow-up score), and their interaction.

Anxiety and depression are highly comorbid, and this was observed at the symptom-level in our sample (the correlation between symptoms of anxiety and depression was 0.76 at baseline and 0.74 at the follow-up). We were concerned that collinearity between measurements of their symptoms would limit the ability to interpret the independent effects of anxiety and depression when estimated simultaneously. For this reason, we tested three models for each combination of the three hypothesized moderators and the two sexual behavior dependent variables (for a total of 18 models): 1) Change in

anxiety symptoms only; 2) change in depression symptoms only; and 3) Change in both anxiety and depression symptoms. For each model, we assessed the interaction between mental health and each respective cohesion or centrality score.

6.3 RESULTS

1113 sexually active men were included in all analyses. Regarding observed network characteristics, the average in-degree centrality at baseline was 0.37 (on average 37% of network peers nominated the individual as a friend; SD: 0.23), and the average network cohesion score was 5.95 (SD: 4.72). Men's perceptions of their camps' cohesion was relatively high, with an average score of 1.53 (range 1 to 5 with lower scores corresponding to greater perceived cohesion). Structural social cohesion was minimally correlated men's perceptions of their camp's cohesion (p= 0.035).

Table 14. Baseline description of hypothesized moderators		
	Mean ± SD	
Observed network measures		
Centrality (popularity)	0.37 ± 0.23	
Network structural cohesion	5.95 ± 4.72	
Perceived cohesion	1.53 ± 0.50	

With regard to condom use, there was some trend indication that structural cohesion was associated with condom use, but there were no statistically significant associations with any hypothesized moderators, nor were there any significant interactions with either anxiety or depression. None of the hypothesized moderators of cohesion or centrality were significantly associated with concurrency, nor were there any significant interactions with either anxiety or depression with concurrency.

	Moderator		
	Structural cohesion	Observed centrality	Perceived cohesion
	aOR (95%CI)	aOR (95%CI)	aOR (95%CI)
1. Anxiety and peer	condom use		
ΔAnxiety	0.83 (0.68, 1.03) l	0.83 (0.67, 1.02) l	0.83 (0.68, 1.02) l
Mod	0.97 (0.94, 1.00) l	0.63 (0.32, 1.25)	1.00 (0.81, 1.23)
∆Anxiety*Mod	1.00 (0.96, 1.04)	1.07 (0.48, 2.38)	1.07 (0.84, 1.37)
2. Depression and peer condom use			
ΔDepression	0.76 (0.62, 0.91)	0.75 (0.62, 0.90)**	0.75 (0.62, 0.91)**
Mod	0.97 (0.94, 1.01) l	0.62 (0.31, 1.23)	1.01 (0.82, 1.25)
∆Depression*Mod	1.00 (0.97, 1.04)	1.25 (0.59, 2.62)	1.03 (0.82, 1.30)
3. Anxiety, depression, and peer condom use			
ΔAnxiety	1.08 (0.80, 1.47)	1.08 (0.80, 1.46)	1.08 (0.79, 1.47)
∆Depression [0.72 (0.54, 0.95)	0.71 (0.54, 0.94)*	0.71 (0.54, 0.94)*
Mod	0.97 (0.94, 1.01) l	0.62 (0.31, 1.23)	1.01 (0.82, 1.24)
∆Anxiety *Mod	0.98 (0.93, 1.04)	0.76 (0.23, 2.46)	1.05 (0.72, 1.52)
∆Depression*Mod	1.02 (0.96, 1.07)	1.51 (0.51, 4.45)	1.00 (0.71, 1.42)
+ p<0.1; *p<0.05; **p<	:0.01; ***p<0.001		

Table 15. Condom use – cohesion and centrality interaction model results

Table 16. Concurrency – cohesion and centrality interaction model results

	Moderator		
	Structural cohesion	Observed centrality	Perceived cohesion
	aOR (95%CI)	aOR (95%CI)	aOR (95%CI)
1. Anxiety and peer	concurrency		
ΔAnxiety	1.60 (1.28, 2.00)***	1.60 (1.28, 2.00)***	1.59 (1.27, 2.00)***
Mod	1.00 (0.96, 1.04)	1.01 (0.47, 2.16)	1.03 (0.83, 1.29)
∆Anxiety*Mod	1.00 (0.95, 1.05)	0.81 (0.31, 2.09)	1.00 (0.75, 1.33)
2. Depression and peer concurrency			
∆Depression (1.57 (1.27, 1.95)***	1.58 (1.27, 1.95)***	1.57 (1.27, 1.94)***
Mod	1.00 (0.96, 1.04)	1.04 (0.48, 2.24)	1.03 (0.83, 1.28)
∆Depression*Mod	1.00 (0.96, 1.05)	1.30 (0.52, 3.25)	1.03 (0.79, 1.33)
3. Anxiety, depression, and peer concurrency			
ΔAnxiety	1.28 (0.94, 1.74)	1.28 (0.94, 1.74)	1.28 (0.94, 1.74)
∆Depression (1.34 (1.00, 1.78) l	1.34 (1.00, 1.79) l	1.33 (1.00, 1.78) l
Mod	1.00 (0.96, 1.04)	1.03 (0.48, 2.22)	1.02 (0.82, 1.27)
∆Anxiety *Mod	1.00 (0.93, 1.07)	0.44 (0.11, 1.72)	1.01 (0.66, 1.55)
∆Depression*Mod	1.00 (0.94, 1.07)	2.28 (0.62, 8.40)	1.03 (0.70, 1.50)
+ p<0.1; *p<0.05; **p<0.01; ***p<0.001			

6.4 DISCUSSION

None of the hypotheses were supported. Structural cohesion, perceived cohesion, and network centrality were not significantly associated with sexual risk behavior, nor did they interact with anxiety or depression in association with sexual risk behavior.

Prior to this research, there were no studies to my knowledge evaluating these hypothesized relationships, nor the direct relationship between network centrality or cohesion and sexual risk behaviors. There is evidence in the literature, though limited, that network centrality predicts receipt of social support [80], which in turn is related to sexual risk behaviors [81, 82]. This evidence suggests social support as the mechanism through which network and centrality and social cohesion may work to buffer the relationship between mental health and sexual risk behaviors. It also indicates a need for future studies to better understand the effects of network structure and position on the relationship between mental health and sexual risks and coping relationships.

The fact that we found no association between perceived or structural cohesion and men's risk behaviors, and no significant interaction with mental health may be due the fact that the intra-class correlation between sexual risk behaviors by camp was relatively low (6.0 percent for condom use and 3.4 percent for concurrency) [170]. Therefore, there was limited variance in condom use and concurrency that was attributable to camp membership which could be explained by camp-level measures of cohesion. With regard to the lack of an observed interaction between men's centrality and their mental health in association with risk behaviors, it is possible that though men lwere more central had more friends than less central men (by definition), that the social support which they received was not of higher quality. Future studies should seek to understand the relationship between network centrality and perceptions of social support.

CHAPTER 7: DISCUSSION AND CONCLUSION

The purpose of this dissertation was to examine the role of anxiety and depression as predictors of sexual risk behaviors among a population of young men living in Dar es Salaam, Tanzania, and to understand the moderating role of normative and supportive elements of young men's peer networks in this relationship. The study was conducted among a population of socioeconomically marginalized young men living in Dar es Salaam belonging to social groups known as "camps," which were identified as important venues for the prevention of intimate partner violence and sexual risk behaviors by the parent study team. The results of the dissertation study contribute important understanding of the psychosocial risk factors for sexual risk behaviors in this key population, and provide important implications for the development of future HIV prevention interventions in this and similar populations.

7.1 Summary of Findings

Aim 1. In the first paper corresponding to the results of Aim 1 (Chapter 4), I found that many men experienced clinically significant symptoms of anxiety and depression (19% and 21%, respectively at baseline). I learned that changes in anxiety and depression were significantly associated with both condom use and concurrency. A secondary model indicated that anxiety and depression may independently affect concurrency but depression may be the primary predictor of condom use. This model suggested that the observed relationship between anxiety and condom use may be attributable to the covariance between anxiety and depression symptoms. These findings provide important evidence of the longitudinal association between mental health and HIV risk in a primarily heterosexual

population of young men in a sub-Saharan urban setting. They also provide much needed understanding of the longitudinal relationship between anxiety and sexual risk in this setting.

Aim 2. In the second paper corresponding to the results of Aim 2 (Chapter 5), I found that changes in perceived norms were associated with young men's behaviors, and that these norms interacted with anxiety to shape sexual risk. Specifically, I found that perceived descriptive norms were associated with both condom use and concurrency, perceived injunctive norms were associated with concurrency, and direct encouragement was associated with condom use. Changes in perceived condom use norms (descriptive norms and encouragement) also interacted with changes in anxiety symptoms to shape condom use. There was not a significant correlation between these perceived norms and the observed behaviors of men's closest friends. Changes in the observed behaviors of men's closest friends. Changes in the observed behaviors of men's closest friends. I further found no evidence of peer influence in comparing observed behaviors among men's closest friends and in their larger camp networks. The results of this study provide novel evidence of the interacting effects of poor mental health and risky perceived sexual behavior norms among a hard to reach population of marginalized young men in Dar es Salaam. The findings provide important information for future norms-based and mental health promotion interventions targeting HIV prevention in this key population.

Aim 3. In the results for Aim 3 (presented in Chapter 6), I found no significant associations between men's sexual behaviors and structural cohesion, perceived social cohesion, or centrality. None of the hypothesized interactions were significant. There was some trend indication that structural cohesion was associated with condom use. As no significant results were found pertaining to Aim 3, they were left out of the second manuscript to be able to present the results related to social norms in more detail, and to be able to present a cohesive narrative in the manuscript.

7.2 Study Strengths

The dissertation study provides important longitudinal evidence of the relationship between mental health and sexual risk in a population of young men living in Dar es Salaam, and the extent to which this relationship is socially influenced. The sociocentric network data from 59 camps of 1249 men available from the parent trial presented the rare opportunity to explore the influence of the observed behaviors of social network peers, and structural elements of the peer network on the relationship between mental health and sexual risk. These strengths are discussed in further detail below.

Key population

The men included in this study represent an important yet hard to reach population. Youth in Tanzania are disproportionately affected by HIV [1], and men are important targets in preventing HIV as gender norms position them to control the terms and conditions of sexual relationships [4], and encourage them to engage in high risk sexual behaviors including inconsistent condom use and sexual partner concurrency [3, 5-8]. Half of sexually active male youth (ages 15-24) surveyed in Tanzania in 2010 did not use a condom at last intercourse [4]. Further, nearly one third of young men in Tanzania reported concurrent sexual relationships in 2010 [4], making condom use and concurrency important target risk behaviors in this population. Many high risk youth are difficult to access for systematic research as they may not be enrolled in school or formally employed. The participants of the dissertation study represent one such difficult to reach population.

The participants in the parent trial who are also included in this study are male members of social groups known as "camps." These groups' spaces were identified by the parent study team as important venues for the prevention intimate partner violence and HIV risk behaviors. An estimated 70% of the Dar es Salaam's population lives in informal settlements [22]. As these camps are located in areas of the city with the lowest income and highest concentration of informal housing, men in camps

represent an important but hard to reach population in the city. Most of the men in these camps are neither in school nor formally employed [28], thus these venues serve as key entry points to understand behavior in a high risk group. Further, these types of camp groups the participants belong to are found elsewhere in eastern and southern Africa [187-190]. Because of changes in the global economy leading to high youth unemployment, the type of population seen in this study may be growing; more and more youth in urban areas of developing countries around the world are participating in social groups or gangs that are similar to the camps I studied [191]. For these reasons, understanding of the social dynamics in these groups which shape sexual risk behaviors may be generalizable to similar hard to reach groups of men throughout the region.

Longitudinal evidence of important risk relationships

The dissertation research provides important understanding of the critical relationship between symptoms of anxiety and depression, and sexual risk behaviors in this key population. The results provide much needed longitudinal evidence of the relationship between internalizing symptoms and HIV risk behaviors among a population of primarily heterosexual men in sub-Saharan Africa. Prior to the dissertation study, we primarily had evidence of cross-sectional associations between both anxiety and depression, and sexual risk behaviors [5-10]. In the extant literature, we had little understanding of the longitudinal association between these internalizing behaviors and sexual risk [8, 9, 11, 12, 49]. We further had limited understanding of such a risk relationship from populations in sub-Saharan Africa, as the bulk of the evidence supporting these relationships came from the United States and other high income contexts [5, 8, 10-15]. Further, due to its prevalence and frequent comorbidity with depression [106], the results provide important evidence of the role of anxiety as a predictor of sexual risk. Such an understanding is particularly needed in this context, as there has been only one prior study examining anxiety as a predictor of sexual risk in a sub-Saharan population to my knowledge [7]. The results of this

study bring us closer to a causal understanding of the relationship between mental health and sexual risk in this context. Future research is needed to fully determine the causality of this relationship which I discuss in further detail in the Limitations section of this chapter.

Social network data

The data available from the parent trial provided access to both egocentric and sociocentric network data within each camp. The availability of information about men's friendships, and the ability to link these data to the reported behaviors of these friends provided two advantages in the dissertation study: 1) the ability to compare the role of observed peer behaviors and perceived behavioral norms; and 2) the ability to understand the moderating role of not only perceived but also structural measures of popularity and social cohesion in the relationship between mental health and sexual risk. By being able to compare friends' observed behaviors and men's normative perceptions of these behaviors, the results raise important questions for future norms-based interventions about the relative salience and relationship between peers' actual behaviors and individuals' perceptions of these behaviors. These implications are discussed further in the Implications for Future Research section of this chapter.

Regarding the ability to assess centrality and social cohesion as moderators of the main effect relationship, though I found no significant moderation by either of these factors, the results provide important preliminary understanding of the role of social networks in shaping the relationship between mental health and sexual risk. It is important to continue to build understanding of peer influence in this relationship as it will allow us to discern if there is something social or learned about sexual risk as a response to poor mental health. This study attempted to understand how to intervene within networks to reduce sexual risk, though I found no evidence that camp social network structure could serve to buffer the relationship between mental health and sexual risk behaviors.

7.3 Study Limitations

There are a number of limitations to this study which may suggest caution in the interpretation of the results. These limitations include: 1) the temporal nature of the two observations used which limits the ability to draw causal conclusions about the associations found; 2) the measurement of sexual behavior by self-report; 3) the measurement of anxiety and depression using a scale which was not developed for this population; 4) the inability to represent the influence of important peers outside of the camp networks. Each limitation is discussed in detail below.

Evaluation of longitudinal relationships

The two time points used in this study as measured a year apart do not allow me to establish whether affective manifestations of anxiety and depression directly precede decisions to engage in risky sexual behaviors. Though there is an established temporal relationship between changes in internalizing symptoms and sexual risk in this study, event-level data is needed to establish the precise temporal relationship [126]. Within the limitations of the data available, the hypotheses as tested allow me to establish that changes in anxiety and depression are associated follow-up sexual behaviors taking into account baseline sexual behavior in this population.

Likewise, with the present data I cannot say if a changes in norms and mental health precede changes in men's sexual behaviors. It is possible that the observed association between changes in perceived norms and men's behaviors are the result of a false consensus effect, i.e. that men's perceptions of norms changed to match their own behaviors. Experimental research with successful manipulation of perceived norms is needed to better determine the temporal direction of a potential causal relationship between perceived norms and these behaviors. With the current analysis I am able to assess the association of changes in social norms with men's behaviors at the follow-up controlling for their baseline behavior. Though the change predictors included in the models indicate that within-

individual changes in perceived norms were associated with men's behaviors, future research is needed to better determine the temporality in these relationships, i.e. to determine if changes in perceived norms precede changes in behaviors or vice versa. It is also possible that norms and behaviors are mutually deterministic, that norms shape behaviors and perceived norms may in turn be revised to align more closely with one's own behaviors. A better understanding of these relationships is needed to inform norm change interventions, such as popular opinion leader interventions which aim to leverage the normative influence of key community members often used to target HIV risk behaviors [184-186].

Measurement of sexual behavior

Sexual behaviors were measured through self-report in this study. Respondents' willingness to report sexual risk behaviors may have been associated with the social acceptability of such an acknowledgment. In addition, reports of sexual risk behaviors were retrospective, and thus recall bias may be an issue. To reduce the influence of social desirability bias in men's responses regarding their sexual behavior, we trained data collectors to establish a rapport with participants and to ask questions as nonjudgmentally as possible. We also used introductory scripts to sets of questions about condom use and concurrency behaviors to remind participants that their responses were strictly confidential, and to normalize the reporting of concurrency behaviors by stating that there is great variety in people's sexual relationships, including the number of sexual partners a person has, and the gender of sexual partners.

I was also careful about the coding of the condom use and concurrency to minimize the influence of recall bias; concurrency was captured dichotomously rather than attempting to quantify the number of concurrent relationships. The measure of concurrency used was developed following best practices recently recommended by USAID [84]. Regarding condom use, because of the possibility of some degree of recall bias over the course of the three most recent relationships, use of a continuous

variable of condom use is not indicated as each unit increase would not necessarily indicate an important difference in condom use. Dichotomizing condom use, however, is not recommended by Noar et al. [83], thus I used a three-level categorical condom use variable to preserve information in differences in use that can be reliably self-reported while reducing measurement error due to recall bias.

Measurement of anxiety and depression

Symptoms of anxiety and depression were measured through retrospective self-report of symptoms over the past week. The HSCL-25 scale which was used to measure anxiety and depression was developed in a different cultural context than the study setting. We used a version of the scale previously translated into Kiswahili and tested in Tanzania [86]. The scale has been previously validated in Tanzania [86, 87]. Given this evidence, I feel comfortable with the theoretical validity of the scale's measurement of anxiety and depression symptoms in this population. This said, given that anxiety and depression symptoms were reported over the past week only, I cannot state that levels of symptoms are reflective of enduring traits of anxiety or depression as opposed to shorter-term states of anxiety or depression which may be highly contextually influenced.

Important relationships outside of camp networks

Finally, men were asked only about their friendships within their own camp, thus the results relating to social norms, centrality, and social cohesion cannot account for the role of important relationships outside of men's camp networks. Specifically in relation to peer norms, there may be other influential peers and family members outside of camp network who serve as important normative reference groups in men's behavioral decision-making. These norms are not captured in these analyses. Nevertheless, the relationships and associated norms captured in this study reflect an important peer

group for men belonging to these camps. The fact that men's perceptions of camp friends' behaviors *were* associated with their own behaviors suggests that these friends are an important normative reference group for these men. Further, in naming their closest friends, men were limited to three nominees which may have excluded other close friends who participants may consider to be just as close as those named. Future studies might rather seek to set a higher limit or have no limit to the number of friends who men can name as close friend.

7.4 Implications for Future Research

The dissertation research raises a number of questions for future research. I discuss the most important questions raised by the study here.

In high HIV prevalence contexts, who is most at risk of poor mental health?

Anxiety and depression are among the most important causes of morbidity globally [193]. Though there are few nationally representative estimates of the prevalence of anxiety and depression in African nations like Tanzania, studies of primary care and HIV treatment clinic populations found approximately a 15% prevalence of depression and a 5-7% prevalence of anxiety [25, 26]. In this study I estimated the prevalence of clinically significant symptoms of anxiety and depression in this population to be about 20%, respectively. Because of the scarcity of mental health resources in Tanzania, we need to understand who is most at risk of anxiety and depression to determine how to best target the existing services to those most in need. There has been little research to date in low resource settings like Tanzania to understand the determinants of depression and anxiety. Among studies conducted in sub-Saharan Africa, depression and associated disorders have been associated with age [134, 135], marital status [134, 136, 137], education [136, 138], and employment and income [135, 136, 138]. There is also evidence that women are at greater risk of depression than men in this setting [134, 136]. Despite the

higher prevalence of depression among women in the literature, the substantial levels of clinically significant symptoms of anxiety and depression among young men in the study population should not be discounted. We need more research specific to men as the determinants of anxiety and depression are likely to be different among men than among women [134, 136].

Why might anxiety relate to concurrency but not condom use?

As depression and anxiety are highly comorbid, I sought to disentangle the respective associations between changes in symptoms of anxiety and depression, and sexual risk behaviors. The analyses suggested that both depression and anxiety were independently associated with concurrency but that most of the observable association between anxiety symptoms and condom use could be attributed to anxiety's covariance with depression. As most research on this topic to date has focused on depression alone as a risk factor for sexual risk behaviors, this represents an important contribution to our understanding of the role of anxiety in this risk pattern. There are multiple theorized pathways through which negative affect in the form of symptoms of anxiety and depression may lead to sexual risk behaviors, including substance use[10], maladaptive coping to deal with stress [47], and impaired decision making [48].

Such mechanisms may help to explain the finding that depression and not anxiety seemed to be independently associated with condom use. Because of the hopelessness, sense of futility, and tendency toward self-destructive behaviors associated with depression [120], men experiencing greater levels of depression may be less likely to use condoms. While the experience of depression is associated with lower condom use, men with symptoms of anxiety may have a range of experiences of anxiety that affect condom use differently. While some individuals may experience excessive worry [121], which may lead to vigilance about condom use, others may experience impulsivity [122] that could lead to disregard for condom use. I hypothesized in the first paper (Aim 1) that if at the individual level feelings

of anxiety encourage or discourage condom use differently for different men, at the population level there may be no consistent observable relationship between general symptoms of anxiety and condom use. I further suggested that future studies should seek to understand how different anxious tendencies (e.g. worry-type vs. impulsive-type) relate to condom use. In the second paper (Aim 2) I found that the relationship between anxiety and condom use was moderated by perceived condom use norms (descriptive and injunctive norms), and that this relationship held when depression was controlled for. It is possible that the normative context may serve to partially suppress the observed relationship between anxiety and condom use.

Why might behavioral norms interact with anxiety and not depression, and only in relation to condom use?

In the second paper (Aim 2) I found an interaction between anxiety and behavioral norms in relation to condom use, but not in relation to concurrency. Further, there were no significant interactions with depression in relation to either behavior. Future studies should seek to understand why perceived norms might not interact with depression to shape young men's sexual behaviors, and why norms related to concurrency may not magnify the effect of mental health on concurrency as they seem to do for condom use.

Previous research regarding the interaction between mental health and peer norms with regard to alcohol consumption have found interactions with both depression and anxiety symptoms. One study found that worse mental health in combination with risky peer drinking norms led to more negative outcomes related to drinking behavior [171]. Similarly, a study among college students found that perceived peer drinking norms moderated the association between social anxiety and alcohol use [176], and another study found a similar relationship for depressive symptoms [177]. To my knowledge, the dissertation research represents the only study to date to test the interaction between mental health and social norms in relation to sexual risk behaviors. Future studies should seek to replicate these

findings to confirm the importance of the specific findings relating to the interaction between anxiety and condom use norms.

Why do different types of norms seem to matter for different behaviors?

In the second paper I found that perceived descriptive norms were associated with both condom use and concurrency, but that injunctive norms were specifically related to condom use, as direct discouragement was to concurrency. Future studies should seek to replicate these findings in other populations. Research is also needed to understand why the effect of injunctive norms and direct encouragement (or peer pressure) would differ by behavior. Previous studies have found injunctive norms to be associated with both condom use [194] and concurrency [195], and have found the same for direct peer encouragement of condom use [196]. For this reason, replication of these findings in other populations is needed before attempting to develop a theoretical understanding of the connections between different behaviors and injunctive and encouragement norms. A better understanding of these relationships and potential mechanisms explaining them would help to understand how to target different types of norms for different behaviors to optimize HIV prevention interventions. Qualitative studies to understand potential differences in the way that men perceive the importance of peer approval and encouragement of different behaviors would be particularly illuminating.

What shapes perceived norms if not friends' actual behaviors?

In the second paper I found no significant correlations between men's perceptions of their friends' behaviors and their friends' own self-reported behaviors. As I found that changes in perceived norms *were* associated with men's sexual risk, it is important to build understanding of what shapes perceptions of sexual behavior norms among youth if not their peers' actual behaviors. Such an

understanding would help to identify important psychosocial targets for behavioral HIV prevention interventions. It should also be considered that perceived norms may indeed reflect peer behaviors, but that individuals may disproportionately weight the behaviors and attitudes of specific peers. Theory indicates that the behavior of highly popular, or highly connected, individuals are more influential than less popular individuals [197]. While perceptions of norms may be particularly shaped by popular peers, they may additionally reflect personal behaviors and previously formed behavioral attitudes and values. Because of the importance of norms perceptions for risk behavior among youth, understanding the relative influence of these factors will be important for the development of group-level risk reduction interventions.

Though perhaps counterintuitive, the lack of correlation I found between friends' self-reported behaviors and men's perceptions of them is similar to previous findings in this population with regard to perceptions of friends' HIV testing [178]. Indeed, individuals' perceptions of norms pertaining to any behavior are related to observable peer behaviors, but these perceptions rarely match what peers are actually doing as perceptions of norms are filtered through each individual's unique position and perspective [179]. Previous research has shown that individuals tend to underestimate their peers' protective behaviors and overestimate their risky behaviors [180]. Other studies have observed a false consensus effect, observing a tendency to misperceive peer behavior in a direction that is that is consistent with their own behavior [181, 182]. Further, as people have limited opportunity to observe their friends' behaviors, they often take mental shortcuts and unconsciously extrapolate ideas of descriptive norms using availability heuristics, interpreting norms from observations and stored memories that are most readily available [183]. For example, a peer who has a second concurrent partner may draw more interest or attention than peers who have only one partner, making the examples of concurrency more cognitively available. Taken together, these psychological biases suggest that in addition to targeting normative opinions among popular or prominent members of a social

group, it may be important to also address realistic perceptions of peer behaviors and attitudes toward risky behaviors, as seen in many college substance use myth busting campaigns [198].

7.5 Intervention Implications

The results of this study have many implications for HIV prevention and related interventions in Dar es Salaam and similar contexts. Chief among these implications are: 1) the need to find feasible approaches to address the burden of anxiety and depression in low-income contexts, particularly in settings with a high HIV prevalence; 2) the importance of targeting individual perceptions to reduce the overestimation of peer risk behaviors; and 3) the importance of targeting multiple risk factors to more holistically account for the susceptibilities which make youth more likely to engage in sexual risk behavior. I discuss both of these implications in further detail below.

Addressing the burden of internalizing disorders in low-income, high HIV prevalence contexts

I estimated the prevalence of clinically significant symptoms of anxiety and depression in this marginalized population of men to be about 20%, respectively. Beyond the study population, very little is known about the mental health of Tanzania's population. There are no estimates of the prevalence of common mental disorders including anxiety and depression to my knowledge, though there have been studies which provide estimates for specific populations. In one study of primary care attendees, the prevalence of depression and "mixed anxiety-depressive disorder" were 15% and 7%, respectively [25]. In another study of HIV-positive patients in rural Tanzania, 16% of patients presented symptoms of depression, and 5% suffered from other anxiety disorders [26]. Though I cannot speak to the prevalence of anxiety and depression in the general population in this setting, the level of mental health services in Dar es Salaam and indeed throughout Tanzania inevitably cannot address the burden of both disorder

classes in the population; per 100,000 population in Tanzania there are only 2 psychiatric nurses, 0.04 psychiatrists, and 0.005 practicing clinical psychologists [27].

Given the paucity of services available and the levels clinically of significant symptoms of anxiety and depression among one out of five men in the study population, efforts to screen and provide treatment for depression and anxiety disorders are indicated as a potentially important component of primary HIV prevention in this setting. Promising models of mental health promotion which may be scalable in this setting include training nurses to implement mental health screening and treatment [129, 130], or even training lay people to screen for common mental disorders to be able to refer individuals meeting clinical criteria to counseling [131-133]. While there are promising interventions to address depression in this context [128, 129], we need more efforts to intervene upon anxiety in low resource settings to the end of reducing sexual partner concurrency at the population level.

The results of this study further indicate the development of effective HIV prevention interventions targeting young men living with anxiety and depression. Because of the indication of my findings that men with worsening symptoms of anxiety and depression may be at greater risk of sexual risk behaviors, efforts to address mental health among high risk populations may contribute to lowering the incidence of HIV. In addition, targeting HIV prevention interventions to sub-populations with higher burdens of anxiety and depression may also prove beneficial to HIV prevention. Previous HIV prevention trials for women have been more effective when they have also been able to improve mental health [127], and we need to test whether such interventions would prove effective among young men as well.

Focusing social norms messaging targeting sexual behavior

As I found that men's perceptions of peer behaviors did not match their friends' actual behaviors, social norms interventions aiming to prevent sexual risk should consider promoting less risky perceived norms rather than or in addition to targeting the behavior of influential individuals. Popular opinion leader

interventions have long attempted to influence group behavior by identifying influential individuals and encouraging them to model behavior and express negative attitudes toward risk behaviors [184-186]. In addition to such interventions, it will be important to develop individual-level interventions to target inaccurate and risky perceptions of sexual behavior norms. As people often overestimate their peers' risk behaviors [76], it may be effective to promote realistic perceptions of peer behaviors and attitudes toward risky behaviors. One successful approach to accomplish this if found in myth-busting interventions, which are often seen in university campaigns to reduce substance use [198]. This approach has also been used to increase condom use and reduce partner concurrency among college students; a brief intervention in which students were asked to estimate condom use and concurrency levels among their peers, and then to compare these estimates to actual data from a campus survey, was effective in increasing condom use and reducing partner concurrency [199]. The feasibility of such interventions is contingent upon the availability of risk behavior data from a salient peer reference group. Where such data are available, the efficacy of this type of intervention should be evaluated among marginalized youth.

Multi-pronged HIV prevention programs

Because of the evidence from this study that mental health and social norms may interact to shape men's condom use behavior, future HIV prevention interventions should consider the importance of targeting perceived social norms and mental health simultaneously. In norms-focused interventions, it may be important to consider how participants' mental health, specifically anxiety, could affect the success of the intervention. In programs aiming to promote mental health, interventionists should consider the additional importance of condom use and concurrency norms, and aim to target perceptions of risky norms to prevent risk behaviors in high HIV prevalence populations. No such combined intervention has ever been conducted to my knowledge. These measures will be particularly

important among marginalized populations similar to the camps in the study, as men living in high HIVprevalence contexts may be exposed to multiple susceptibilities to risk behaviors, including poor mental health and risky social norms.

APPENDIX A: HSCL-25 ITEMS

HSCL-25 items

How much did each of the following symptoms bother or distress you in the last week, including today? 1. Not at all; 2. A little; 3. Quite a bit; OR 4. Extremely

ANXIETY SUB-SCALE

- 1. Suddenly scared for no reason
- 2. Feeling fearful
- 3. Faintness, dizziness, or weakness
- 4. Nervousness or shakiness inside
- 5. Heart pounding or racing
- 6. Trembling
- 7. Feeling tense or keyed up
- 8. Headaches
- 9. Spells of terror or panic
- 10. Feeling restless, can't sit still

DEPRESSION SUB-SCALE

- 11. Feeling low in energy, slowed down
- 12. Blaming yourself for things
- 13. Crying easily
- 14. Loss of sexual interest or pleasure
- 15. Poor appetite
- 16. Difficulty falling asleep, staying asleep
- 17. Feeling hopeless about the future
- 18. Feeling blue
- 19. Feeling lonely
- 20. Thoughts of ending your life
- 21. Feeling of being trapped or caught
- 22. Worrying too much about things
- 23. Feeling no interest in things
- 24. Feeling everything is an effort
- 25. Feelings of worthlessness

APPENDIX B: SEXUAL BEHAVIOR ITEMS

Script: I would now like to ask you about your <u>sexual relationships.</u> When I talk about having sex I mean that you had either vaginal or anal sex with the person.

Many people have different types of sexual partnerships. Some sexual partners are people we have sex with once or twice and some sexual partners are people we have sex with regularly. Some people have many sexual partners and other people have fewer sexual partners. Some people have sex with partner of the same gender.

In the past 12 months, you mentioned that you have had sex with # people. I would like to begin by asking you about the person you have had the longest sexual relationship with during the last 12 months.

No.	Questions and Filters	Coding Categories
Q1501	Tell me about the person you have had the	
	longest sexual relationship with. What is the	
	nickname or initials I could use to talk about this person?	PARTNER #1
Q1502	When was the most recent time you had sex with PARTNER 1?	MONTHS
		WEEKS
		DAYS
Q1503	IS PARTNER 1 a man or a woman?	MAN
		WOMAN
Q1505	How long have you been having sex with this	YEARS
	partner?	MONTHS
		WEEKS
Q1506	What type of relationship do you have with	SPOUSE
	PARTNER 1?	POLYGAMOUS SPOUSE
		COHABITING PARTNER
		GIRLFRIEND/BOYFRIEND
		FRIEND
		CASUAL ACQUAINTANCE
		COMMERCIAL SEX WORKER
		ONE TIME PARTNER
		OTHER
Q1509	Are you still having sex with PARTNER 1?	NO
		YES
Q1510	How often do/did you have sex with PARTNER 1?	EVERY DAY
		A FEW TIMES A WEEK
		FEW TIMES A MONTH
		ABOUT ONCE A MONTH
		ONLY 1-2 TIMES EVER

No.	Questions and Filters	Coding Categories
Q1511	During the last (most recent) time you had sex	NO
	with this partner, did you use a condom?	YES
		DON'T KNOW
Q1512	How many times would you say you had sex with	
	PARTNER 1 during the last one month you had	
	sex with him/her?	
Q1513	Of these (Response from Q1512) times, how	
	many times would you say you used a condom?	
Q1519	While you were in a sexual relationship with	NO
	PARTNER 1, did you have sex with anybody else?	YES
Q1520	As far as you know, during the time you were	NO
	having a sexual relationship with PARTNER 1, do	YES
	you think PARTNER 1 had sex with other people?	
Q1521	How confident are you that you can use a	VERY CONFIDENT
	condom with PARTNER 1?	SOMEWHAT CONFIDENT
		NOT AT ALL CONFIDENT

APPENDIX C: SOCIAL NORM ITEMS

No.	Questions and Filters	Coding Categories
Q1212	Do you think FRIEND 1 uses condoms all the	NO
	time?	YES
Q1213	Do you think FRIEND 1 thinks that he/she should	NO
	be using condoms all the time?	YES
Q1214	Has FRIEND 1 encouraged you to use condoms	NO
	all the time?	YES
Q1215	Do you think FRIEND 1 has more than one sexual	NO
	partner at the same time?	YES
Q1216	Do you think FRIEND 1 thinks that it's ok for	NO
	him/her to have more than one sexual partner at	YES
	the same time?	
Q1217	Has FRIEND 1 discouraged you from having more	NO
	than one sexual partner at the same time?	YES

APPENDIX D: PERCEIVED SOCIAL COHESION ITEMS

Script: Now I'd like to ask you a few questions about how you and your fellow camp members get along in this camp. For each statement, please tell me whether you strongly agree, agree, strongly disagree, or disagree with the statement. When answering these questions, keep in mind that I'm asking you to say whether you agree with the statement based on what is true for **y**ou, not what other people think is true or what your camp thinks is true.

No.	Questions and Filters	Coding Categories
Q401	People in my camp are willing to help each other.	STRONGLY AGREE
		AGREE
		DISAGREE
		STRONGLY DISAGREE
Q402	We are very close to each other in this camp.	STRONGLY AGREE
		AGREE
		DISAGREE
		STRONGLY DISAGREE
Q403	I can trust my fellow camp members.	STRONGLY AGREE
		AGREE
		DISAGREE
		STRONGLY DISAGREE
Q404	The members of my camp get along with each	STRONGLY AGREE
	other.	AGREE
		DISAGREE
		STRONGLY DISAGREE
Q405	The members of my camp share the same values.	STRONGLY AGREE
		AGREE
		DISAGREE
		STRONGLY DISAGREE

REFERENCES

1. ICF International. Tanzania HIV/AIDS and malaria indicator survey. 2013.

2. Tanzania Ministry of Health and Social Welfare. HIV/AIDS/STI surveillance report. 2011; 22.

3. Swai RO, Matee MI, Killewo J, Lyamuya EF, Kwesigabo G, Tulli T, Kabalimu TK, Isingo R, Ndayongeje J. Surveillance of HIV and syphilis infections among antenatal clinic attendees in Tanzania-2003/2004. *BMC Public Health.* 2006; 6(1): 91.

4. ICF Macro. Tanzania demographic and health survey, 2010. National Bureau of Statistics; 2011.

5. Zhan W, Shaboltas AV, Skochilov RV, Kozlov AP, Krasnoselskikh TV, Abdala N. Depressive symptoms and unprotected sex in St. Petersburg, Russia. *J Psychosom Res.* 2012 MAY; 72(5): 371-375.

6. de Morais Peixoto ER, Ribeiro de Barros FC, Crosland Guimaraes MD. Factors associated with unprotected sexual practice among men and women with mental illnesses in Brazil. *Cadernos De Saude Publica*. 2014 JUL; 30(7): 1475-1486.

7. Agardh A, Cantor-Graae E, Ostergren P. Youth, sexual risk-taking behavior, and mental health: A study of university students in Uganda. *Int J Behav Med.* 2012 JUN; 19(2): 208-216.

8. Elkington KS, Bauermeister JA, Zimmerman MA. Psychological distress, substance use, and HIV/STI risk behaviors among youth. *Journal of Youth and Adolescence*. 2010 MAY; 39(5): 514-527.

9. Lundberg P, Rukundo G, Ashaba S, Thorson A, Allebeck P, Ostergren PO, Cantor-Graae E. Poor mental health and sexual risk behaviours in Uganda: A cross-sectional population-based study. *BMC Public Health*. 2011; 11(125): (21 February 2011)-(21 February 2011).

10. Turner AK, Latkin C, Sonenstein F, Tandon SD. Psychiatric disorder symptoms, substance use, and sexual risk behavior among African-American out of school youth. *Drug Alcohol Depend.* 2011; 115(1): 67-73.

11. Traube D, Holloway I, Zhang J. National study of the role of recent illicit substance use on the relationship between depressive symptoms and sexual risk behavior among child welfare-involved adolescents. *Journal of Studies on Alcohol and Drugs.* 2013 jul; 74(4): 589-597.

12. Sterk CE, Theall KP, Elifson KW. The impact of emotional distress on HIV risk reduction among women. *Substance Use & MisusE.* 2006; 41(2): 157-173.

13. Chesney MA, Koblin BA, Barresi PJ, Husnik MJ, Celum CL, Colfax G, Mayer K, McKirnan D, Judson FN, Huang Y, Coates TJ, EXPLORE Study Team. An individually tailored intervention for HIV prevention: Baseline data from the EXPLORE study. *Am J Public Health.* 2003 Jun; 93(6): 933-938. PMCID: PMC1447873.

14. Rogers G, Curry M, Oddy J, Pratt N, Beilby J, Wilkinson D. Depressive disorders and unprotected casual anal sex among australian homosexually active men in primary care. *HIV medicine*. 2003; 4(3): 271-275.

15. Reisner SL, Mimiaga MJ, Skeer M, Bright D, Cranston K, Isenberg D, Bland S, Barker TA, Mayer KH. Clinically significant depressive symptoms as a risk factor for HIV infection among black MSM in massachusetts. *AIDS and Behavior.* 2009; 13(4): 798-810.

16. Hill AL, Rand DG, Nowak MA, Christakis NA. Emotions as infectious diseases in a large social network: The SISa model. *Proceedings of the Royal Society B-Biological Sciences*. 2010 DEC 22; 277(1701): 3827-3835.

17. Rosenquist JN, Fowler JH, Christakis NA. Social network determinants of depression. *Mol Psychiatry*. 2011 MAR; 16(3): 273-281.

18. Shakya HB, Christakis NA, Fowler JH. Association between social network communities and health behavior: An observational sociocentric network study of latrine ownership in rural India. *Am J Public Health.* 2014 MAY; 104(5): 930-937.

19. National Bureau of Statistics. United republic of Tanzania 2012 population and housing census. 2013.

20. World bank data: Tanzania. [cited 9/20/2015]. Available from: <u>http://data.worldbank.org/country/tanzania</u>.

21. World Health Organization. United republic of Tanzania: WHO statistical profile. 2015.

22. United Nations. World urbanization prospects: The 2014 revision. 2014.

23. UNAIDS. Global AIDS response country progress report: Tanzania. 2014.

24. UNFPA. Tanzania country report. 2001.

25. Kerr M, Blizard R, Mann A. General practitioners and psychiatrists: Comparison of attitudes to depression using the depression attitude questionnaire. *Br J Gen Pract.* 1995 Feb; 45(391): 89-92. PMCID: PMC1239142.

26. Marwick KF, Kaaya SF. Prevalence of depression and anxiety disorders in HIV-positive outpatients in rural Tanzania. *AIDS Care.* 2010; 22(4): 415-419.

27. World Health Organization. Mental health atlas. World Health Organization; 2014.

28. Yamanis TJ, Maman S, Mbwambo JK, Earp JAE, Kajula LJ. Social venues that protect against and promote HIV risk for young men in Dar es Salaam, Tanzania. *Soc Sci Med.* 2010; 71(9): 1601-1609.

29. Yamanis TJ, Doherty IA, Weir SS, Bowling JM, Kajula LJ, Mbwambo JK, Maman S. From coitus to concurrency: Sexual partnership characteristics and risk behaviors of 15-19 year old men recruited from urban venues in Tanzania. *AIDS and Behavior*. 2013 SEP; 17(7): 2405-2415.

30. Hill LM, Maman S, Kilonzo MN, Kajula LJ. Anxiety and depression strongly associated with sexual risk behaviors among networks of young men in Dar es Salaam, Tanzania. *AIDS Care.* 2016: 1-7.

31. Sikkema KJ, Watt MH, Drabkin AS, Meade CS, Hansen NB, Pence BW. Mental health treatment to reduce HIV transmission risk behavior: A positive prevention model. *AIDS and Behavior*. 2010; 14(2): 252-262.

32. Barker G, Ricardo C. Young men and the construction of masculinity in sub-Saharan Africa: Implications for HIV/AIDS, conflict, and violence. World Bank Washington, DC; 2005.

33. Noar SM, Morokoff PJ. The relationship between masculinity ideology, condom attitudes, and condom use stage of change: A structural equation modeling approach. *International Journal of Men's Health.* 2002; 1(1): 43-58.

34. Dunkle KL, Jewkes RK, Nduna M, Levin J, Jama N, Khuzwayo N, Koss MP, Duvvury N. Perpetration of partner violence and HIV risk behaviour among young men in the rural Eastern Cape, South Africa. *AIDS*. 2006 OCT 24; 20(16): 2107-2114.

35. Raj A, Santana MC, La Marche A, Amaro H, Cranston K, Silverman JG. Perpetration of intimate partner violence associated with sexual risk behaviors among young adult men. *Am J Public Health.* 2006 OCT; 96(10): 1873-1878.

36. Martin S, Kilgallen B, Tsui A, Maitra K, Singh K, Kupper L. Sexual behaviors and reproductive health outcomes - associations with wife abuse in India. *JAMA-Journal of the American Medical Association*. 1999 NOV 24; 282(20): 1967-1972.

37. Pettifor A, Bekker L, Hosek S, DiClemente R, Rosenberg M, Bull SS, Allison S, Delany-Moretlwe S, Kapogiannis BG, Cowan F. Preventing HIV among young people: Research priorities for the future. *JAIDS J Acquired Immune Defic Syndromes.* 2013; 63: S155-S160.

38. Joint United Nations Programme on HIV/AIDS. Intensifying HIV prevention: UNAIDS policy position paper. World Health Organization; 2005.

39. CDC. Basic information about HIV and AIDS. 2011.

40. National Institute of Allergy and Infectious Diseases, National Institute of Health, Department of Health and Human Services. Workshop summary: Scientific evidence on condom effectiveness for STD prevention. 2001.

41. Koumans E, Farley T, Gibson J, Langley C, Ross M, McFarlane M, Braxton J, St Louis M. Characteristics of persons with syphilis in areas of persisting syphilis in the United States - sustained transmission associated with concurrent partnerships. *Sex Transm Dis.* 2001 SEP; 28(9): 497-503.

42. Potterat J, Zimmerman-Rogers H, Muth S, Rothenberg R, Green D, Taylor J, Bonney M, White H. Chlamydia transmission: Concurrency, reproduction number, and the epidemic trajectory. *Am J Epidemiol.* 1999 DEC 15; 150(12): 1331-1339.

43. Guwatudde D, Wabwire-Mangen F, Eller LA, Eller M, McCutchan F, Kibuuka H, Millard M, Sewankambo N, Serwadda D, Michael N, Robb M, Kayunga Cohort Res Team. Relatively low HIV infection rates in rural Uganda, but with high potential for a rise: A cohort study in Kayunga district, uganda. *Plos One.* 2009 JAN 7; 4(1): e4145.

44. Morrison A, Ellsberg M, Bott S. Addressing gender-based violence: A critical review of interventions. *World Bank Research Observer.* 2007 SPR; 22(1): 25-51.

45. Pilcher CD, Tien HC, Eron JJ, Vernazza PL, Leu S, Stewart PW, Goh L, Cohen MS. Brief but efficient: Acute HIV infection and the sexual transmission of HIV. *J Infect Dis*. 2004; 189(10): 1785-1792.

46. Morris M. Concurrent partnerships and syphilis persistence - new thoughts on an old puzzle. *Sex Transm Dis.* 2001 SEP; 28(9): 504-507.

47. Bachanas PJ, Morris MK, Lewis-Gess JK, Sarett-Cuasay EJ, Flores AL, Sirl KS, Sawyer MK. Psychological adjustment, substance use, HIV knowledge, and risky sexual behavior in at-risk minority females: Developmental differences during adolescence. *J Pediatr Psychol.* 2002 Jun; 27(4): 373-384.

48. Bennett DL, Bauman A. Adolescent mental health and risky sexual behaviour. young people need health care that covers psychological, sexual, and social areas. *BMJ.* 2000 Jul 29; 321(7256): 251-252. PMCID: PMC1118255.

49. Nduna M, Jewkes RK, Dunkle KL, Shai NPJ, Colman I. Associations between depressive symptoms, sexual behaviour and relationship characteristics: A prospective cohort study of young women and men in the Eastern Cape, South Africa. *JOURNAL OF THE INTERNATIONAL AIDS SOCIETY*. 2010 nov; 13.

50. Pitpitan EV, Kalichman SC, Eaton LA, Cain D, Sikkema KJ, Watt MH, Skinner D, Pieterse D. Cooccurring psychosocial problems and HIV risk among women attending drinking venues in a South African township: A syndemic approach. *Annals of Behavioral Medicine*. 2013 mar; 45(2): 153-162.

51. Stahlman S, Grosso A, Ketende S, Sweitzer S, Mothopeng T, Taruberekera N, Nkonyana J, Baral S. Depression and social stigma among MSM in Lesotho: Implications for HIV and sexually transmitted infection prevention. *AIDS and behavior*. 2015 may.

52. Kessler RC, Berglund P, Demler O, Jin R, Merikangas KR, Walters EE. Lifetime prevalence and age-ofonset distributions of DSM-IV disorders in the national comorbidity survey replication. *Arch Gen Psychiatry.* 2005; 62(6): 593-602.

53. Smit J, Myer L, Middelkoop K, Seedat S, Wood R, Bekker L-, Stein DJ. Mental health and sexual risk behaviours in a South African township: A community-based cross-sectional study. *Public Health.* 2006 jun; 120(6): 534-542.

54. Asante KO, Andoh-Arthur J. Prevalence and determinants of depressive symptoms among university students in Ghana. *J Affect Disord.* 2015; 171: 161-166.

55. Gupta R, Dandu M, Packel L, Rutherford G, Leiter K, Phaladze N, Korte FP, Iacopino V, Weiser SD. Depression and HIV in Botswana: A population-based study on gender-specific socioeconomic and behavioral correlates. *PLoS ONE*. 2010; 5(12).

56. Othieno CJ, Okoth R, Peltzer K, Pengpid S, Malla LO. Risky HIV sexual behaviour and depression among University of Nairobi students. *Annals of general psychiatry*. 2015 apr; 14: 16-015--0054--2. 2015.

57. Peltzer K, Pengpid S, Tiembre I. Mental health, childhood abuse and HIV sexual risk behaviour among university students in Ivory Coast. *Annals of General Psychiatry*. 2013 Jun; 12: 18.

58. Catania JA, Coates TJ, Greenblatt RM, Dolcini MM, Kegeles SM, Puckett S, Corman M, Miller J. Predictors of condom use and multiple partnered sex among sexually-active adolescent women: Implications for aids-related health interventions. 1989.

59. DiClemente RJ. Predictors of HIV-preventive sexual behavior in a high-risk adolescent population: The influence of perceived peer norms and sexual communication on incarcerated adolescents' consistent use of condoms. *Journal of Adolescent Health*. 1991; 12(5): 385-390.

60. Romer D, Black M, Ricardo I, Feigelman S, Kaljee L, Galbraith J, Nesbit R, Hornik RC, Stanton B. Social influences on the sexual behavior of youth at risk for HIV exposure. *Am J Public Health.* 1994 Jun; 84(6): 977-985. PMCID: PMC1614950.

61. Walter HJ, Vaughan RD, Gladis MM, Ragin DF, Kasen S, Cohall AT. Factors associated with AIDS risk behaviors among high school students in an AIDS epicenter. *Am J Public Health.* 1992 Apr; 82(4): 528-532. PMCID: PMC1694102.

62. Zapka JG, Stoddard AM, McCusker J. Social network, support and influence: Relationships with drug use and protective AIDS behavior. *AIDS Education and Prevention*. 1993.

63. Berkman LF, Glass T, Brissette I, Seeman TE. From social integration to health: Durkheim in the new millennium. *Soc Sci Med.* 2000; 51(6): 843-857.

64. Friedkin NE. A structural theory of social influence. Cambridge University Press; 2006.

65. Erickson BH. The relational basis of attitudes. *Social structures: A network approach.* 1988; 99: 121.

66. Rice RE, Aydin C. Attitudes toward new organizational technology: Network proximity as a mechanism for social information processing. *Adm Sci Q.* 1991: 219-244.

67. Felps W, Mitchell TR, Byington E. How, when, and why bad apples spoil the barrel: Negative group members and dysfunctional groups. *Research in organizational behavior*. 2006; 27: 175-222.

68. Kemper TD. Reference groups, socialization and achievement. Am Social Rev. 1968: 31-45.

69. Borgatti SP, Everett MG. A graph-theoretic perspective on centrality. *Social networks.* 2006; 28(4): 466-484.

70. Borgatti SP. Centrality and network flow. Social networks. 2005; 27(1): 55-71.

71. Wasserman S, Faust K. Social network analysis: Methods and applications. Cambridge university press; 1994.

72. Bruhn J. The concept of social cohesion. In: The Group Effect. Springer; 2009. p. 31-48.

73. Durkheim E. Suicide: A study in sociology [1897]. *Translated by JA Spaulding and G.Simpson (Glencoe, Illinois: The Free Press, 1951).* 1951.

74. Bollen KA, Hoyle RH. Perceived cohesion: A conceptual and empirical examination. *Social forces*. 1990: 479-504.

75. Moody J, White DR. Structural cohesion and embeddedness: A hierarchical concept of social groups. *Am Social Rev.* 2003: 103-127.

76. Lin N, Ye X, Ensel WM. Social support and depressed mood: A structural analysis. *J Health Soc Behav.* 1999: 344-359.

77. Borgatti SP, Everett MG, Johnson JC. Analyzing social networks. SAGE Publications Limited; 2013.

78. Cohen S, Wills TA. Stress, social support, and the buffering hypothesis. *Psychol Bull.* 1985; 98(2): 310.

79. Lazarus RS, Folkman S. Transactional theory and research on emotions and coping. *European Journal of personality*. 1987; 1(3): 141-169.

80. JuheeHahn, 박혜원, 강은주. Network centrality, social support and self-efficacy in team-based activities. *Korean Business Education Review*. 2010 2010; 62: 47-71.

81. Cordero Coma J. HIV prevention and marriage: Peer group effects on condom use acceptability in rural Kenya. *Soc Sci Med.* 2014 SEP; 116: 169-177.

82. Wilson D, Lavelle S. Psychosocial predictors of intended condom use among Zimbabwean adolescents. *Health Educ Res.* 1992 MAR; 7(1): 55-68.

83. Noar SM, Cole C, Carlyle K. Condom use measurement in 56 studies of sexual risk behavior: Review and recommendations. *Arch Sex Behav.* 2006; 35(3): 327-345.

84. Zelaya CE, Go VF, Davis WW, Celentano DD. Improving the validity of self-reported sexual concurrency in South Africa. USAID; 2012. Report No.: <u>http://www.jhsph.edu/research/centers-and-institutes/research-to-prevention/publications/validity-self-reported-concurrency.pdf</u>.

85. Parloff MB, Kelman HC, Frank JD. Comfort, effectiveness, and self-awareness as criteria of improvement in psychotherapy. *Am J Psychiatry*. 1954; 111(5): 343-352.

86. Lee B, Kaaya SF, Mbwambo JK, Smith-Fawzi MC, Leshabari MT. Detecting depressive disorder with the hopkins symptom checklist-25 in Tanzania. *Int J Soc Psychiatry*. 2008 Jan; 54(1): 7-20.

87. Kaaya SF, Fawzi MC, Mbwambo JK, Lee B, Msamanga GI, Fawzi W. Validity of the Hopkins Symptom Checklist-25 amongst HIV-positive pregnant women in Tanzania. *Acta Psychiatr Scand.* 2002 Jul; 106(1): 9-19.

88. Winokur A, Winokur DF, Rickels K, Cox DS. Symptoms of emotional distress in a family planning service: Stability over a four-week period. *Br J Psychiatry.* 1984 Apr; 144: 395-399.

89. John-Langba J. The relationship of sexual and gender-based violence (SGBV) to sexual-risk behaviors among refugee women in Botswana: The mediating role of depression [dissertation]. US: ProQuest Information & Learning; 2006.

90. Sandanger I, Moum T, Ingebrigtsen G, Dalgard O, Sørensen T, Bruusgaard D. Concordance between symptom screening and diagnostic procedure: The hopkins symptom checklist-25 and the composite international diagnostic interview I. *Soc Psychiatry Psychiatr Epidemiol.* 1998; 33(7): 345-354.

91. Filmer D, Pritchett LH. Estimating wealth effects without expenditure data—Or tears: An application to educational enrollments in states of India. *Demography.* 2001; 38(1): 115-132.

92. Vyas S, Kumaranayake L. Constructing socio-economic status indices: How to use principal components analysis. *Health Policy Plan.* 2006 Nov; 21(6): 459-468.

93. Borgatti SP, Everett MG, Freeman LC. UCINET for windows: Software for social network analysis. 2002.

94. Sampson RJ, Raudenbush SW, Earls F. Neighborhoods and violent crime: A multilevel study of collective efficacy. *Science*. 1997 Aug 15; 277(5328): 918-924.

95. McCulloh I. Social network analysis with applications. Armstrong H(L and Johnson AN, editors. Hoboken, New Jersey: Wiley & Sons, Inc; 2013.

96. Bauer DJ, Curran PJ. Multilevel modeling: Course notes. In press 2015.

97. Huisman M. Imputation of missing network data: Some simple procedures. *Journal of Social Structure*. 2009; 10(1): 1-29.

98. Raudenbush SW. Hierarchical linear models: Applications and data analysis methods. Bryk AS, editor. Thousand Oaks: Sage Publications; 2002.

99. Bauer DJ, Curran PJ. Probing interactions in fixed and multilevel regression: Inferential and graphical techniques. *Multivariate Behavioral Research.* 2005; 40(3): 373-400.

100. Hoff PD. Hierarchical multilinear models for multiway data. *Comput Stat Data Anal.* 2011; 55(1): 530-543.

101. Mathieu JE, Aguinis H, Culpepper SA, Chen G. Understanding and estimating the power to detect cross-level interaction effects in multilevel modeling. *J Appl Psychol.* 2012; 97(5): 951.

102. Joint United Nations Programme on HIV/AIDS (UNAIDS). The gap report. Geneva: UNAIDS. 2014.

103. Idele P, Gillespie A, Porth T, Suzuki C, Mahy M, Kasedde S, Luo C. Epidemiology of HIV and AIDS among adolescents: Current status, inequities, and data gaps. *J Acquir Immune Defic Syndr.* 2014 Jul 1; 66 Suppl 2: S144-53.

104. Agardh A, Odberg-Pettersson K, Östergren P. Experience of sexual coercion and risky sexual behavior among ugandan university students. *BMC Public Health*. 2011; 11(1): 527.

105. Lundberg P, Rukundo G, Ashaba S, Thorson A, Allebeck P, Ostergren PO, Cantor-Graae E. Poor mental health and sexual risk behaviours in Uganda: A cross-sectional population-based study. *BMC Public Health.* 2011 Feb 21; 11: 125-2458-11-125. PMCID: PMC3056745.

106. Kessler RC, Gruber M, Hettema JM, Hwang I, Sampson N, Yonkers KA. Co-morbid major depression and generalized anxiety disorders in the national comorbidity survey follow-up. *Psychol Med.* 2008; 38(03): 365-374.

107. Kajula L, Balvanz P, Kilonzo MN, Mwikoko G, Yamanis T, Mulawa M, Kajuna D, Hill L, Conserve D, Reyes HLM. Vijana Vijiweni II: A cluster-randomized trial to evaluate the efficacy of a microfinance and peer health leadership intervention for HIV and intimate partner violence prevention among social networks of young men in Dar es Salaam. *BMC Public Health.* 2016; 16(1): 1.

108. Weir SS, Pailman C, Mahlalela X, Coetzee N, Meidany F, Boerma JT. From people to places: Focusing AIDS prevention efforts where it matters most. *AIDS*. 2003; 17(6): 895-903.

109. Keinan G. Decision making under stress: Scanning of alternatives under controllable and uncontrollable threats. *J Pers Soc Psychol.* 1987; 52(3): 639.

110. Leith KP, Baumeister RF. Why do bad moods increase self-defeating behavior? Emotion, risk tasking, and self-regulation. *J Pers Soc Psychol.* 1996; 71(6): 1250.

111. Isen AM. Toward understanding the role of affect in cognition. *Handbook of social cognition.* 1984; 3: 179-236.

112. Morris WN, Reilly NP. Toward the self-regulation of mood: Theory and research. *Motiv Emotion.* 1987; 11(3): 215-249.

113. Aneshensel CS, Huba GJ. Depression, alcohol use, and smoking over one year: A four-wave longitudinal causal model. *J Abnorm Psychol*. 1983; 92(2): 134.

114. Cooper ML, Russell M, Skinner JB, Windle M. Development and validation of a three-dimensional measure of drinking motives. *Psychol Assess.* 1992; 4(2): 123.
115. Cooper ML, Shapiro CM, Powers AM. Motivations for sex and risky sexual behavior among adolescents and young adults: A functional perspective. *J Pers Soc Psychol.* 1998; 75(6): 1528.

116. McKirnan DJ, Ostrow DG, Hope B. Sex, drugs and escape: A psychological model of HIV-risk sexual behaviours. *AIDS Care.* 1996; 8(6): 655-670.

117. Parkinson B. Changing moods: The psychology of mood and mood regulation. Addison-Wesley Longman Limited; 1996.

118. Quadland MC. Compulsive sexual behavior: Definition of a problem and an approach to treatment. *J Sex Marital Ther.* 1985; 11(2): 121-132.

119. Beck AT. Depression: Clinical, experimental, and theoretical aspects. University of Pennsylvania Press; 1967.

120. Beck AT, Steer RA, Beck JS, Newman CF. Hopelessness, depression, suicidal ideation, and clinical diagnosis of depression. *Suicide and Life-Threatening Behavior*. 1993; 23(2): 139-145.

121. Ruscio AM, Lane M, Roy-Byrne P, Stang PE, Stein DJ, Wittchen H, Kessler RC. Should excessive worry be required for a diagnosis of generalized anxiety disorder? results from the US national comorbidity survey replication. *Psychol Med.* 2005; 35(12): 1761-1772.

122. Gray JA. Perspectives on anxiety and impulsivity: A commentary. *Journal of research in personality.* 1987; 21(4): 493-509.

123. Agerup T, Lydersen S, Wallander J, Sund A. Longitudinal course of diagnosed depression from ages 15 to 20 in a community sample: Patterns and parental risk factors. *Child Psychiatry & Human Development*. 2014; 45(6): 753-764.

124. Wu P. Longitudinal stability of the beck depression inventory II A latent trait-state-occasion model. *Journal of Psychoeducational Assessment.* 2015: 0734282915582101.

125. Nivard M, Dolan C, Kendler K, Kan K, Willemsen G, van Beijsterveldt C, Lindauer R, van Beek J, Geels L, Bartels M. Stability in symptoms of anxiety and depression as a function of genotype and environment: A longitudinal twin study from ages 3 to 63 years. *Psychol Med.* 2015; 45(05): 1039-1049.

126. Kalichman SC, Weinhardt L. Negative affect and sexual risk behavior: Comment on Crepaz and Marks (2001). 2001.

127. Lennon CA, Huedo-Medina TB, Gerwien DP, Johnson BT. A role for depression in sexual risk reduction for women? A meta-analysis of HIV prevention trials with depression outcomes. *Soc Sci Med.* 2012; 75(4): 688-698.

128. Gaynes BN, Pence BW, Atashili J, O'Donnell JK, Njamnshi AK, Tabenyang ME, Arrey CK, Whetten R, Whetten K, Ndumbe P. Changes in HIV outcomes following depression care in a resource-limited setting: Results from a pilot study in Bamenda, Cameroon. *PloS one.* 2015; 10(10): e0140001.

129. Wagner GJ, Ngo V, Glick P, Obuku EA, Musisi S, Akena D. INtegration of DEPression treatment into HIV care in Uganda (INDEPTH-Uganda): Study protocol for a randomized controlled trial. *Trials.* 2014 JUN 25; 15: 248.

130. Pappin M, Wouters E, Booysen FLR, Lund C. Public sector antiretroviral treatment programme in South Africa: Health care workers' attention to mental health problems. *AIDS Care.* 2015 APR 3; 27(4): 458-465.

131. Chibanda D, Mesu P, Kajawu L, Cowan F, Araya R, Abas MA. Problem-solving therapy for depression and common mental disorders in Zimbabwe: Piloting a task-shifting primary mental health care intervention in a population with a high prevalence of people living with HIV. *BMC Public Health.* 2011 OCT 26; 11.

132. Petersen I, Hancock JH, Bhana A, Govender K. A group-based counselling intervention for depression comorbid with HIV/AIDS using a task shifting approach in South Africa: A randomized controlled pilot study. *J Affect Disord.* 2014 APR; 158: 78-84.

133. Iheanacho T, Obiefune M, Ezeanolue CO, Ogedegbe G, Nwanyanwu OC, Ehiri JE, Ohaeri J, Ezeanolue EE. Integrating mental health screening into routine community maternal and child health activity: Experience from prevention of mother-to-child HIV transmission (PMTCT) trial in Nigeria. *Soc Psychiatry Psychiatr Epidemiol.* 2015 MAR; 50(3): 489-495.

134. Whittier AB, Gelaye B, Deyessa N, Bahretibeb Y, Kelkile TS, Berhane Y, Williams MA. Major depressive disorder and suicidal behavior among urban dwelling Ethiopian adult outpatients at a general hospital. *J Affect Disord.* 2016; 197: 58-65.

135. Jenkins R, Mbatia J, Singleton N, White B. Common mental disorders and risk factors in urban Tanzania. *International journal of environmental research and public health*. 2010; 7(6): 2543-2558.

136. Sipsma H, Ofori-Atta A, Canavan M, Osei-Akoto I, Udry C, Bradley EH. Poor mental health in ghana: Who is at risk? *BMC Public Health*. 2013; 13(1): 1.

137. Nseluke MT, Siziya S. Prevalence and socio-demographic correlates for mental illness among inmates at Lusaka central prison, Zambia. *Med J Zambia*. 2011; 38(2): 3-7.

138. Kebede D, Alem A. Major mental disorders in Addis Ababa, Ethiopia. II. affective disorders. *Acta Psychiatr Scand*. 1999; 100(S397): 18-23.

139. Greif MJ, Dodoo FN, Jayaraman A. Urbanisation, poverty and sexual behaviour: The tale of five African cities. *Urban Stud.* 2010.

140. Zulu EM, Dodoo FN, Chika-Ezeh A. Sexual risk-taking in the slums of Nairobi, Kenya, 1993-98. *Population studies*. 2002; 56(3): 311-323.

141. Magadi MA. The disproportionate high risk of HIV infection among the urban poor in sub-Saharan Africa. *AIDS and Behavior*. 2013; 17(5): 1645-1654.

142. United Nations Human Settlements Programme. Tanzania: Dar es Salaam city profile. 2009.

143. Dar es Salaam City Council. City profile for Dar es Salaam, United Republic of Tanzania. 2004.

144. National Bureau of Statistics. Tanzania integrated labour force survey 2014. 2014.

145. Bowleg L, Neilands TB, Tabb LP, Burkholder GJ, Malebranche DJ, Tschann JM. Neighborhood context and black heterosexual men's sexual HIV risk behaviors. *AIDS and Behavior*. 2014 NOV; 18(11): 2207-2218.

146. Akers AY, Muhammad MR, Corbie-Smith G. "When you got nothing to do, you do somebody": A community's perceptions of neighborhood effects on adolescent sexual behaviors. *Soc Sci Med.* 2011 JAN; 72(1): 91-99.

147. Bobashev GV, Zule WA, Osilla KC, Kline TL, Wechsberg WM. Transactional sex among men and women in the south at high risk for HIV and other STIs. *J Urban Health*. 2009 JUL; 86: S32-S47.

148. Bauermeister JA, Zimmerman MA, Caldwell CH. Neighborhood disadvantage and changes in condom use among African American adolescents. *Journal of Urban Health.* 2011; 88(1): 66-83.

149. South SJ, Baumer EP, Lutz A. Interpreting community effects on youth educational attainment. *Youth & Society.* 2003; 35(1): 3-36.

150. MacLeod J. Ain't no makin'it; 2010.

151. Smith DM, Elander J. Effects of area and family deprivation on risk factors for teenage pregnancy among 13–15-year-old girls. *Psychol Health Med.* 2006; 11(4): 399-410.

152. Robbins RN, Bryan A. Relationships between future orientation, impulsive sensation seeking, and risk behavior among adjudicated adolescents. *J Adolesc Res.* 2004; 19(4): 428-445.

153. Prince DM, Epstein M, Nurius PS, Gorman-Smith D, Henry DB. Reciprocal effects of positive future expectations, threats to safety, and risk behavior across adolescence. *Journal of Clinical Child & Adolescent Psychology*. 2016: 1-14.

154. Curry A, Latkin C, Davey-Rothwell M. Pathways to depression: The impact of neighborhood violent crime on inner-city residents in Baltimore, Maryland, USA. *Soc Sci Med.* 2008; 67(1): 23-30.

155. Wilson WJ. The truly disadvantaged: The inner city, the underclass, and public policy. University of Chicago Press; 2012.

156. Wallis AB, Winch PJ, O'Campo PJ. "This is not a well place": Neighborhood and stress in Pigtown. *Health Care Women Int.* 2010; 31(2): 113-130.

157. Morello-Frosch R, Zuk M, Jerrett M, Shamasunder B, Kyle AD. Understanding the cumulative impacts of inequalities in environmental health: Implications for policy. *Health Aff (Millwood)*. 2011 May; 30(5): 879-887.

158. Latkin CA, Curry AD, Hua W, Davey MA. Direct and indirect associations of neighborhood disorder with drug use and high-risk - sexual partners. *Am J Prev Med.* 2007 JUN; 32(6): S234-S241.

159. Kapadia F, Vlahov D, Donahoe RM, Friedland G. The role of substance abuse in HIV disease progression: Reconciling differences from laboratory and epidemiologic investigations. *Clin Infect Dis.* 2005 Oct 1; 41(7): 1027-1034.

160. Friedman SR, Mateu-Gelabert P, Curtis R, Maslow C, Bolyard M, Sandoval M, Flom PL. Social capital or networks, negotiations, and norms? A neighborhood case study. *Am J Prev Med.* 2007; 32(6): S160-S170.

161. Karasek D, Ahern J, Galea S. Social norms, collective efficacy, and smoking cessation in urban neighborhoods. *Am J Public Health*. 2012; 102(2): 343-351.

162. Baumer E, South S. Community effects on youth sexual activity. *J Marriage Fam.* 2001 MAY; 63(2): 540-554.

163. Davey-Rothwell MA, Siconolfi DE, Tobin KE, Latkin CA. The role of neighborhoods in shaping perceived norms: An exploration of neighborhood disorder and norms among injection drug users in Baltimore, MD. *Health Place*. 2015 MAY; 33: 181-186.

164. Giordano PC. Relationships in adolescence. Annual Review of Sociology. 2003: 257-281.

165. Bernburg JG, Thorlindsson T. Violent values, conduct norms, and youth aggression: A multilevel study in Iceland. *The Sociological Quarterly*. 2005; 46(3): 457-478.

166. Cialdini RB, Reno RR, Kallgren CA. A focus theory of normative conduct: Recycling the concept of norms to reduce littering in public places. *J Pers Soc Psychol.* 1990; 58(6): 1015.

167. Harding DJ. Collateral consequences of violence in disadvantaged neighborhoods. *Social Forces*. 2009; 88(2): 757-784.

168. Browning C, Leventhal T, Brooks-Gunn J. Neighborhood context and racial differences in early adolescent sexual activity. *Demography*. 2004 NOV; 41(4): 697-720.

169. Oostveen T, Knibbe R, DeVries H. Social influences on young adults' alcohol consumption: Norms, modeling, pressure, socializing, and conformity. *Addict Behav.* 1996 MAR-APR; 21(2): 187-197.

170. Mulawa M, Yamanis TJ, Hill LM, Balvanz P, Kajula LJ, Maman S. Evidence of social network influence on multiple HIV risk behaviors and normative beliefs among young tanzanian men. *Soc Sci Med.* 2016; 153: 35-43.

171. Pedersen ER, Miles JN, Hunter SB, Osilla KC, Ewing BA, D'Amico EJ. Perceived norms moderate the association between mental health symptoms and drinking outcomes among at-risk adolescents. *Journal of studies on alcohol and drugs.* 2013; 74(5): 736.

172. Cialdini RB, Trost MR. Social influence: Social norms, conformity and compliance. 1998.

173. Bandura A, McClelland DC. Social learning theory. 1977.

174. Heilbron N, Prinstein MJ. Peer influence and adolescent nonsuicidal self-injury: A theoretical review of mechanisms and moderators. *Applied and Preventive Psychology*. 2008; 12(4): 169-177.

175. Brechwald WA, Prinstein MJ. Beyond homophily: A decade of advances in understanding peer influence processes. *J Res Adolesc.* 2011; 21(1): 166-179.

176. Buckner JD, Ecker AH, Proctor SL. Social anxiety and alcohol problems: The roles of perceived descriptive and injunctive peer norms. *J Anxiety Disord*. 2011; 25(5): 631-638.

177. Chang M, Tiffany K, Escobar M, Oscar S, Richter B, AS JG. The influence of perceived alcohol norms and depression on the drinking behaviors of Latino college students.

178. Mulawa M, Yamanis TJ, Balvanz P, Kajula LJ, Maman S. Comparing perceptions with actual reports of close friend's HIV testing behavior among urban Tanzanian men. *AIDS and Behavior*. 2016: 1-9.

179. Tankard ME, Paluck EL. Norm perception as a vehicle for social change. *Social Issues and Policy Review*. 2016; 10(1): 181-211.

180. Black SR, Schmiege S, Bull S. Actual versus perceived peer sexual risk behavior in online youth social networks. *Translational behavioral medicine*. 2013 2013-Sep; 3(3): 312-9.

181. Ross L, Greene D, House P. The "false consensus effect": An egocentric bias in social perception and attribution processes. *J Exp Soc Psychol.* 1977; 13(3): 279-301.

182. Prinstein MJ, Wang SS. False consensus and adolescent peer contagion: Examining discrepancies between perceptions and actual reported levels of friends' deviant and health risk behaviors. *J Abnorm Child Psychol.* 2005; 33(3): 293-306.

183. Fiske ST, Taylor SE. Social cognition reading. MA: Addison-Wesley. 1984.

184. Kelly JA, Murphy DA, Sikkema KJ, McAuliffe TL, Roffman RA, Solomon LJ, Winett RA, Kalichman SC, Collaborative, The Community HIV Prevention Research. Randomised, controlled, community-level HIV-prevention intervention for sexual-risk behaviour among homosexual men in US cities. *The Lancet.* 1997; 350(9090): 1500-1505.

185. Jones KT, Gray P, Whiteside YO, Wang T, Bost D, Dunbar E, Foust E, Johnson WD. Evaluation of an HIV prevention intervention adapted for black men who have sex with men. *Am J Public Health*. 2008; 98(6): 1043-1050.

186. Sikkema KJ, Kelly JA, Winett RA, Solomon LJ, Cargill VA, Roffman RA, McAuliffe TL, Heckman TG, Anderson EA, Wagstaff DA, Norman AD, Perry MJ, Crumble DA, Mercer MB. Outcomes of a randomized community-level HIV prevention intervention for women living in 18 low-income housing developments. *Am J Public Health.* 2000 Jan; 90(1): 57-63. PMCID: PMC1446110.

187. Covey HC. Street gangs throughout the world. Charles C Thomas Publisher; 2010.

188. Kynoch G. From the ninevites to the hard livings gang: Township gangsters and urban violence in twentieth-century South Africa. *African Studies*. 1999; 58(1): 55-85.

189. Paz Soldan VA. How family planning ideas are spread within social groups in rural Malawi. *Stud Fam Plann.* 2004; 35(4): 275-290.

190. Spergel IA. Youth gangs: Continuity and change. *Crime and justice.* 1990; 12: 171-275.

191. Nilan P, Feixa C. Global youth?: Hybrid identities, plural worlds. Routledge; 2006.

192. Crepaz N, Marks G. Are negative affective states associated with HIV sexual risk behaviors? A metaanalytic review. *Health Psychology*. 2001 jul; 20(4): 291-299.

193. World Health Organization. Global health estimates 2014 summary tables: Deaths by cause, age and sex, 2000–2012. *World Health Organization*. 2014.

194. Devine-Wright H, Abraham C, Onya H, Ramatsea S, Themane M, Aaro LE. Correlates of condom use and condom-use motivation among young South Africans. *J Appl Soc Psychol.* 2015 DEC; 45(12): 674-683.

195. Davey-Rothwell MA, Villarroel MA, Grieb SD, Latkin CA. Norms, attitudes, and sex behaviors among women with incarcerated main partners. *Journal of Urban Health.* 2013; 90(6): 1151-1165.

196. Arrington-Sanders R, Morgan A, Oidtman J, Gomez MC, Ogunbajo A, Trent M, Fortenberry JD. Context of first same-sex condom use and nonuse in young black gay and bisexual males. *J Res Adolesc*. 2016 DEC; 26(4): 1009-1021.

197. Friedkin NE. Theoretical foundations for centrality measures. *American journal of Sociology.* 1991: 1478-1504.

198. Stewart LP, Lederman UC, Golubow M, Cattafesta JL, Goodhart FW, Powell RL, Laitman L. Applying communication theories to prevent dangerous drinking among college students: The RU SURE campaign. *Communication Studies.* 2002; 53(4): 381-399.

199. Chernoff R, Davison G. An evaluation of a brief HIV/AIDS prevention intervention for college students using normative feedback and goal setting. *AIDS Education and Prevention*. 2005 APR; 17(2): 91-104.